

TENDER FOR

DESIGN, CONSTRUCTION, SUPPLY, TESTING AND COMMISSIONING OF 50 PAX HYBRID ELECTRIC PASSENGER FERRY

TENDER No. IWAI/TECH/RP/ELECTRICCATAMARAN/2025

INLAND WATERWAYS AUTHORITY OF INDIA

A-13, Sector-1, Gautam Buddha Nagar, NOIDA (U.P.) - 201301

Web Site:-www.iwai.nic.in

June 2025

DISCLAIMER

- 1. This Tender Document is neither an agreement nor an offer by the Inland Waterways Authority of India (IWAI) to the prospective Applicants or any other person. The purpose of this Tender Document is to provide information to the interested parties that may be useful to them in the formulation of their Bid pursuant to this Tender Document.
- 2. IWAI does not make any representation or warranty as to the accuracy, reliability or completeness of the information in this Tender Document and it is not possible for IWAI to consider particular needs of each party who reads or uses this Tender Document. This Tender Document includes statements which reflect various assumptions and assessments arrived at by IWAI in relation to the Works of Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry such assumptions, assessments and statements do not purport to contain all the information that each Applicant may require. Each prospective Applicant should conduct its own independent investigations and analyses and check the accuracy, reliability and completeness of the information provided in this Tender Document and obtains independent advice from appropriate sources.
- 3. IWAI will not have any liability to any prospective Company/ Firm/Consortium or any other person under any laws (including without limitation the law of contract, tort), the principles of equity, restitution or unjust enrichment or otherwise for any loss, expense
 - or damage which may arise from or be incurred or suffered in connection with anything contained in this Tender Document, any matter deemed to form part of this Tender Document, the award of the Assignment, the information and any other information supplied by or on behalf of IWAI or their employees, any Contractors or otherwise arising in any way from the selection process for the Assignment. IWAI will also not be liable in any manner whether resulting from negligence or otherwise however caused arising from reliance of any Applicant upon any statements contained in this Tender Document.
- 4. IWAI will not be responsible for any delay in receiving the Bids. The issue of this Tender Document does not imply that IWAI is bound to select a Bidder or to appoint the Successful Bidder, as the case may be, for the Works and IWAI reserves the right to accept / reject any or all of Bids submitted in response to this Tender Document at any stage without assigning any reasons whatsoever. IWAI also reserves the right to withhold or withdraw the process at any stage with intimation to all who submitted the Bids.
- 5. The information given is not an exhaustive account of statutory requirements and should not be regarded as a complete or authoritative statement of law. IWAI accepts no responsibility for the accuracy or otherwise for any interpretation or opinion on the law expressed herein.
- 6. IWAI reserves the right to change / modify / amend any or all provisions of this

Design, Construction, S	Supply,	Testing	and	Commis	sioning	of 50	Pay
		Hy	brid	Electric	Passen	ger F	erry

Tender Document. Such revisions to the Tender Document / amended Tender Document will be made available on the website of IWAI.

Table of Contents

SECTION	ON-1: NOTICE INVITING E-TENDER	6
SECTI	ON-II: INSTRUCTIONS TO BIDDERS (ITB)	9
1	Background	10
2	Introduction	10
3	Bidder Eligibility Criteria	11
4	Pre-bid Meeting.	13
5	Clarifications and Addendums	13
6	Preparation of Bids	14
7	Conflict of Interest	19
8	Acknowledgement by Bidder	20
9	Guidelines for e-submission of the Bids	20
10	Submission of Bids	23
11	Extension of Bid submission date	26
12	Late Proposals	27
13	Liability of the employer	27
14	Modification/Substitution/ Withdrawal of Bids	27
15	Bid opening and evaluation process	27
16	Qualification Criteria and Bid evaluation	28
17	Award of Contract	31
18	Mobilization of Equipment, Men & Materials Error! Bookmark not define	
19	Ownership of Document and Copyright	31
SECTI	ON - III: DATA SHEET	33
	ON-IV: TECHNICAL BID STANDARD FORMS	
FORM	4B: Eligible Projects	41
FORM	4C: Average Annual Turnover of the Bidder Along with Loss & Profit Statement	43
	4 D: Power of Attorney	
FORM	4E: List of Key Personnel	46
FORM	4F: List of Ongoing Assignments	47
	4G: Declaration by the Bidders	
	4H: Bidders Information Form	49
FORM	4I: Availability of Vessel Building Facility Including Infrastructural Facility, Equipment,	
	and Machineries ETC to be used for Tender Work	
	4J: Format for Pre-Bid Queries by Bidders	
	4K: Power of Attorney for Lead Member of JV/Consortium	
	4L: Statement of Legal Capacity	
	4M: Joint Bidding Agreement	
	4N: General Construction Experience	
	4O: Bid Capacity	
FORM	4 P: Bid Security Declaration	64

SECTION - V: FINANCIAL BID STANDARD FORMS	66
Form Fin – 1: Financial Bid Submission Form	67
Item Rate BoQ	68
SECTION-VI: TERMS OF REFERENCE (ToR)	69
PART I – DETAILED SCOPE OF WORK	70
PART II – TECHNICAL SPECIFICATIONS	78
SECTION - VII: GENERAL CONDITIONS OF CONTRACT (GCC)	124
SECTION - VIII: SPECIAL CONDITIONS OF CONTRACT (SCC)	199
SECTION-IX: ANNEXES	207
ANNEX - I: BANK GUARANTEE FORM FOR PERFORMANCE SECURITY	208
ANNEX - II: AGREEMENT FORM	211
ANNEX -III: DETAILS OF BANK ACCOUNT	213
ANNEX-IV: BANK CERTIFICATION	214
ANNEX-V: TENDER ACCEPTANCE LETTER	215
ANNEX-VI: BANK GUARANTEE FORM FOR EMD	216
ANNEX-VII: LETTER OF CONSENT	218
ANNEX-VIII: INTEGRITY AGREEMENT	219

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
SECTION-I: NOTICE I	NVITING E-TENDER
	6

INLAND WATERWAYS AUTHORITYOF INDIA

(MINISTRY OF PORTS, SHIPPING AND WATERWAYS, GOVT. OF INDIA) A-13, Sector-1, Gautam Buddha Nagar, NOIDA (U.P.) - 201301 Tel: (0120) 2474007

Web site: www.iwai.nic.in; https://eprocure.gov.in/eprocure/app

NOTICE INVITING E-TENDER

Introduction:

Chief Engineer (NER & Other NNW's), IWAI on behalf of Chairman, Inland Waterways Authority of India (IWAI) invites online Bids **from the reputed Contractors/Companies/Firms** in two cover system (Cover – I: Technical Bid and Cover – II: Financial Bid) for "**Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry**"

Critical Data sheet:

Interested parties may download the Tender document online from the site https://eprocure.gov.in/eprocure/app and IWAI's website "www.iwai.nic.in" and pay INR 11,800/- (Rs.10,000+ Rs.1,800 i.e., 18% GST) (Rupees Eleven Thousand and Eight Hundred only) as the cost of tender document / tender fee to be deposited to IWAI fund through RTGS / NEFT.

As this is a composite work contract, exemptions to MSME is not allowed.

Some important dates for this tender process are as follows:

Document Download Start Date	01.07.2025 at 09.00 hrs.
Pre-bid meeting	11.07.2025 at 1500 hrs.
Bid Submission Last Date	31.07.2025 up to 1500 hrs.
Technical Bid Opening Date	01.08.2025 at 1530 hrs.
Cost of Tender Document	INR 11,800

a) Scope of the work:

In brief, the Scope of Work for the appointed firm shall be **Design**, **Construction**, **Supply**, **Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry** as per the terms and conditions and technical specifications stipulated in the Tender Document. The detailed Terms of Reference (ToR) / Scope of Work shall be as described in the **Section-VI** of this Tender Document

b) Method of Selection:

Bidder will be selected under Cost Based Selection-L1 (CBS) and procedures described in this Tender Document.

c) Clarifications:

Clarification / Query if any on the Tender Document shall be obtained from the following address:

Chief Engineer (NER & other waterways)

A-13, Sector-1, Gautam Buddha Nagar,

NOIDA (U.P.) - 201301

Email: <u>ssingh@iwai.gov.in</u>, <u>srafat@iwai.gov.in</u>, <u>ppolagani.iwai@nic.in</u>, <u>rpatel@iwai.gov.in</u>

Website: www.iwai.nic.in

d) IWAI reserves the right to accept or reject any or all Tenders without assigning any reason and no correspondence shall be entertained in this regard.

Chief Engineer (NER & other waterways), IWAI, NOIDA

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India) Design, Construction, Supply, Testing and Com (Hybrid Elec	missioning of 50 Pax tric Passenger Ferry
SECTION-II: INSTRUCTIONS TO BIDDERS (ITB)

1 Background

- Inland Waterways Authority of India (IWAI) [hereinafter referred to as "employer"] is a statutory body under the Ministry of Shipping, Government of India. IWAI was set up in 1986 for regulation and development of Inland Waterways for the purposes of shipping and navigation. IWAI is primarily responsible for development, maintenance and regulation of Inland Water Transport (IWT) in the country and specifically National Waterways (NW). In April, 2016, the Government of India has declared 106 new National Waterways in addition to the existing five National Waterways.
- 1.2 Inland Water Transport (IWT) has the potential to form the most economic, reliable, safe and environmentally friendly form of transport. When developed for use by modern inland waterway vessels, it can reduce investment needs in rail and road infrastructure, promote greater complementarities in the riparian states, enhance intra- regional trade and, through increased economies of scale, significantly reduce overall logistics costs for the benefit of the entire economy and India's global trade competitiveness.
- 1.3 Urban water transport in India is gaining momentum as a sustainable solution for easing congestion and reducing pollution in metropolitan areas. With initiatives like the development of waterways and introduction of modern vessels, cities are embracing water transport to enhance connectivity and promote eco-friendly commuting options.
- 1.4 The Harit Nauka Guidelines showcases the Govt's commitment towards transforming the maritime landscape by adopting green vessels & establish operationalization of green ecosystem. It also aims to embrace low/zero emission sources of fuel & attain 100% green vessels in the Indian waters by 2047.
- 1.5 With this aim, IWAI introduced the concept of green fuelled boats, the Government of India also aims to foster the confidence of the stakeholders in the maritime sector and usher them towards transitioning to greener and cleaner fuels as emphasized under the Harit Nauka- Inland Vessels Green Transition Guidelines, published on 8th January 2024.

Through this tender, IWAI invites bid for procurement of 50 pax electric catamaran from Shipbuilders.

2 Introduction

2.1 The Employer will select a firm / organization (the "Contractor") in accordance with the evaluation process and method of selection specified in clause 15 & clause 16.2 Section- II: Instruction to Bidder (ITB).

- 2.2 The name of the Assignment/Job has been mentioned in Section III: Data Sheet.

 Detailed scope of the Assignment/Job has been described in Section VI: Terms of Reference
- 2.3 The date, time, and address for submission of the Bids have been given in Section III: Data Sheet.
- 2.4 Bidder shall bear all costs associated with the preparation and submission of their Bids and contract negotiation.
- 2.5 The Employer is not bound to accept any Bid and reserves the right to annul the selection process at any time prior to contract award, without thereby incurring any liability to the Bidder.

3 Bidder Eligibility Criteria

The Bidders shall meet the following pre-qualification criteria: -

- 3.1 Bidder may be a firm that is a private entity, a government owned entity or any combination of such entities in the form of JV/consortium under an existing agreement or with the intent to enter into such agreement supported by a Letter of Intent. Bidders that are Government owned entity in the employer's country may participate only if they can establish that they (i) operate under commercial law and (ii) are not dependent agencies of the employer.
- The Bidder shall meet the Qualification criteria of executing similar works of the 3.2 value as mentioned in clause 16.1 of ITB. The Bidder shall indicate the value of the order executed by him together with the details of name of the party, order value, scope of work / component breakup, completion period stipulated in the order and actual completion period. The completion certificate, awarded by the client on its letter head should have a mention of start date, date of completion and value of the work executed by the Bidder. The Bidder can provide the details of the works that have been substantially completed. "Substantially Completed Works" shall be based on completion of 80% or more works that have been financially completed under the claimed relevant component of the contract. Substantial completion of works shall be supported by client certificates with the name of the work including component breakup of relevant work to the extent of similar work, work order value, start date, percentage of similar work completed till date of issue of the Tender in financial terms (Net Payment received against the Contract) as indicated in Form 4B. In case the work was performed by the Bidder in a JV, the same shall be supported by a Client Certificate enumerating the claimant share also. In case the work was performed by the bidder as a sub-contractor, the bidder shall submit similar completion certificate awarded to it by the main contractor and countersigned by the Employer / Client of the main contractor.

- 3.3 Copy of work order/letter of award/letter of work agreement alone shall not suffice Bidders claim for executing the similar work as defined in Section III data sheet. Submitting completion certificate from the client on its letter head is mandatory. In case the work of a foreign firm is being submitted for pre qualification, the document should be apostilled.
- Average Annual Turnover during the last three (03) years ending 31st March of the previous financial year should be as mentioned in clause 16.1.2 of ITB. The bidder should be profitable organization based on Profit before Tax (PBT) for at least 02 (two) out of 3 (three) financial years mentioned above. The Bidders shall provide financial turnover of the firm for the last three years along with profit & Loss statement for corresponding years duly certified by statutory auditor(s).
- 3.5 Any entity which has been barred by the Central Government, any State Government, a statutory Employer or a public sector undertaking, or International Funding Agency (World Bank, ADB, JICA etc.), as the case may be, from participating in any project, and the bar subsists as on the Bid submission date, would not be eligible to submit Bid.
- 3.6 The similar work experience of parent company/subsidiary/sister Company or any associated firm/company of the Bidder shall not be considered unless the parent company/subsidiary/sister company or associated firm/company is part of the JV/consortium participating in the Bid.
- 3.7 The Bidder shall offer and make available the list of all Key Personnel as specified in Clause 7 of Section VI: Terms of Reference.
- 3.8 Bidder should have, during the last three years, neither failed to perform on any agreement, as evidenced by imposition of a penalty by an arbitral or judicial Employer or a judicial pronouncement or arbitration award against the Bidder, nor been expelled from any project or agreement nor have had any agreement terminated for breach by such Bidder.
- 3.9 The Bidder shall also indicate following:
- 3.9.1 The Bidder shall have adequate resources for successful execution for **Design**, **Construction**, **Supply**, **Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry** and should be financially solvent. Bidder shall provide a solvency certificate from any nationalized/scheduled bank in India for an amount as indicated in Section III: Data Sheet.
- 3.9.2 The Bidder shall be income tax assesses and accordingly the Bidder shall submit copy of Income Tax Return (ITR) filed by the Bidder for the last three financial years.
- 3.9.3 A preliminary design and detailed schedule of detailed design, supply, installation, testing and commissioning is required to be submitted.

3.9.4 The details of manufacturing yard (work facility duly approved by IR Class / any other IACS body) as the proof of adequate capacity available for the work which must include the size, capacity, and facilities etc, is required to be furnished along with the registration and survey certificate (if any) from the concerned organization.

4 Pre-bid Meeting

A Pre-Bid meeting in hybrid mode shall be held as per the date and time mentioned in Section III - Data Sheet. Bidders willing to attend the pre-bid should inform the employer beforehand in writing and email. The maximum no. of participants, who chose to attend the pre-bid meeting, shall not be more than two per bidder. The representatives attending the pre-bid meeting must carry an authority letter duly signed by the authorized signatory of his/her organization permitting the representatives to attend the pre-bid meeting on behalf of the respective bidder.

During the course of Pre-Bid Meeting, the Bidders will be free to seek clarifications and make suggestions for consideration by the Employer. The Employer will endeavour to provide clarifications and such further information as it may, in its sole discretion, consider appropriate for facilitating a fair, transparent and competitive selection process.

The bidders may put forth their pre-bid queries in the format prescribed in Form 4J, Section IV.

The link of the pre bid meeting is as follows:

https://teams.microsoft.com/l/meetup-

join/19%3ameeting MmQ0Mzg5MDItNjI5Yy00ODQ1LWIxZjUtZTFlNWQ0NW ViMWZj%40thread.v2/0?context=%7b%22Tid%22%3a%22c4d675de-e1da-4ab4-ac52-3299a6812ab9%22%2c%22Oid%22%3a%22823c1ccb-6a8f-4855-9dc6-45316b8dcd88%22%7d

Meeting ID: 411 590 695 795 1 Passcode: EX72p3m8

5 Clarifications and Addendums

- 5.1 Bidders may request a clarification on any clause of the document up to the number of days indicated in Section III: Data Sheet before the Bid submission date. Any request for clarification must be sent in writing, or by e-mail to the Employer"s address indicated in Section III: Data Sheet. No request for the clarification shall be entertained if such request is received by the client after the deadline for submitting clarifications.
- 5.2 The Employer will respond in writing or by e-mail and will send written copies of the response (including an explanation of the query but without identifying the

source of query) to Bidders. Should the Employer deem it necessary to amend the Tender Document as a result of a clarification, it shall do so following the procedure mentioned hereunder:

(i) At any time before the submission of Bids, the Employer may amend the Tender Document by issuing an addendum/corrigendum (amendment) in writing or by email. The information of issue of such amendment will be uploaded on the website and will be binding on the Contractor. The Contractors shall acknowledge receipt of all amendments. However, failure of any contractor to furnish such acknowledgement shall not by itself take away the binding nature of the any / all amendments. To give Contractors reasonable time to the bidders to take an amendment into account, the Employer may, if the amendment is substantial, extend the deadline for the submission of Bids. The amendment /clarification, if any, to the document will be available on https://eprocure.gov.in/eprocure/app and IWAI"s website "www.iwai.nic.in" All bidders participating in the bid shall be deemed to have kept them informed and updated about each such amendment / clarification which is posted on the above website from time to time.

6 Preparation of Bids

In preparing their Bid, Bidders are expected to examine in detail the documents comprising the Tender Document. Material deficiencies in providing the requested information may result in rejection of the Bidder's Bid. Bidders shall adhere to the requirements mentioned below:

6.1 EMD

Bidders shall furnish EMD of the required amount for "Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry" as mentioned in Section III Data Sheet. As this is a composite work contract, exemptions to MSME is not allowed. In case of a JV, the EMD shall be furnished by the Lead Member.

EMD for the amount mentioned above shall be deposited to IWAI Fund through RTGS in the following account:

i.)	Name of Bank Account:	IWAI FUND
ii.)	Bank Name and Address	Union Bank of India, Sector 15, Noida
iii.)	Bank Account Number	513202050000007
iv.)	IFSC Code	UBIN0551325

6.1.2 The Earnest Money of the successful Bidder submitted in the form of RTGS will be retained as Security Deposit.

- 6.1.3 Bids submitted without EMD shall be treated as non-responsive.
- 6.1.4 No interest shall be payable by the Employer for the sum deposited as earnest money deposit.
- 6.1.5 The EMD of the unsuccessful bidders would be returned within one month of signing of the contract with the successful bidder.
- 6.1.6 The Earnest Money of the successful Bidder submitted in the form of RTGS will be retained as Security Deposit.
- 6.1.7 The EMD shall be forfeited by the Employer in the following events:
 - (i) If Bid is withdrawn during the validity period or any extension agreed by the Bidder thereof.
 - (ii) If the Bid is varied or modified in a manner not acceptable to the Employer after opening of Proposal during the validity period or any extension thereof.
 - (iii) If the Bidder tries to influence the evaluation process.
 - (iv) If the Bidder withdraws his proposal during negotiations.
 - (v) In case the Bidder, submits false certificate in terms of any documents supported to such Tender.
 - (vi) If the Bidder fails to sign the contract in accordance with Conditions of Contract on receipt of award of work
 - (vii) In case of a Bidder revoking or withdrawing his Tender or varying any terms in regard thereof without the consent of the Employer in writing.
 - (viii) In case the bidder is found to indulge in corrupt or fraudulent practices at any stage of the execution of the contract, as described in Clause 16.5 of GCC.
 - (ix) In case the bidder fails to furnish the prescribed performance guarantee within the prescribed period as mentioned in clause 4.4.1 of GCC
 - (x) In case of forfeiture of earnest money, as prescribed in vii and ix above, the tenderer shall not be allowed to participate in the retendering process of the work.

6.2 Cost of Tender Document/Tender Fee

All Bidders are required to pay the cost of Tender Document separately for design, construction, supply, Testing and Commissioning of 50 pax hybrid electric passenger ferry as mentioned in Section-III, Clause 8 of Data Sheet through RTGS/NEFT to be deposited in the following account. The cost of Tender Document is Non-Refundable.

i) Name of Bank Account: IWAI FUND

ii) Bank Name and Address: Canara Bank, Sector-18, Noida Branch Morna

iii) Bank Account number: 90622150000086

iv) IFSC: CNRB0018778

6.3 Bank Solvency

All bidders shall submit bank solvency certificate from a nationalized/scheduled bank in India for the amount as mentioned in Section III: Data Sheet.

The solvency certificate submitted by the bidder shall not be older than one (01) year from the Bid Submission Last Date. In case bidder does not adhere to this criterion, his bids shall be considered non-responsive and shall not be considered for further evaluation process. The bank solvency certificate shall be from any Nationalized/Scheduled Bank in India in the name of the bidder. The bank solvency certificate can be submitted by any one member of the JV / Consortium and the name of that member submitting the bank solvency certificate shall be mentioned in the JV / Consortium Agreement.

6.4 Taxes

The Bidders shall fully familiarize themselves with the applicability of all types of taxes and all such taxes, as prevailing on date of submission of the bid, must be included by the Bidder in the Financial Proposal along with the conditions mentioned therein, except for GST which will be quoted separately by the Bidder as per Form Fin -2. It may be noted that the bidder shall have to be registered with GST and shall submit the proof of the same. The GST and all other relevant taxes shall be as per existing rules and regulations at the time of payment.

6.5 Currency

Bidders shall quote the price of their Assignment/job in Indian Rupees.

6.6 Language

The Bid as well as all related correspondence exchanged between the Bidders and the Employer shall be in English language and shall be strictly as per the formats attached in this Tender Document. The Employer will evaluate only those Bids that are received in the specified formats and are complete in all respects as per this Tender Document. Any supporting documents submitted by the Bidder with its Bid or subsequently, in response to any query/ clarification from the Employer shall be in English and in case any of these documents is in another language, it must be accompanied by an accurate translation of all the relevant passages in English, and in such case, for all purposes of interpretation of the Bid, the translation in English shall prevail.

6.7 Bid Validity

The Section – III: Data Sheet indicates for how long the Bids submitted by the bidders must remain valid after the submission date. During this period, Bidders shall maintain the availability of Professional staff nominated in the Bid and also the amount quoted for the services in the Financial Bid shall remain unchanged. Should the need arise; however, the Employer may request Bidders to extend the validity period of their Bids. Bidders who agree to such extension shall confirm that they will maintain the availability of the Professional staff Proposed in the Bid and that their financial Bid will remain unchanged. Also, in their confirmation of extension of validity of the Bid, bidders could submit new staff in replacement, which would be considered in the final evaluation for contract award. The bidders have the right to refuse to extend the validity of the bids and bids of such bidders, who do not extend the validity of their bids, shall not be considered for further evaluation.

6.8 Number of Bids

A bidder can submit one bid only either as a single entity or in the form of a JV/consortium. In case a bidder submits or participates in more than one bid, the application of the bidder shall be rejected summarily.

6.9 Bids by Joint venture/consortium

- 6.9.1 The Joint Venture/consortium can be entered between two firms only.
- 6.9.2 The lead member should have highest share of participation in a JV/consortium.
- 6.9.3 The minimum percentage share of the 2nd member shall be 25%, with total share of all the JV/consortium members being 100%.
- 6.9.4 There shall be a Joint Venture Agreement/ Memorandum of Understanding specific for the contract between the constituent firms, indicating clearly, amongst other things, the proposed distribution of responsibilities both financial as well as technical for execution of the work amongst them. A copy of the Joint Venture Agreement/ MOU in accordance with requirements mentioned shall be submitted along with the bid. The bidder has to submit documentary proof of "intent of forming JV/consortium" on Rs.100/- registered stamp paper at the time of submission of bid. The bidder is required to submit the Letter of Intent to enter into a JV / Consortium agreement should contain at least the following:
 - Name of the Lead Partner
 - Clearly mentioned Percentage share of JV / Consortium members adhering to the clause 6.9.3 mentioned above.
 - "All the partners shall jointly and severally liable for the execution of the Contract in accordance with the Contract terms"

However, successful bidder is required to submit proof of registration of JV/consortium after issue of Letter of Intent and before signing of agreement. The members of JV / Consortium shall incorporate a special purpose vehicle as Contractor under the provisions of Companies Act, 2013, to execute the Project, if awarded to the JV/Consortium.

- 6.9.5 Lead partner's authorization shall be evidenced by submitting a power of attorney, duly notarized, signed by the legally authorized signatories of all the partners /members of JV/consortium.
- 6.9.6 The Lead Partner shall be authorized to incur liabilities and to receive instructions for and on behalf of the partners of the Joint Venture, whether jointly or severally, and entire execution of the Contract (including payment) shall be carried out exclusively through the partner-in-charge. A copy of the said authorization shall be furnished in this Bid.
- 6.9.7 In the event of default by any partner, in the execution of his part of the Contract, the Employer shall be so notified within 30 days by the Lead Partner, or in the case of the Lead Partner being the defaulter, by the partner nominated as partner-in-charge of the remaining Joint Venture/consortium. The partner-in-charge shall, within 60 days of the said notice, assign the work of the defaulting partner to any other equally competent party acceptable to the Employer to ensure the execution of that part of the Contract, as envisaged at the time of bid. Failure to comply with the above provisions will make the Contractor liable for action by the Employer under the Conditions of Contract. If the Most experienced i.e. Lead Partner, defined as such in the Communication approving the qualification, defaults, it shall be construed as default of the Contractor and Employer will take action under the Conditions of Contract.
- 6.9.8 Notwithstanding the permission to assigning the responsibilities of the defaulting partner to any other equally competent party acceptable to the Employer as mentioned in sub clause 6.9.7 above, all the partners of the Joint Venture/consortium will retain the full and undivided responsibility for the performance of their obligations under the Contract and/ or for satisfactory completion of the Works.
- 6.9.9 The bid submitted shall contain all relevant information for each member of JV/consortium as per the requirement stipulated under clause 10.1 of ITB
- 6.9.10 Lead member should have stake in the Joint Venture / consortium as stipulated in Clause 6.9.3 of Instructions to Bidders (ITB). However, the JV / Consortium members together shall meet the overall qualification Criteria stipulated in Clause 16.1 of ITB.
- 6.9.11 For sustainability of JV, clause 23 of GCC shall be referred.

7 Conflict of Interest

- 7.1 Employer requires that selected bidder (Contractor) provides professional, objective, and impartial advice and at all times holds the Employer's interest is paramount, strictly avoids conflicts with other assignment(s)/job(s) or his own corporate interests and act without any consideration for future work.
- 7.2 Without limitation on the generality of the foregoing, Bidders, and any of their affiliates, shall be considered to have a conflict of interest and shall not be recruited, under any of the circumstances set forth below:-
 - (a) Conflicting Activities: A firm or any of its affiliates who have been engaged by the Employer to provide goods, works or assignment/job other than design and build assignment/job for a project shall be disqualified from providing design and build assignment/job related to those goods, works or assignments/ jobs. Conversely, a firm or any of its affiliates who have been hired to provide design and build assignment/ job for the preparation or implementation of a project, and any of its affiliates, shall be disqualified from subsequently providing goods or works or assignment/job other than design and build assignment/job resulting from or directly related to the firms design and build assignment/job for such preparation or implementation. For the purpose of this paragraph, assignment/ job other than design and build assignment/job are defined as those leading to a measurable physical output; for example, surveys, exploratory drilling, aerial photography, satellite imagery etc.
 - (b) Conflicting assignment/ job: A Contractor {including its Personnel and Sub- Contractor(s)} or any of its affiliates shall not be hired for any assignment/ job that by its nature may be in conflict with another assignment/ job of the Contractor to be executed for the same or for another Employer, for example a Contractor hired to prepare engineering design for an infrastructure project shall not be engaged to prepare an independent environmental assessment for the same project and a Contractor assisting an Employer in the privatization of public assets shall not purchase nor advise purchasers of such assets.
 - Conflicting Relationships: A Contractor (including its Personnel and Sub-Contractors) that has a business or family relationship with a member of the Employer's staff who is directly or indirectly involved in any part of (i) the preparation of the Terms of Reference of the assignment/job (ii) the selection process for such assignment/job or (iii) supervision of the Contract, may not be awarded a Contract unless the conflict stemming from this relationship has been resolved in a manner acceptable to the Employer throughout the selection process and the execution of the Contract.

- 7.3 Contractors have an obligation to disclose any situation of actual or potential conflict that impacts their capacity to serve the best interest of their Employer, or that may reasonably be perceived as having this effect. Any such disclosure shall be made as per the standard forms of technical proposal provided herewith. If the Contractor fails to disclose said situations and if the Employer comes to know about any such situation at any time, it may lead to the disqualification of the Contractor during bidding process or termination of its Contract during execution of assignment on this ground alone.
- No agency or current employees or consultants of the Employer shall work as Contractors under their own ministries, departments, or agencies. The contract is liable for cancellation if either the Contractor himself or any of his employees or representatives are found to be persons/person who have held class I post under IWAI immediately before retirement and has within two years of such retirement accepted without obtaining the previous permission of IWAI, or the chairman as the case may be, and employment as contractor, or in connection with the execution of the public works, or as an employee of such contractor. If the contract is terminated on account of the failure of the contractor to comply with this clause, IWAI shall be entitled to recover from him such damages as my be determined by the Engineer in Charge with due regard to the inconvenience caused to IWAI on account of such termination without prejudice to IWAI's right to proceed for legal action against such officer.

8 Acknowledgement by Bidder

It shall be deemed that by submitting the Bid, the Bidder has:-

- 8.1 Made a complete and careful examination of this Tender for **Design**, **Construction**, **Supply**, **Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry** received all relevant information requested from the Employer;
- 8.2 Satisfied itself about all matters and necessary information required for submitting a competitive bid; inter-alia including fully familiarizing itself with the site and other conditions prevailing at site.
- 8.3 Updated itself about any amendments / clarifications that have been posted on the website in terms of clause 5.2 above.
- 8.4 Acknowledged that he does not have a Conflict of Interest; and
- 8.5 Agreed to be bound by the undertaking provided by him under the terms and conditions laid in this tender document.

9 Guidelines for e-submission of the Bids

9.1 The Bids should be submitted through Central Public Procurement Portal for e-Procurement https://eprocure.gov.in/eprocure/app.

- 9.2 Possession of valid Digital Signature Certificate (DSC) and enrolment/registration of the Contractors/Bidders on the e-procurement/e-Tender portal is a prerequisite for e-Tendering.
- 9.3 Bidder should enroll in the e-Procurement site using the https://eprocure.gov.in/eprocure/app option available "Enrol Here" on the home page portal. Enrolment is free of charge. During enrolment/registration, the Bidders should provide the correct/true information including valid e-mail id. All the correspondence shall be made directly with the Contractors/Bidders through email id provided.
- 9.4 Bidders need to login to the site through their user ID/password chosen during enrolment/registration.
- 9.5 Then the Digital Signature Certificate (Class II or Class III Certificates with signing key usage) issued by SIFY/TCS/node/eMudra or any Certifying Employer recognized by CCA India on e-Token/Smart Card, should be registered.
- 9.6 Only the registered DSC should be used by the Bidder and should ensure safety of the same.
- 9.7 Contractor/Bidder may go through the Tenders published on the site and download the required Tender documents/schedules in which the Bidder is interested.
- 9.8 After downloading/getting the Tender document/schedules, the Bidder should go through them carefully and then submit the documents as asked.
- 9.9 If there are any clarifications, this may be obtained online through the Tender site, or through the contact details as specified in Section III: Data Sheet. The Bidder should also take into account the addendum/corrigendum published before submitting the Bids online.
- 9.10 Then the Bidder may log into the site through the secured log in by giving the user id/ password chosen during enrolment/registration and then by giving the password of the e-Token/Smart card to access DSC.
- 9.11 Bidder selects the tender which he/she is interested in by using the search option & then moves it to the "my favorites" folder.
- 9.12 From the favorite's folder, he selects the tender to view all the details indicated.
- 9.13 It is construed that the bidder has read all the terms and conditions before submitting their offer. Bidder should go through the tender schedules carefully and upload the documents as asked; otherwise, the bid will be rejected.
- 9.14 Bidder, in advance, should get ready the Bid documents to be submitted as indicated in the Tender document/schedule and generally, they can be in general PDF/XLS/RAR/JPG formats. If there is more than one document, they can be

- clubbed together and can be provided in the requested format. Each document to be uploaded online should be less than 2 MB. If any document is more than 2MB, it can be reduced through Zip/RAR and the same can be uploaded, if permitted.
- 9.15 The Bidders can update well in advance, the documents such as certificates, annual report details etc., under My Space option and these can be selected as per Tender requirements and then sent along with Bid documents during Bid submission. This will facilitate the Bid submission process making it faster by reducing upload time of Bids.
- 9.16 Bidder should submit the Cost of Tender document/ EMD for the amount as specified in Section –III: Data Sheet. The original payment instruments should be posted/couriered/given in person to the employer within the due date as mentioned in this Tender document. Scanned copy of the instrument should be uploaded as part of the offer, if asked for.
- 9.17 While submitting the Bids online, the Bidder should accept the Terms & Conditions and proceed further to submit the Bid packets.
- 9.18 The Bidder has to select the payment option as offline to pay the Cost of Tender Document/ EMD as applicable and enter details of the instruments.
- 9.19 The details of the DD/any other accepted instrument, physically sent, should tally with the details available in the scanned copy and the data entered during Bid submission time. Otherwise submitted Bid will not be acceptable.
- 9.20 The Bidder has to digitally sign and upload the required Bid documents one by one as indicated. Bidders to note that the very act of using DSC for downloading the Bids and uploading their offers shall be deemed to be a confirmation that they have read all sections and pages of the Bid document including conditions of contract without any exception and have understood the entire document and are clear about the requirements of the Tender requirements.
- 9.21 The Bidder has to upload the relevant files required as indicated in the cover content. In case of any irrelevant files, the Bid will be automatically rejected.
- 9.22 If the price Bid format is provided in a spread sheet file like BoQ_xxxx.xls, the rates offered should be entered in the allotted space only and uploaded after filling the relevant columns. The Price Bid/BOQ template must not be modified/replaced by the Bidder; else the Bid submitted is liable to be rejected for this Tender.
- 9.23 The Bidders are requested to submit the Bids through online e-tendering system to the Tender Inviting Authority (TIA) well before the Bid submission end Date &Time (as per Server System Clock). The TIA will not be held responsible for any sort of delay or the difficulties faced during the online submission of Bids by the Bidders at the eleventh hour.

- 9.24 After the Bid submission, the acknowledgement number, given by the e-tendering system should be printed by the Bidder and kept as a record of evidence for online submission of Bid for the particular Tender and will also act as an entry pass to participate in the Bid opening date.
- 9.25 The Bidder should ensure/see that the Bid document submitted is free from virus and if the documents could not be opened, due to virus, during Tender opening, the Bid is likely or liable to be rejected.
- 9.26 The time settings fixed in the server side & displayed at the top of the Tender site, will be valid for all actions of requesting, Bid submission, Bid opening etc., in the e-Tender system. The Bidders should follow these time settings during Bid submission.
- 9.27 All the data being entered by the Bidders would be encrypted using PKI encryption techniques to ensure the secrecy of the data. The data entered will not viewable by unauthorized persons during Bid submission & not be viewable by any one until the time of Bid opening.
- 9.28 Any Bid document that is uploaded to the server is subjected to symmetric encryption using a system generated symmetric key. Further this key is subjected to asymmetric encryption using buyers/Bid openers" public keys. Overall, the uploaded Tender documents become readable only after the Tender opening by the authorized Bid openers.
- 9.29 The confidentiality of the Bids is maintained since the secured Socket Layer 128 bit encryption technology is used. Data storage encryption of sensitive fields is done.
- 9.30 The Bidder should logout of the Tendering system using the normal logout option available at the top right hand corner and not by selecting the (X) exit option in the browser.
- 9.31 Any queries relating to the Tender document and the Terms and Conditions contained therein should be addressed to the Tender Inviting Employer for a Tender or the relevant contact person indicated in the Tender.

10 Submission of Bids

The scanned copy of the Technical and Financial Bids, complete in all respects for "Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry" should be submitted. Bids shall be submitted in two covers. The Bidders are further advised to number all the pages and prepare a table of contents in the beginning of each Part referring the page numbers of the indexed items.

The Technical and Financial Bids, complete in all respects, should be submitted as per sequence mentioned below. Bids should be submitted in Two Covers:

10.1 Cover-I: Technical Bid

10.1.1 Enclosure – I

- a. Scanned copy of the proof of Cost of Tender Document as specified in Section III: Data sheet (The Original payment instruments should be posted / couriered / given in person so as to reach to the Employer on or before the Bid closing date and time.)
- b. Scanned copy of the proof of EMD as specified in Section III: Data Sheet and Annex VI (if submitted in the form of BG) (The Original payment instruments should be posted / couriered / given in person so as to reach to the Employer on or before the Bid closing date and time.)
- c. Proof of bank solvency for the amount as specified in Section III: Data Sheet
- d. Power of Attorney for the authorized person of the bidder as per Form 4D. This form shall be accompanied by copy of company identity card or general identity card (passport/Driving license/Voter's ID etc.) of the authorized representative.
- e. Power of Attorney for lead member of the JV/Consortium as per Form 4K
- f. Joint Bidding Agreement as per Form 4M
- g. Statement of legal Capacity as per Form 4L
- h. Scanned copy of Tender Acceptance Letter duly filled and signed by the authorized signatory of the Bidder as per Annex V
- i. Scanned copy of Form of Tender (Form 4A)
- j. Scanned copy of a signed declaration by the bidders (Form 4G)
- k. Bidders Information Form (Form 4H)
- 1. Composition/Ownership/Shareholding pattern of the organization
- m. Board Resolution, details of top management (Board members), key officials with documentary evidence, Articles of Association/memorandum of association of the company.
- n. Registration/incorporation certificate of the company/Firm.
- o. Original tender document with all addendums and corrigendum issued till date duly stamped and signed by the authorized signatory of the bidder.

10.1.2 Enclosure – II

- a. Scanned copy of the Annual Report/Audited balance sheets, for the last three financial years ending 31st March of the previous financial year i.e. 2021-22, 2022-23 and 2023-24.
- b. Scanned copy of GST Registration certificate.
- c. Scanned copy of PAN card of the Bidder.
- d. Form 4C of Section IV for average annual turnover with profit & loss statement
- e. Scanned copy of Bank account details, along with a cancelled cheque, for transaction through e-payment in format given at Annex IV.
- f. Integrity agreement in format given at Annex-VIII.

10.1.3 Enclosure – III

Scanned copy of complete Company profile with details such as:

- a. Background of the organization.
- b. Copies of completion certificate on client letter head for similar projects executed by the bidder in last seven years. The submitted certificates shall comply to conditions laid in clause 3 of ITB (Bidder Eligibility Criteria) Such eligible projects shall be supplied in Form 4B of Section IV.
- c. Copies of work order/agreement with value and status (% completed till submission) in case of on-going work shall be submitted separately as proof of on-going assignments as per Form- 4F of Section IV. The bidder shall also submit, along with Form 4F, plan/provision to move the existing machinery to the project site when required.
- d. General Construction experience of the bidder to be submitted as per Form 4N
- e. List of litigation history.

10.1.4 Enclosure – IV

- a. Scanned copies of following, keeping in view the scope of work listed in the ToR:
 - (i) The approach to the work and methodology to be adopted, and
 - (ii) Detailed work plan
 - (iii) List of Equipment to be deployed as per Form 4I of Section IV (Details of make and manufacture to be provided).
- b. Details of shipyard and its approval by IR Class / IACS body.

It may be noted that the Technical Bid shall not contain any reference to any fee or charges. In case, any reference to fee or charges is found in Technical bid, the bid will be summarily rejected.

All the submissions enumerated under Enclosure II & III shall be submitted by all the JV Partners separately.

10.2 Cover-II: Financial Bid

Financial Bid in excel format (BoQ) provided along with this Tender as Form Fin – 2 shall be used for quoting prices/offer.

- (i) This will contain fixed price contract rate to be charged for completing the work.
- (ii) While working out the price, following points should be noted:
 - (a) The Contractor will have to bear the cost of mobilization of equipment to the floating terminal site and their demobilization as required. All other incidental cost during the work is also to be borne by the Bidder.
 - (b) The Contractors will have to make their own arrangements for the transport/accommodation/TA/DA of their personnel assigned to this project. The price quoted shall also include the Contractor's representative visit to various offices, and other places for meetings, data collection, presentations, public consultation, secretarial staff, their salary, allowances, overhead expenditure etc.
 - (c) All duties, taxes, royalties, and other levies payable by the Contractor under the Contract, or for any other cause, shall be included in the rates, prices, and total Bid price submitted by the Bidder except for GST which will be as per existing rules and regulations at the time of payment. The rates and prices quoted by the Bidder shall be fixed for the duration of the Contract and shall not be subject to any adjustment other than that defined in clause 5.2 Section VI, Part I. The prices shall be quoted by the Bidder entirely in Indian Rupees. All payments shall be made in Indian Rupees (INR).
- 10.3 The total duration of Works shall be as specified in Section III: Data Sheet.

11 Extension of Bid submission date

The employer may extend the date of submission of bids by issuing an addendum and uploading the same in website.

12 Late Proposals

Proposals received by the employer after the specified bid submission date or any extension thereof, pursuant to clause 11, shall not be considered for evaluation and shall be summarily rejected.

13 Liability of the employer

The bidders are advised to avoid last moment rush to submit bids online and they should upload their bids well in advance before the bid submission deadline. The employer shall not be liable for failure of online submission of bids by the bidder that may arise due to any reason whatsoever. It shall be construed that the guidelines for online submission of bids, mentioned under clause 9 of ITB, have been read and understood by the bidder.

14 Modification/Substitution/Withdrawal of Bids

The tender for works Design, Construction, Supply, Testing And Commissioning Of 50 Pax Hybrid Electric Passenger Ferry once submitted may be modified, substituted, or withdrawn by the bidders before the submission deadline through e-Procurement Mode as mentioned in NIT of this tender document.

No bid shall be modified after the deadline for submission of bids.

15 Bid opening and evaluation process

- 15.1 From the time the Proposals are opened to the time the Contract is awarded, the Bidders should not contact the Employer on any matter related to its Technical and/or Financial Proposal. Any effort by Bidders to influence the Employer in the examination, evaluation, ranking of Proposals, and recommendation for award of Contract may result in the rejection of the Bidders" Proposal."
- The employer will constitute a tender evaluation committee (TEC) which will carry out the evaluation process.
- Online Bid opening shall be carried out in two stages. Firstly, 'Technical Bid' of all the online Bids received shall be opened on the date and time mentioned in Section III: Data Sheet. 'Financial Bid' of those Bidders whose Technical Bid has been determined to be responsive and on evaluation fulfils the criteria as stipulated in the Tender Document, shall be opened on a subsequent date, which will be notified to such Bidders. In the event of the specified date for the submission of bids being declared a holiday for The Employer, the Bids will be opened at the defined time and location on the next working day. Bids for which a notice of withdrawal has been submitted shall not be opened.
- The TEC shall evaluate the Technical Bids on the basis of their responsiveness to the Terms of Reference and by applying the evaluation criteria specified in Clause 16.

Bid shall be rejected if it is found deficient or found not meeting the eligibility criteria as mentioned in clause 3 and clause 16.1 of ITB. Only responsive bids shall be further taken up for evaluation. A Bid shall be considered responsive only if:

- 15.4.1 It is received by the Bid submission date and time including any extension thereof, pursuant to Clause 11;
- 15.4.2 It is accompanied by the Earnest Money Deposit a signed **Bid security declaration** as specified in Clause 6.1;
- 15.4.3 It is received in the forms specified in section IV (Technical Proposal) and in section V (financial proposal);
- 15.4.4 It does not contain any condition or qualification
- The Employer shall inform the Bidders, whose Technical Bids are found responsive and on evaluation fulfils the criteria stipulated in the Tender document, of the Date, Time and Place of opening of the Financial Bids. The Bidders so informed, or their representative, may attend the meeting of online opening of Financial Bids.
- At the time of the online opening of the "Financial Bids", the names of the Bidders, whose Bids were found responsive and qualified for opening of financial bid, the total amount of each Bid, and such other details as the Employer may consider appropriate will be announced by the Employer at the time of Bid opening.
- 15.7 Bidder may, if deemed necessary by him, send a representative to attend the financial bid opening. Such representative shall have a letter of authorization from the bidder to attend the bid opening on its behalf. In case a bidder sends a representative on its behalf to attend the financial bid opening, it shall be construed that the representative has the authority of the bidder.

16 Qualification Criteria and Bid evaluation

16.1 Qualification criteria

To qualify for the tender for **Design**, **Construction**, **Supply**, **Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry**, the Bidder must satisfy each of the qualifying criteria stipulated in Clauses 16.1.1 to 16.1.5 of ITB. Not satisfying any of the qualification criteria will render the bid non-responsive and financial bids of such bidders will not be opened.

16.1.1 Qualification for works

The bidder should have successfully completed "Similar Works" in previous 7 years before the last date of the Bid submission as per the criteria specified below:

a) 3 similar works each costing not less than **Rs. 6.32 Cr.**

- b) 2 similar works each costing not less than Rs. 7.90 Cr.
- c) 1 similar work each costing not less than **Rs. 12.65 Cr.**

Note: The value of the "Completed Work(s)" considered by the Bidders shall be rounded off to the nearest two digits.

For this purpose, the "Similar Works" means Fabrication, Transportation, Installation, Testing and Commissioning of inland vessels / sea going vessels / propelled / non propelled dredgers / electric catamarans / other propelled green vessels duly certified by any classification society.

Further, substantially completion of latest shall be as per Clause 3.2 of ITB.

In the event of a Joint Venture, all Parties combined shall together meet the technical eligibility as well as financial eligibility requirement.

Further, the value of contracts completed by its members shall not be aggregated to determine whether the requirement of the minimum value of a single contract has been met. Instead, each contract performed by each member shall satisfy the minimum value of a single contract as required for a single entity. In determining whether the JV meets the requirement of total number of contracts, only the number of contracts completed by all members, each of value, equal or more than the minimum value required shall be aggregated.

In case a bidder intends to sub-contract the work, submission of Letter of Consent signed by the sub-contractor is mandatory. The format for the Letter of Consent from the sub-contractor shall be as per Annexure-VII of Section IX: Annexes of the Tender Document. The work claimed by bidder / proposed sub-contractor shall be substantiated with client certificates as stipulated in clause 3.2 of ITB.

16.1.2 Qualification Criteria for Average annual turnover for last 3 financial years i.e. 2021-22, 2022-23 and 2023-24 (for works Design, Construction, Supply, Testing And Commissioning Of 50 Pax Hybrid Electric Passenger Ferry separately)

Rs. 4.74/- Cr.

In the event of a Joint Venture, all parties combined shall together meet the Qualification requirement of "Average Annual Turnover".

- 16.1.3 Qualification Criteria for Equipment and experience
 - i. For minimum eligibility pertaining to equipment, the Bidder may refer clause 8 of ToR.
 - ii. Details of similar works (Clause 16.1.1 of ITB) executed in last 7 years including type of vessel designed and constructed, quantity, scheduled period of execution, date of commencement and date of completion, substantial

completion, and certificate/testimonials from the concerned clients regarding successful completion / substantial completion (more than 75% complete only) of the job to substantiate technical capacity of the Bidder required as per Clause 3 and 16.1.1 of ITB.

16.1.4 Qualification Criteria for Bid capacity (for works listed in tender document)

- i. The Bidder shall submit the details asked for the Bid capacity as per the format prescribed in Form 4O.
- ii. The Bid Capacity of the Bidder must be equal to or more than the estimated cost of the work.
- iii. Work put to the Tender.
- iv. In case the Bid capacity of the Bidder is less than the estimated cost of the work, his bid shall be cancelled and such Bidder shall not be considered for opening of Financial Bid even if he has been determined eligible in other criteria's set forth in the Tender Document.
- v. In the event of a Joint Venture, all parties combined shall together meet the Qualification requirement of Bid Capacity.

16.1.5 Qualification Criteria for Work Plan and Methodology Statement

- i. The Bidder shall submit the work plan indicating the method of execution of work in detail and its completion timeline in bar chart format.
- ii. The Bidder should submit the salient details of its manufacturing facility (duly certified by any IACS body) and shall also furnish the details of manpower & equipment available with him, for execution of tendered work.

16.2 Bid evaluation

- 16.2.1 The Bids shall be evaluated based on the qualification criteria mentioned in Clause 16.1 of ITB. In case a bidder fails to meet the above-mentioned qualification criteria, their bids shall be treated as non- responsive and financial bids of such bidders shall not be opened.
- 16.2.2 A substantially responsive bid is one that conforms to all the terms, conditions, and specifications of the bidding document without material deviation or reservation. A material deviation or reservation is one:
 - (a) that affects in any substantial way the scope, quality, or performance of the Works;
 - (b) that limits in any substantial way, inconsistent with the bidding documents, the Employer's rights, or the bidder's obligations under the contract; or

- (c) whose rectification would affect unfairly the competitive position of other bidders presenting substantially responsive bids. Further, a bid will not be considered substantially responsive if the bidder has not offered to undertake all the items of the work as listed in Bill of Quantity (BoQ) of section V of bid document.
- 16.2.3 The estimated effect of the price adjustment provisions of the Conditions of Contract, applied over the period of execution of the Contract, shall not be taken into account while evaluating bids.
- The lowest priced bidder i.e. L-1 shall be chosen based on the amount quoted by the bidders in BoQ of Section V.
- 16.2.5 If the bid, which results in the lowest Evaluated Bid Price, is seriously unbalanced or front loaded as compared to the estimate of the items of work to be performed under the Contract, the bidder shall be asked to produce detailed price analyses for any or all items of the Bill of Quantities, to demonstrate the internal consistency of those prices with the construction methods and operating methodology proposed. After evaluation of the price analysis, taking into consideration the schedule of estimated Contract payments, price may be negotiated with the lowest bidder to remove the imbalance, make an appropriate adjustment on sound technical and/or financial ground for any quantifiable, acceptable aspects and thereby bring the prices to justified level sufficient to protect the Employer's interest, before taking a decision on the bid.

If during the execution of the works, deviation / variation takes place to the extent of amount exceeding more than 20% of the accepted contract value, the Contractor at the written request of Engineer-In-Charge, shall promptly increase the value of the Performance Bank Guarantee.

17 Award of Contract

- 17.1 The Employer shall issue a Letter of Award to the selected Bidder. It may also notify all other Bidders who submitted their bids about the decision taken.
- 17.2 The Contractor will sign the Contract Agreement after fulfilling all the formalities/pre- conditions mentioned in the Standard Form of Contract in Section VII, within 45 days of issuance of the Letter of Award.
- 17.3 The Contractor is expected to commence the Assignment/job on the date and at the location specified in Section III Data Sheet.

18 Ownership of Document and Copyright

All the deliverables and study outputs including primary data shall be compiled, classified, and submitted by the Contractor to the Employer in hard copies and

editable soft copies in addition to the requirements for the reports and deliverables indicated in the Terms of Reference.

The study outputs shall remain the property of the Employer and shall not be used for any purpose other than that intended under these Terms of Reference without the prior written permission of the Employer. In the case of any deliverables by Contractor consisting of any Intellectual Property Rights ("IPR") rights of the Contractor, the Contractor shall provide the Employer with necessary irrevocable royalty-free license to use such IPR. Further, for the avoidance of any doubt, it is clarified that any intellectual property developed during the course of, or as a result of, the services rendered in relation to the Works, shall be and shall remain property of the Employer.

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
SECTION - III: D	ATA SHEET
SECTION III. D	
3	33

DATA SHEET

CLAUSE NO. OF DATA SHEET	REF. OF ITB	PARTICULARS	DESCRIPTION
1.	-	Employer	The Chairman, Inland Waterways Authority of India (IWAI), A- 13, Sector- 1, Noida-201301
2.	2.2	Name of the Assignment/job	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
3.	2.1	Method of Selection	CBS (Cost Based Selection) – L1
4.	2.3	Date & time for submission of Bid	Date: 31.07.2025 Time: Latest by 15:00 Hrs (IST) Submission: Online Submission Address: Chief Engineer (NER & other NNW's), Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301
5.	4	A Pre-Bid meeting will be held on	Date: 11.07.2025 Time: 15:00 hrs. Venue: Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301
6.	5.1	Last date for seeking clarifications	Date: 10.07.2025 Email Id: mt.iwai@nic.in , ssingh.iwai@nic.in & srafat.iwai@nic.in
7.	6.1	EMD	INR 31.61 Lakhs
8.	6.2	Cost of Tender Document	INR 11,800/- (Rupees Eleven Thousand and Eight Hundred only) Including GST
9.	6.3	Bank Solvency	Rs. 6.32/- Cr.
10.	-	Estimated Cost (Cost put to tender)	INR 15.80 Crore excluding GST
11.	16.1.2	Average Annual Turnover	Rs. 4.74 Cr.
12.	6.7	Bid Validity	120 days after Bid submission date
13.	3.3	Similar Works	As stipulated in Clause 16.1.1 of ITB
14.	-	JV/consortium	Yes

CLAUSE NO. OF DATA SHEET	REF. OF ITB	PARTICULARS	DESCRIPTION
		allowed	
15.	-	The formats for the Technical Bid (Applicable for works listed)	FORM 4A: Form of Tender
			FORM 4B: Eligible Projects
			FORM 4C: Average Annual Turnover
			FORM 4D: Power of Attorney (for authorized representative of the bidder)
			FORM 4E: List of Key Personnel
			FORM 4F: List of Ongoing assignments
			FORM 4G: Declaration by the Bidders
			FORM 4H: Bidder Information Form
			FORM 4I: Availability of Vessel building facility including infrastructural facility, equipment, machineries etc to be used for tender work
			FORM 4J:Format for pre-bid queries
			FORM 4K: Power of Attorney for Lead Member of JV/Consortium
			FORM 4L: Statement of Legal Capacity
			FORM 4M: Joint Bidding Agreement
			FORM 4N: General Construction Experience
			FORM 40: Bid Capacity
			FORM 4P: Bid Security Declaration
16.	10.4	Contract Duration	06 months from the date of Signing of Contact Agreement
17.	15.3	Bid Opening date	Date: 01.08.2025 Time: 15:30 hrs Venue: Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
	36

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
SECTION-IV: TECHNICAL I	DID CTANDADD EODMC
SECTION-IV. TECHNICAL I	DID STANDARD FORMS

FORM 4A: FORM OF TENDER

(To be submitted on the letter head of the Bidder)

To

Chief Engineer (NER & other NNW's), Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301

Sub: Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry

Sir.

- 1. Having visited the site and examined the information and instructions for submission of tender, General Conditions of Contract (GCC), Special Conditions of Contract (SCC), Technical, General and Detailed specifications, Bill of Quantities (BoQ) agreement and Bank Guarantee forms, etc. for the above-named works, I/ We(Name of Bidder) hereby tender for execution of the works referred to in the tender document in conformity with the said GCC, SCC, Schedule of quantities for the sum as stated in BoQ of this tender document or such other sum as may be ascertained in accordance with the said conditions of contract.
- 2. I/ We undertake to complete and deliver the whole of the work comprised in the Contract within the time as stated in the tender and also in accordance with all respects with the specifications, Scope of work and instructions as mentioned in the tender document.
- 3. I am tendering for the works mentioned in the table below and submitting the EMD vide RTGS / BG in IWAI Account as per the details given therein:

RT	GS / NEFT	Bank	Total EMD (INR)	
Amount (INR)	Details of RTGS / NEFT (No. and Date) and details of the bank (Name of bank, branch, address)	Amount (INR)	Details of the Bank Guarantee (No and Date) and details of the bank (Name of bank, branch, address)	

- 4. I/ We agree to abide by this tender. I/ We agree to keep the tender open for a period of 120 days from the date of opening of technical bids or extension thereto as required by the IWAI and not to make any modifications in its terms and conditions.
- 5. I/ We agree, if I/ we fail to keep the validity of the tender open as aforesaid or I/ we make any modifications in the terms and conditions of my/ our tender if I/ We fail to commence the execution of the works as above, I/ We shall become liable for forfeiture of my/ our Earnest money, as aforesaid and IWAI shall without any prejudice to another right or remedy, be at the liberty to forfeit the said Earnest Money absolutely otherwise the said earnest money shall be retained by IWAI towards part of security deposit to execute all the works referred to in the tender document upon the terms and conditions contained or referred to therein and to carry out such deviations as may be ordered. Should this tender be accepted, I/ We agree(s) to abide by and fulfil all the terms and conditions and provisions of this tender. No interest is payable on earnest money deposit and/ or security deposit.
- 6. If We have independently considered the amount of Liquidated Damages shown in the tender here to and agree that it represents a fair estimate of the loss likely to be suffered by IWAI in the event of works not being completed in time.
- 7. If this tender is accepted, I/ We undertake to enter into execute at my/ our cost when called upon by the employer to do so, a contract agreement in the prescribed form. Unless and until a formal Agreement is prepared and executed this tender together with your written acceptance thereto shall constitute a binding contract.
- 8. If my/ our tender is accepted, I/We am/are to be jointly and severely responsible for the due performance of the Contract. I/We also declare that the firm has not been banned or blacklisted by any Govt. or its department or any Quasi Govt. agency or Public Sector Undertaking or Multilateral or International Aid Agency/Development Bank. Any such discovery by IWAI at any stage of the tender/contract may result in disqualification of the firm or cancellation of the contract.
- 9. If We understand that you are not bound to accept the lowest or any tender you may receive and may reject all or any tender without assigning any reason.
- 10. If We certify that the tender submitted by me, us is strictly in accordance with the terms, conditions, specifications etc. as contained in the tender document, and it is further certified that it does not contain any deviation to the aforesaid documents.

Date
Signature
Name
Designation

Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India) Duly authorized to sign & submit tender for an on behalf of (Name and address of firm) $M/s \ \dots \dots \dots \dots$ Telephone no's.....FAX No.....

FORM 4B: Eligible Projects

Format for Responsiveness of Bid (Eligible Projects) Project Specific Experience

[Using the format below, provide information on each assignment for which your firm, and each associate for this assignment, was legally contracted either individually or as a corporate entity or as one of the major companies within a Joint Venture (JV) for carrying similar works under this assignment.]

"Similar Works" means Fabrication, Transportation, Installation, Testing and Commissioning of inland vessels / sea going vessels / propelled / non propelled dredgers / electric catamarans / other propelled green vessels duly certified by any classification society

S. No.	Client Name, Name of work & location of project	Contract Value in INR Financial ³ value of similar work satisfactorily completed / substantially completed	Date of start of work	Scheduled completion date	Actual completion date	Details of work (including Similar Work)	Remarks

¹ Exhibit only those projects completed in the last Seven (7) years from the Last Date of Bid Submission.

4 Substantially completed projects as defined in Clause 3.2 of ITB shall be supported by client certificates with the name of the
work including component breakup of relevant work to the extent of similar work, work order value, start date, percentage of
similar work completed till date of issue of the Tender in Financial terms (Net Payment received against the Contract).

Firm	16	Name ·			
	•	INAIIIE			

² The Bidder shall submit proof of experience from the Client for meeting the minimum qualification details. The Client Certificate submitted by the Client shall contain the details as enlisted in Clause 3.2 of ITB. The works claimed by the Bidder, if not supported with proof of completion / substantial completion as laid down under Clause 3.2 of ITB and defined hereinafter in Footnote 4 below from the Client will not be considered.

³ Against the Contract of works having several components other than the Similar Works, only the relevant component shall be evaluated for contract value, payment value and the actual execution period for the relevant component only should be submitted / specified.

Authorized Sigr	nature :	
-----------------	----------	--

Notes:

- 1. For the purpose of evaluation, Bidders should assume 7% inflation for Indian Rupees every year and 2% for foreign currency portions per year compounded annually.
- 2. Bidders should mention the maximum value of similar works as defined in Clause
- 3. 16.1.1 of ITB executed during the last seven years (adjusted last day of the month previous to the one in which this Tender is invited).
- 4. In case of foreign currency, it should first be escalated at the rate mentioned above and then the amount so derived shall be converted to INR at the exchange rate prevailing last day of the month previous to the one in which this Tender is invited.
- 5. Exchange rate should be taken from official website of RBI (https://www.rbi.org.in/scripts/ReferenceRateArchive.aspx)
- 6. In case, exchange rate for the currency in consideration, is not available on RBI website (mentioned above), Bidders shall quote exchange rates from websites such as www.xe.com, www.oanda.com, along with copy of the exchange rate used by the Bidder for the conversion.
- 7. Any additional comments / information to substantiate that the said work conforms to the specified similar works can also be indicated by the bidder, as deemed fit.

Please limit the description of each project in two A4 size sheet of paper. Descriptions exceeding two (02) A4 size sheet of paper may or may not be considered for evaluation.

FORM 4C: Average Annual Turnover of the Bidder Along with Loss & Profit Statement

Name of Work: Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry

S. No.	Financial Years	Average Annual Turnover & loss & profit statement of Applicant (INR) in Last Three Financial Years
1.	2021-2022	
2.	2022-2023	
3.	2023-2024	
Average Annual Turnover		[indicate sum of the above figures divided by 3]

Certificate from the Statutory Auditor

This is to certify that	[Name of the Firm]	[Registered	Address] ha	as received	the
payments shown above against the	e respective years.				

Name of Authorized Signatory

Designation:

Name of Firm:

(Signature of the Statutory Auditor Seal of the Firm)

Note:

1. This Form shall be submitted on the letter head of Statutory Auditor.

FORM 4 D: Power of Attorney

(for authorized representative of the bidder)

(To be executed on Non-Judicial Stamp Paper of Rs.100 and duly notarized)

Know all men by these presents, We,(name of organization and address of the
registered office) do hereby constitute, nominate, appoint and authorise Mr. /Ms.
son/daughter/wife and presently residing at who is presently employed
with/retained by us and holding the position ofas our true and lawful attorney
(hereinafter referred to as the "Authorized Representative"), with power to sub-delegate to
any person, to do in our name and on our behalf, all such acts, deeds and things as are
necessary or required in connection with or incidental to submission of our Bid for Design,
Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger
Ferry (strike out whichever is not applicable). The selection of Contractor for Inland
Waterways Authority of India (the "Employer") including but not limited to signing and submission of all applications, Bids and other documents and writings, participating in pre-bid and other conferences and providing information/responses to the Employer, representing us in all matters before the Employer, signing and execution of all contracts and undertakings consequent to acceptance of our Bid and generally dealing with the Employer in all matters in connection with or relating to or arising out of our Bid for the said Project and/or upon award thereof to us till the entering into of the Contract with the Employer.
AND, we do hereby agree to ratify and confirm all acts, deeds and things lawfully done or
caused to be done by our said Authorized Representative pursuant to and in exercise of the powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Authorized Representative in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us.
powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Authorized Representative in exercise of the powers hereby conferred shall and shall always be
powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Authorized Representative in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us. IN WITNESS WHEREOF WE, THE ABOVE-NAMED PRINCIPAL HAVE EXECUTED
powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Authorized Representative in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us. IN WITNESS WHEREOF WE, THE ABOVE-NAMED PRINCIPAL HAVE EXECUTED THIS POWER OF ATTORNEY ON THIS
powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Authorized Representative in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us. IN WITNESS WHEREOF WE, THE ABOVE-NAMED PRINCIPAL HAVE EXECUTED THIS POWER OF ATTORNEY ON THISDAY OF, 2024 For
powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Authorized Representative in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us. IN WITNESS WHEREOF WE, THE ABOVE-NAMED PRINCIPAL HAVE EXECUTED THIS POWER OF ATTORNEY ON THIS

Accepted

(Signature, name, designation and address of the Attorney)

Notes:

- 1. The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, laid down by the applicable law and the charter documents of the executant(s) and when it is so required the same should be under common seal affixed in accordance with the required procedure.
- 2. Wherever required, the Applicant should submit for verification the extract of the charter documents and other documents such as a resolution / power of attorney in favour of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Applicant
- 3. For a Power of Attorney executed and issued overseas, the document will also have to be legalized by the Indian Embassy and notarized in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Applicants from countries that have signed The Hague Legislation Convention, 1961 are not required to be legalized by the Indian Embassy if it carries a conforming Apostille certificate.

FORM 4E: List of Key Personnel

Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry

S. No.	Position / Role	No.
1		
2		
3		
4		
5		
6		
7		
8		

Note:

- 1. The list of key personnel is tentative and indicative only. The Bidder can propose the key personal required for works as per the scope defined in the Tender Document. It may be noted that the key personnel proposed and the number of personnel required must also meet the statutory requirement.
- 2. During the tenure of contract if need arise, then the Bidder is allowed to change / replace the Key Personnel with a prior written approval from EIC.

FORM 4F: List of Ongoing Assignments

S. No.	Full Postal Address of Client & Name of officer -in- charge	Description of the work including type of vessel\ferry vessel with suitable floating terminal	Total Contract Value	Value of the work complete d till last date of the previous month from the Bid Submiss ion Date	Outstanding / Balance Value of the work till last date of the previous month from the Bid Submission Date	Date of commencement of work	Scheduled completion period	Average completion as on date	Expected date of completion

FORM 4G: Declaration by the Bidders

To,		Date:
	Chief Engineer (NER & other NNW's), Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301	

Kind Attention: Chief Engineer (NER & Other NNW's),

Sub: Declaration from the Bidder.

Tender Reference No:.....

Dear Sir,

This is with reference to the above-mentioned Tender document.

I/We hereby make the following declarations:

1.	No alteration has been made in any form in the downloaded Tender document.
2.	I/We have not been banned or de listed by any Government or quasi-Government agency or public sector undertaking.
3.	I/We accept the payment terms of clause 14.2 of General Conditions of Contract.
4.	I/We provide our acceptance to all Tender Te rms and Conditions.
5.	Acknowledgment by Bidder as per Clause 8 of ITB
6.	I/We confirm that neither we have failed nor we have been expelled from any project or agreement during the last 03 years
7.	I/We agree to disqualify us for any wrong declaration with respect to the submissions made by us for this tender and reject my/our tender summarily
8.	I/We agree to disqualify us from this tender and black list us for tendering in IWAI projects in future, if it comes to the notice of IWAI that the documents/submissions made by me/us are not genuine

Yours Faithfully (Signature of the Bidder, with Official Seal

FORM 4H: Bidders Information Form

Bidder name:
[insert full name]
Bidder's Party name:
[insert full name of Applicant's Party]
Bidder's Party country of registration:
[indicate country of registration]
Bidder's year of constitution:
[indicate year of constitution]
Bidder's legal address in country of constitution:
[insert street/ number/ town or city/ country]
Bidder's authorized representative information Name: [insert full name]
Address: [insert street/ number/ town or city/ country]
Telephone/Fax numbers: [insert telephone/fax numbers, including country and city codes]
E-mail address: [indicate e-mail address]
1. Attached are copies of original documents of
Articles of Incorporation (or equivalent documents of constitution or association), and/or registration documents of the legal entity named above
☐ In case of a Government-owned enterprise or institution, documents establishing legal and financial autonomy, operation in accordance with commercial law, and absence of dependent status
2. Included are the organizational chart, a list of Board of Directors, and the beneficial ownership.

Note:

This Form shall be supplied with Identity proof of the authorized representative

FORM 4I: Availability of Vessel Building Facility Including Infrastructural Facility, Equipment, and Machineries ETC to be used for Tender Work

N	ame & Address of the Ship Builder	Detailed Particulars
(i)	Building Dock / Slipway (if any) including fabrication/ construction bays (details including number, dimension, location and layout of shipyard to be given). Whether covered or open.	
	IACS approval certificate of work facility	
(ii)	Skids / Mould loft for modular construction (details including number, dimension and location to be given). Whether covered or open.	
(iii)	Design and drawing office of own or tie up with experienced Design Consultant/ Naval Architect with brief resume	
(iv)	List of workshop machinery, equipment.	Description Make Quantity
	 a) Welding Sets b) Gas Cutting Sets c) Grinding / Bending machines d) Shot Blasting equipment e) Material handling equipment f) Machine Shop Machineries g) Paint Shop h) Any other details 	
(v)	Source of electric power, whether captive power unit available, if so, the details thereof.	
(vi)	Facilities for doing outfitting job in afloat condition indicating location and area.	

Note:

- 1. If owned, the registration certificates and year of procurement for the proposed machinery should be appended with this Form
- 2. In case of hired equipment, the consent letter from the owner of the equipment is to be appended.
- 3. In case of equipment to be procured, status of timelines of procurement to be included.
- 4. If required, separate / additional sheets can be used.

(Signature of authorized representative)

FORM 4J: Format for Pre-Bid Queries by Bidders

(To be submitted in Bidders Letter head)

Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry

Name of Bidder:	Date of Submission:
-----------------	---------------------

Pre – Bid Queries

S. No.	Section No. Clause, Sub Clause No and Page No. of Tender	Tender clause description	Query
1.			
2.			
3.			
4.			
5.			
6.			
•			
•			

FORM 4K: Power of Attorney for Lead Member of JV/Consortium

(To be executed on Non-Judicial Stamp Paper of Rs.100 and duly notarized. For a Power of Attorney executed and issued overseas, the document will also have to be legalized by the Indian Embassy and notarized in the jurisdiction where the undertaking is being issued.)

Whereas it is necessary for the Members of the JV/Consortium to designate one of them as the Lead Member with all necessary power and authority to do for and on behalf of the JV/Consortium, all acts, deeds, and things as may be necessary in connection with the JV/Consortium's Bid for the Project and its execution.

NOW THEREFORE KNOW ALL MEN BY THESE PRESENTS

We,having our registered office at, M/s.
having our registered office at, M/s.
having our registered office at , and having our
registered office at, (hereinafter collectively referred to as the
"Principals") do hereby irrevocably designate, nominate, constitute, appoint and authorize
M/s having its registered office at, being one of
the Members of the JV/Consortium, as the Lead Member and true and lawful attorney of the
JV/Consortium (hereinafter referred to as the "Attorney"). We hereby irrevocably authorize
the Attorney (with power to sub-delegate) to conduct all business for and on behalf of the
JV/Consortium and any one of us during the bidding process and, in the event the
JV/Consortium is awarded the contract, during the execution of the Project and in this regard,
to do on our behalf and on behalf of the JV/Consortium, all or any of such acts, deeds or
things as are necessary or required or incidental to the pre-qualification of the JV/Consortium
and submission of its Bid for the Project, including but not limited to signing and submission
of all, Bid and other documents and writings, participate in bidders and other conferences,
respond to queries, submit information/ documents, sign and execute contracts and
undertakings consequent to acceptance of the Bid of the JV/Consortium and generally to
represent the JV/Consortium in all its dealings with the Authority, and/ or any other
Government Agency or any person, in all matters in connection with or relating to or arising
out of the JV/Consortium's Bid for the "The Work".

AND hereby agree to ratify and confirm and do hereby ratify and confirm all acts, deeds and things done or caused to be done by our said Attorney pursuant to and in exercise of the

powers conferred by this Power of Attorney and that all acts, deeds, and things done by our said Attorney in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us/ Consortium and shall be legally binding upon us/Consortium.

For	(Signature)	
	(Name & Title)	
For	(Signature)	
	(Name & Title)	
	(Signature)	
	(Name & Title)	
Witnesses:		
1.		
2.		
	(Executants)	
(To be execute	d by all the Members of the JV/Consor	tium)

Notes:

- The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, laid down by the applicable law and the charter documents of the executants and when it is so required, the same should be under common seal affixed in accordance with the required procedure.
- Also, wherever required, the Bidder should submit for verification the extract of the charter documents and documents such as a board or shareholders' resolution/power of attorney in favour of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Bidder.
- For a Power of Attorney executed and issued overseas, the document will also have to be legalized by the Indian Embassy and notarised in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Bidders from

and Waterways Authority of India nistry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry					
countries that have signed the Hague Legislation Convention 1961 are not required to						
be legalized by the Indian Embassy if i	t carries a conforming Apostil certificate.					

FORM 4L: Statement of Legal Capacity

(To be forwarded on the letterhead of the Bidder/ Lead Member of JV/Consortium)

Ref.	Date:
To,	
Chief Engineer (NER & other NNW's), Inland Waterways Authority of India (IWA A-13, Sector-1, Noida (U.P.)-201301	AI),
Dear Sir,	
We hereby confirm that we/ our members in the been described in the Bid) satisfy the terms and co	•
We have agreed that (insert Member of our JV/consortium. *	t member's name) will act as the Lead
We have agreed that (inserepresentative/ will act as the representative of the duly authorized to submit the tender document. With requisite powers to furnish such leactions/representations of the Lead Member/Authorithe Consortium.	ne consortium on its behalf* and has been Further, the authorized signatory is vested tter and authenticate the same. All
Thanking you,	
Yours faithfully,	
(Signature, name and designation of the au of	thorized signatory) For and on behalf
*Please strike out whichever is not applicable.	

FORM 4M: Joint Bidding Agreement

(To be executed on Stamp paper of appropriate value)

THIS	JOINT	BIDDING	AGREEMENT	is	entered	into	on	this	the	 day	of
	202	4									

AMONGST

1. {... Limited, a company incorporated under the Companies Act, 1956} and having its registered office at (hereinafter referred to as the "First Part" which expression shall, unless repugnant to the context include its successors and permitted assigns)

AND

2. {... Limited, a company incorporated under the Companies Act, 1956} and having its registered office at (hereinafter referred to as the "Second Part" which expression shall, unless repugnant to the context include its successors and permitted assigns)

AND

3. {... Limited, a company incorporated under the Companies Act, 1956} and having its registered office at (hereinafter referred to as the "Third Part" which expression shall, unless repugnant to the context include its successors and permitted assigns)

The above-mentioned parties of the FIRST, SECOND and THIRD PART are collectively referred to as the "Parties" and each is individually referred to as a "Party"

WHEREAS.

- (A) Inland Waterways Authority of India (the "Authority"), has invited Bids (the "Bids") by its Request for Bid dated (The "tender Document" for "Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry." (The "Work")
- (B) The Parties have read and understood the TENDER Document and are interested in jointly bidding for the Project as members of a JV/Consortium and in accordance with the terms and conditions of the TENDER document and other TENDER documents in respect of the Project, and
- (C) It is a necessary condition under the TENDER document that the members of the JV/Consortium shall enter into a Joint Bidding Agreement and furnish a copy thereof with the Bid.

NOW IT IS HEREBY AGREED as follows:

1. Definitions and Interpretations

In this Agreement, the capitalized terms shall, unless the context otherwise requires, have the meaning ascribed thereto under the TENDER.

2. JV/Consortium

- 2.1 The Parties do hereby irrevocably constitute a JV/consortium for the purposes of jointly participating in the Bidding Process for the Project.
- 2.2. The Parties hereby undertake to participate in the Bidding Process only through this JV/Consortium and not individually and/ or through any other JV/consortium constituted for this Project, either directly or indirectly or through any of their Associates.

3. Covenants

The Parties hereby undertake that in the event the JV/Consortium is declared the Preferred Bidder and awarded the Project, it shall incorporate a wholly owned special purpose company under the provisions of Indian Companies Act, 1956, as the O&M Contractor; in whose subscribed and paid-up capital, the Preferred Bidder i.e. the parties herein shall collectively hold 100% equity during the License Period.

4. Role of the Parties

The Parties hereby undertake to perform the roles and responsibilities as described below:

- (a) Party of the First Part shall be the Lead member of the JV/Consortium and shall have the power of attorney from all Parties for conducting all business for and on behalf of the JV/Consortium during the Bidding Process and until the signing of the tender for "The Work" when all the obligations shall become effective;
- (b) Party of the Second Part shall be assisting the Lead member in the manner as recorded herein for carrying out the entire scope of work awarded under the tender for "The Work". Notwithstanding anything contained in this agreement, party of the second part shall not be absolved of any of its obligations /liabilities/responsibilities merely because party of the first part has been designated as the lead member and shall be individually equally liable / responsible to the same extent as the Lead Member.
- (c) Parties shall jointly and severally endeavour to carry out the works, if awarded to them pursuant to the bidding process conducted by the Authority, in accordance with the terms and conditions specified in the TENDER Documents and such other agreements/Contracts/Work Orders as may be

executed from time to time between the Authority and the JV/Consortium's Special Purpose Company.

5. Joint and Several Liabilities

The Parties do hereby undertake to be jointly and severally responsible for all obligations and liabilities relating to the Project and in accordance with the terms of the TENDER and the tender for "The Work, till such time as prescribed in accordance with the tender for "The Work".

6. Shareholding

- 6.1 The Lead Member of such Preferred Bidder JV/consortium shall at all times during the License Period hold equity equivalent to(as per clause 6.9.2) of the subscribed and paid-up capital in the special purpose company incorporated by the parties to work as the contractor. Further, other consortium members whose technical/financial eligibility shall have been used for the purpose of qualification under this TENDER shall hold (as per clause 6.9.3) equity in the subscribed and paid-up capital contractor during the License Period; Provided however that Authority may in its sole and absolute discretion permit a JV/consortium member to divest [in full/partially] its equity shareholding in the subscribed and paid-up capital of the contractor.
- 6.2 The Parties undertake that they shall comply with all equity lock-in requirements set forth in the tender for "The Work".

7. Representation of the Parties

Each Party represents to the other Parties as of the date of this Agreement that:

- (a) Such Party is duly organized, validly existing and in good standing under the laws of its incorporation and has all requisite power and authority to enter into this Agreement;
- (b) The execution, delivery and performance by such Party of this Agreement has been authorized by all necessary and appropriate corporate or Governmental action and a copy of the extract of the charter documents and board resolution/power of attorney in favour of the person executing this Agreement for the delegation of power and authority to execute this Agreement on behalf of the Consortium Member is annexed to this Agreement, and will not, to the best of its knowledge:
 - (i) Require any consent or approval not already obtained;
 - (ii) Violate any Applicable Law presently in effect and having applicability to it;

- (iii) Violate the memorandum and articles of association, by-laws or other applicable organizational documents thereof;
- (iv) Violate any clearance, permit, concession, grant, license or other Governmental authorization, approval, judgment, order or decree or any mortgage agreement, indenture, or any other instrument to which such Party is a party or by which such Party or any of its properties or assets are bound or that is otherwise applicable to such Party; or
- (v) Create or impose any liens, mortgages, pledges, claims, security interests, charges or Encumbrances or obligations to create a lien, charge, pledge, security interest, encumbrances, or mortgage in or on the property of such Party, except for encumbrances that would not, individually or in the aggregate, have a material adverse effect on the financial condition or prospects or business of such Party so as to prevent such Party from fulfilling its obligations under this Agreement;
- (c) This Agreement is the legal and binding obligation of such Party, enforceable in accordance with its terms against it; and
- (d) There is no litigation pending or, to the best of such Party's knowledge, threatened to which it or any of its Affiliates is a party that presently affects or which would have a material adverse effect on the financial condition or prospects or business of such Party in the fulfilment of its obligations under this Agreement.
- (e) Such Party has read and understood the tender Documents and is executing this Agreement for the purposes as recorded hereinabove out of its own free will;

8. Termination

This Agreement shall be effective from the date hereof and shall continue in full force and effect at all times during the subsistence of the tender for "The Work" is achieved under and in accordance with the tender for "The Work" in case the Project is awarded to the JV/Consortium. However, in case the JV/Consortium is either not prequalified for the Project or does not get selected for award of the Project, the Agreement will stand terminated in case the Bidder is not pre-qualified or upon return of the Bid Security by the Authority to the Bidder, as the case may be.

9. Miscellaneous

- 9.1 This Joint Bidding Agreement shall be governed by laws of India.
- 9.2 The Parties acknowledge and accept that this Agreement shall not be amended by the Parties without the prior written consent of the Authority.

IN WITNESS WHEREOF THE PARTIES ABOVE NAMED HAVE EXECUTED AND DELIVERED THIS AGREEMENT AS OF THE DATE FIRST ABOVE WRITTEN.

SIGNED, SEALED AND DELIVERED

For and on behalf of For and on behalf of the LEAD MEMBER

(Signature) (Name) (Designation) (Address)

SIGNED, SEALED AND DELIVERED

For and on behalf of For and on behalf of SECOND PART

(Signature) (Name) (Designation) (Address)

In the presence of:

1)

2)

Notes:

- The mode of the execution of the Joint Bidding Agreement should be in accordance with the procedure, if any, laid down by the Applicable Law and the charter documents of the executant(s) and when it is so required, the same should be under common seal affixed in accordance with the required procedure.
- Each Joint Bidding Agreement should attach a copy of the extract of the charter documents and documents such as resolution / power of attorney in favour of the person executing this Agreement for the delegation of power and authority to execute this Agreement on behalf of the Consortium Member.
- For a Joint Bidding Agreement executed and issued overseas, the document shall be legalized by the Indian Embassy and notarized in the jurisdiction where the Power of Attorney has been executed.

FORM 4N: General Construction Experience

(Each Bidder or member of a JV must fill in this form)

Name of the Bidder / Members JV Consortium:

General Experience								
Contract Identification and Name	Starting Month Year	Ending Year	Month	Years	Role of Bidder			
Name and Address of Employer								
Brief Description of the similar Works Executed by the								
Bidder								

FORM 40: Bid Capacity

Eligibility and Qualification Criteria		Compliance	Documentation		
Cincin	Single	Joint Vent	Joint Venture (existing or intended)		
	Entity	Member 1	Member 2	requirements	
Available Bid Capacity should be More than value of contract/ contracts applied for				Form 4F & Form 4N	

Bid Capacity= [(A*N*2)-B], where

A= Maximum value of works executed in any one year during the last five years (updated to the current price level), in INR

B= Value at current price level of the existing commitments and ongoing works to be completed during the period of completion of work for which the bids have been invited

N= No. of years prescribed for completion of works for which bids are invited (i.e. 1)

FORM 4 P: Bid Security Declaration

(On Bidders Letter head)

Tender 1	No:	Date:		
То				
I A	Chief Engineer (NER & Other NNW's), Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301			
Subject	t: Design, Construction, Supply, Testing and C Hybrid Electric Passenger Ferry.	Commissioning of 50 Pax		
I/We und	ndersigned declare that:			
	I/We understand that, according to your conditions, bids Securing Declaration.	must be supported by a Bid		
A	I/We accept that I/We may be disqualified / debarred from Authority of India tenders for a period of 02 (Two) years if I/We are in a breach of any obligation under the bid con	from the date of notification		
а	a) have withdrawn/modified/amended, impairs, or of my/our bid during the period of bid validity specified	-		
t	b) having been notified of the acceptance of our Bid period of bid validity (i) fail or refuse to execute the fail or refuse to furnish the Performance Securi accordance with the instructions to Bidders & as per	e contract, if required, or (ii) ity within the timeline, in		
	I/We understand this Bid Security Declaration shall cease from following,	e to be valid on the 45 th day		
8	a) If I am/we are not the successful Bidder, the receip name of the successful Bidder; or	t of your notification of the		
ł	b) The expiration of the validity of my/our Bid or any e	extension thereof.		
(Signatu	ure)			
Authoriz	zed Signatory			
Name: .	Name:			

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
Designation:	
	65

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India) Design, Construction, Supply, Testing and Commissioning of 50 Pax (Hybrid Electric Passenger Ferry)
SECTION - V: FINANCIAL BID STANDARD FORMS
66

FORM FIN – 1: Financial Bid Submission Form

[Location, Date]
To:
Chief Engineer (NER & Other NNW's), Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301
Subject: Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
Dear Sir:
We, the undersigned, offer to provide the services for the Assignment/Job for [Insert title of Assignment/Job] in accordance with your Request for Bid dated [Insert Date] and our Technical Bid. Our attached Financial Bid is for the sum of [Insert amount(s) in words and figures]. This amount is inclusive of all types of taxes (such as GST, Income tax, duties, fees, levies etc.). We hereby confirm that the financial Bid is unconditional and we acknowledge that any condition attached to financial Bid shall result in rejection of our financial Bid.
Our Financial Bid shall be binding upon us subject to the modifications resulting from Contract negotiations, up to expiration of the validity period of the Bid, i.e. before the date indicated in Clause
We understand you are not bound to accept any Bid you receive.
Yours sincerely,
Authorized Signature [In Full and initials]:
Name and Title of Signatory :
Name of Firm:

Item Rate BoQ

Tender Inviting Authority: Inland Waterways Authority of India

Name of work: Design, Construction, Supply, Testing and

Commissioning of 50 Pax Hybrid Electric Passenger

Ferry

Tender No: IWAI/TECH/RP/ELECTRICCATAMARAN/2025

Bidder	
Name:	

PRICE SCHEDULE

(This BOQ template must not be modified/replaced by the bidder and the same should be uploaded after filling the relevant columns, else the bidder is liable to be rejected for this tender. Bidders are allowed to enter the Bidder Name and Values only)

NUMBER	TEXT #	NUMBER #	TEXT #	NUMBER #	NUMBER #	TEXT #
S. No.	Item Description	Quantity	Units	RATE inclusive of 1% Labour Welfare Cess	TOTAL AMOUNT Without Taxes	TOTAL AMOUNT In Words
1	2	3	4	5	6	7
1.00	El Cat Passenger vessel					
1.01	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry	1.0	Number		0.00	INR Zero Only
1.02	GST as applicable	1.0	Number		0.00	INR Zero Only
Total in Figures			1		0.00	Zero Only
Quoted Rate in Words				I	NR Zero Only	1

Signature of	Authorized	Signatory
--------------	------------	------------------

Name	•
Maine	•

Designation:

Name of Firm:

Address:

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
SECTION-VI: TERMS O	F REFERENCE (TOR)
	69

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
PART I – DETAILED SC	OPE OF WORK
70	

1. Background and brief about Project

- 1.1 Inland Waterways Authority of India (IWAI) [hereinafter referred to as "employer"] is a statutory body under the Ministry of Shipping, Government of India. IWAI was set up in 1986 for regulation and development of Inland Waterways for the purposes of shipping and navigation. IWAI is primarily responsible for development, maintenance and regulation of Inland Water Transport (IWT) in the country and specifically National Waterways (NW). In April, 2016, the Government of India has declared 106 new National Waterways in addition to the existing five National Waterways.
- 1.2 Inland Water Transport (IWT) has the potential to form the most economic, reliable, safe and environmentally friendly form of transport. When developed for use by modern inland waterway vessels, it can reduce investment needs in rail and road infrastructure, promote greater complementarities in the riparian states, enhance intra- regional trade and, through increased economies of scale, significantly reduce overall logistics costs for the benefit of the entire economy and India's global trade competitiveness.
- 1.3 Urban water transport in India is gaining momentum as a sustainable solution for easing congestion and reducing pollution in metropolitan areas. With initiatives like the development of waterways and introduction of modern vessels, cities are embracing water transport to enhance connectivity and promote eco-friendly commuting options.
- 1.4 The Harit Nauka Guidelines showcases the Govt's commitment towards transforming the maritime landscape by adopting green vessels & establish operationalization of green ecosystem. It also aims to embrace low/zero emission sources of fuel & attain 100% green vessels in the Indian waters by 2047.
- 1.5 With this aim, IWAI introduced the concept of green fuelled boats, the Government of India also aims to foster the confidence of the stakeholders in the maritime sector and usher them towards transitioning to greener and cleaner fuels as emphasized under the Harit Nauka- Inland Vessels Green Transition Guidelines, published on 8th January 2024.

Through this tender, IWAI invites bid for procurement of 50 pax electric catamaran from Shipbuilders.

2. Scope of Work

2.1 Tender for **Design**, **Construction**, **Supply**, **Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry** from the reputed Contractors/Companies/Firms.

The reputed Contractors/Companies/Firms.

2.2 Preliminary Works

The detailed engineering, design and drawing (approved by IR Class) will be submitted by the contractor along with the work plan, which shall include:

- (i) The Works Program.
- (ii) Health & Safety Plan.

3. Timelines

- (a) The Contract duration for the entire scope of work for **Design**, **Construction**, **Supply**, **Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry** as detailed out in this clause shall be 6 (six) months from the date of signing of the Contract Agreement to this effect.
- (b) The Contractor shall submit a detailed work plan indicating the timelines for main and miscellaneous activities to the EIC for approval.
- (c) The Contractor, in the course of the work, after deliberations with the EIC, shall agree and incorporate the necessary changes in within timelines. The Contractor shall get the macro and micro level activity plan and methodology statement(s) approved by the EIC.
- (d) The Contractor shall abide with the timelines in the Work Plan.
- (e) No idle time charges on any account shall be paid to the contractor during the contract period.

4. Milestones

- 4.1 The Contractor shall carry out **Design**, **Construction**, **Supply**, **Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry** within a period of 6 months till satisfaction of engineer in charge and handing over the electric catamaran to owner's representative.
- 4.2 The contractor, within 15 days from the Letter of award, should submit detailed Schedule of activities including procurement and fabrication schedule, supply and testing schedule, installation and commissioning schedule, method statements, QC Manual including all data and reporting formats, for comment from EIC and should include all comments and observations to get the same approved by EIC.
- 4.3 The Contractor shall during the execution of works adhere to the Project Milestone set forth for the works on monthly basis so as to ensure the completion of Works within the stipulated duration.

S1.	Milestone	Time	Allowed	Payments
-----	-----------	------	---------	----------

No.		from the Date of signing of Contract	
1.	Submission of Draft Design for approval of Employer	20 days	Nil
2.	Submission of Final Design Certified by IRS/ Classification Society for approval of Employer	45 days	10% of Contract Value
3.	a) After the fabrication of entire hull structure after inspection and certification by IRS / Classification Society.	120 days	25% payment
3.	b) After delivery and installation of all machinery & equipment's, completion of Superstructure & Interiors in all respects and Safety measures etc. and certified by IRS/Classification Society thereof.	165 days	25% payment
4.	a) On trial & testing and successful commissioning at site	180 days	30% payment
	b) On successful running of 12 (twelve) months.	••••	10% payment

5. It may be noted that the milestones defined above are for the complete scope of the work. The Contractor shall ensure to adopt all necessary means during execution of works to achieve all the milestones. Failure of achieving milestones would attract LD as defined in Clause 8.12 of GCC.

6. Deliverables and Payment Schedule

- 6.1 The Contractor shall submit stage wise Bills for performing the works, from the Date of Commencement of Work, which will be verified by the Engineer-in-charge subject to deductions as applicable, and recovery towards repayment of Mobilization Advance & other recoveries, if any.
- 6.2 Subject to any deduction, recovery towards repayment of mobilization advance and other recoveries if any and addition authorized by and to the other provisions of this contract, Employer shall pay to the contractor for the design, building, equipment, testing and delivery at specified destination, for the vessels including Dry Docking (import and customs Duty) and for all other works, matters, things and obligations to be executed, done, supplied and performed by the contractor under this contract including the provision of the hull inventory.
- 6.3 The total Contract Price payable under the Contract shall be stipulated in the Letter of Award and there upon shall become part of this Contract and be paid in accordance with the terms ref. 6.5 hereon. The price quoted will squarely and totally include all the charges to be paid to the contractor by the Employer
- 6.4 Total Contract Price shall be firm and fixed and shall not be subject to any escalation during the period of the Contract except for any adjustment in accordance with the terms of the Contract.

6.5 The Contractor shall submit Running Account Bills (RA Bills) on or before the milestone dates fixed by Engineer-In-Charge (EIC) for the quantity and/or work executed with respect to the work progress requirements. The Contractor shall be paid as per the RA Bills submitted for the work done within one month from submission of correct claim of bill. The contractor will submit the RA Bill in triplicate accompanied with 3 sets of supporting documents. Subject to deduction and other changes authorized by and to the other provisions of this contract, employer shall pay to the contractor for the Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry at specified destination, for the vessels including Dry docking and for all other works, matters, things and obligations to the executed, done, supplied, and performed by the contractor under this contract including the provision of the hull inventory as specified by the time and manner with respect to the contract price. Unless otherwise agreed in writing between Chairperson on behalf of Authority and the Contractor payment for the works shall be made by Chairperson on behalf of Authority, by electronic system i.e. RTGS in instalments upon production of the certificate of the inspector and the Director / Dy. Director / SHS / Asst. Director / AHS appointed by owner for the inspection of the construction of the vessel(s) against the instalment due subject to the condition that the amount of an instalment payment shall in no case exceed the value of the work done as per following stages.

Applicable for vessels

- 6.6 Stages of Payment
- 6.6.1 10% after the design approval by IR Class.
- 6.6.2 15% when 50% of hull fabrication and erection is completed
- 6.6.3 10% when 100% of hull fabrication and erection is completed
- 6.6.4 25% on successful launching
- 6.6.5 30% after successful tests and trails, satisfactory final delivery of vessels listed in **Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry** at specified location with the certification of inspection authority and its testing & commissioning complete in all respect.
- 6.6.6 10% on successful running of 12 months.

7. Manpower

The Contractor shall deploy the manpower and mention the experience summary for vessels listed in Tender document as tabulated below:

S. No. Position / Role		No.	Remarks
1	Site In-charge / Manager (BE / B.		

	Tech in Mechanical / Marine Engineering with at least 10 years' experience).	
2	Structural Design Engineer (B. Tech in Naval Architecture with at least 5 years' experience)	
3	Fabrication Engineer (BE / B. Tech in Mechanical Engineering with at least 5 years' experience).	
4	Electrical / Electronics Engineer (BE / B. Tech in Electrical / Electronics & Communication / Instrumentation Engineering with at least 5 years experience).	
5	Surveyor	
6	Supervisors	
7	Class Tested Qualified Welder	
8	Skilled Mechanic	
9	Carpenter	
10	Skilled Labourer	
11	Unskilled Labourer	

Note:

- (a) The above list need not be exhaustive. The Contractor shall however undertake that all the time required number of personnel necessary for efficient execution of contract and as required by applicable law.
- (b) The number of personnel mentioned is indicative as may be required for the Works. The Contractor shall engage persons with necessary qualification / experience during the entire period of the Contract.
- (c) The list of all the personnel engaged in the Contract will be given to EIC before commencement of work. Personnel will be changed only after approval of EIC.

8. Equipment

8.1 The Contractor shall have the vessel fabrication facility (work facility certified by IR Class / any other IACS body) along with all necessary equipment to carry out the work, such as cutting, grinding, blasting, painting, welding etc. The contractor shall provide required list of owned / hired equipment which are to be used for the work.

S. No.	Equipment type and characteristics			Minimum number	
,	Equipment	make	Rated Capacity	Max. Age (Years)	required
1					
2					
3					
4					
5					
6					

9. Organization

The selected Contractor shall depute well-qualified officers/team having sufficient experience in execution of similar works as indicated in this Tender Document. If the progress of work is found unsatisfactory during the currency of the contract, Contractor/Bidder shall promptly mobilize additional personnel/resources for ensuring satisfactory progress and timely completion of the proposed work, as per direction of and satisfaction of Engineer-In-Charge under the contract without extra cost to IWAI.

10. Duties & Responsibilities of the Contractor

10.1 The primary responsibility of the Contractor is to undertake **Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry**at specified location as per terms and conditions of this tender and directions of EIC during the entire period of the contract. Correction of Defects If any defects including workmanship of hull, structure, machineries, or any other part of the structure appears within twelve (12) Months of taking over certificate, the owner shall give notice to the contractor of any defect before the end of the defects liability period which begins at completion and is for Twelve (12) Months thereafter. The defects liability shall be extended for as long as defects remain to be corrected.

When the notice regarding the defect is given, the contractor shall correct the notified defect at his own cost within the length of time specified by the owner in notice issued by him. If the contractor is in default, the owner shall cause the same to be made good by other workmen and deduct the expense from any sum that may be due to the contractor.

If the contractor has not corrected a defect, to the satisfaction of the engineer-incharge, within the time specified in the notice, the owner will assess the cost of having the defect corrected and the contractor will pay this amount.

11. Supervision and Monitoring

Monitoring shall be undertaken by the employer; this shall be done by IWAI representative or an appointed supervision consultant or EIC (engineer in charge). Any supervision and monitoring shall not relieve the contractor from his contractual obligations, including his duty of quality control, survey and all such factors.

12. Association of Part I and Part II of ToR

The technical specifications detailed in part II of ToR shall be read in conjunction with part I – scope of work. Additional works detailed in part II of ToR shall also fall in the scope of work of the contractor.

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
PART II – TECHNICAL S	SPECIFICATIONS
78	3

TECHNICAL SPECIFICATIONS OF DESIGN, CONSTRUCTION, SUPPLY, TESTING AND COMMISSIONING OF 50 PAX HYBRID ELECTRIC PASSENGER FERRY

1. GENERAL

1.1 INTENT

The purpose of this document (hereinafter called the "Specifications") is to outline the technical and functional requirements for the **50 Pax Hybrid Electric Passenger** Ferry meant for inland ferry services, (hereinafter called the "Vessel") for Inland Waterways Authority of India. The 50 Pax Hybrid Electric Passenger vessel shall be operated to introduce safe, scientific, economical, and efficient way of transporting of passengers round the clock during the entire year including flood season. The vessel to carry 50 passengers (excluding crew members).

The vessel including all its material, equipment, piping, machinery, workmanship, etc., shall be in accordance with these specifications and to the requirements of the classification Society and in accordance with IV Act- 2021 (Design & Construction Rule 2022).

All fittings, arrangements, systems, and equipment not mentioned in the specifications but required under Classification rules and other statutory requirements shall be provided at no extra cost. Anything not described or left out of this specification, but being considered as normal and necessary for the intended services, shall be supplied, and fitted without extra charge.

1.2 OPERATIONAL REQUIREMENT AND TYPE OF BOAT

1.2.1 OPERATIONAL REQUIREMENTS

The vessels will operate in perennial rivers (annual variation of water level – upto 8m approx.) and zone 1 (as per IV Act 2021). The short duration trips with multiple stops, high passenger counts, and quick battery charging requirements present significant operational challenges, which are further explained below:: -

- a) Route: Operations in rivers / creeks present significant operational and reliability challenges due to multiple manoeuvring and docking evolutions, incoming vessel traffic, channel restrictions, short duration high speed segments and intermittent low speed segments. These are potentially hard on the vessel's propulsion machinery and structure.
- b) Boarding/Alighting: The vessels shall be designed in such a way as to facilitate the safe and efficient transfer of passengers and their belongings within the minimum time per disembarkation/embarkation.
- c) Boarding facilities for passengers: Safety treads or non-skid material shall be installed on the traffic areas of all boarding areas. The Builder shall ensure that the

vessel on offer is compatible with planned (and/or existing) terminal facilities (free board of 0.4m - 0.7m is available at IWAI terminals). The Boats will primarily operate alongside Concrete / steel / HDPE floating jetties but shall also be capable of berthing in fixed jetties as well. The Tenderer shall briefly document his plan for fulfilling these obligations by submitting a Boarding Plan to the owner.

- d) Wake wash and speed Restrictions: Some of the segments of the operating routes are prone to issues due to wake wash and port traffic. The vessel shall exhibit low wake wash characteristics at cruising speed of 10 knots. Builder is encouraged to incorporate such capability to the extent possible. The Builder's vessel design shall minimize adverse effect of wake and propulsor wash. These elements shall be characterized by the wave height of the proposed design's wave and propulsor wash signatures while in the full load end-of-service-life condition.
- e) Vessel shall exhibit low wake wash characteristics at all times within the design speed of 10 knots. The Proposed design's wake and propulsor wash shall have a wave height less than the maximum values shown in Table at the speed of 10 knots in deep water. The wave height shall be predicted using CFD calculations during the design phase at a distance of 2 x vessel length from the vessel transit line and to be proved during model test.

ITEM	WAVE HEIGHT
Max value at a speed 10 Knots	24 cms
Target value to be achieved	20 cms

f) Environmental Conditions: The vessel should be able to perform consistently under the following conditions:

• Significant wave height: 2 m

• Maximum ambient temp: 55 deg C

• Relative Humidity: 90%

- g) Sea keeping and manoeuvrability: Vessel shall exhibit excellent motions to maximise passenger comfort. Vessel manoeuvrability characteristics shall allow for rapid, safe and controlled docking in all weather and current conditions prevalent in the area of operation.
- h) Mooring/Berthing arrangement: Docking (Berthing) of the vessels shall be Semi-Automatic arrangement with minimum manual assistance in shortest time and safely tied up for shore connections for Battery charging. Battery charging connections to be made in shortest time, with minimum manual effort in a safe manner at Jetties.

i) Propulsion system: Propulsion system shall be of 2 nos podded electric type or electric outboard steerable propellers, fitted on aft. Provision shall be made to tilt/lift the system in order to get access to propellers for cleaning / removal of entangled weeds etc from the deck/platform.

1.2.2 TYPE OF BOAT

The Boats shall be Electric Hybrid Catamaran Boats with twin hulls separated by appropriate distance to avoid unfavorable wave interference, having a single deck connecting the two hulls, capable of carrying 50 passengers including provision for carrying 1-wheel chair. The boats shall be driven by Two Nos. podded electric type or electric outboard steerable propellers. The hull and main deck shall be Marine grade Aluminum (IACS member approved grade 5083 AA or equivalent for plating and 6082 AA s or equivalent for extrusion). The passenger accommodation shall be of FRP construction. The passenger cabin area shall be kept to a minimum and shall be air-conditioned. The vessel shall be simple and aesthetically pleasing with modern features.

The Propulsion system shall be 2 nos podded electric type or electric outboard steerable propellers, supported by LTO Batteries, capable of being fast charged by electric supply from grid ashore at high charge acceptance rate and Diesel Generator(s) on board for hybrid operation and back up in emergency.

Principal Dimensions and Particulars

The following are the approximate principal dimensions for 50 Pax boats.

Length Overall	18.0 m appx
Breadth Overall	About 5.6m (considering the wake wash requirement)
Breadth of Demi hull	1.50 m appx
Freeboard	0.8 m
Draught (max)	0.6 m
Air draft (max)	3.6 m - 4.2 m
Propulsion (Battery+ Diesel Generators)	2 nos podded electric type or electric outboard steerable propellers
Design Speed- Hybrid mode Battery power+ DG power)	10 knots

Service Speed (Pure electrical)	7 knots
Service Speed (Pure diesel mode)	7 knots
Endurance (Battery)	2 Hrs (pure electrical battery operation) with max drop in SoC 60% starting from 80% SOC
FO Tanks capacity	Sufficient for operation of 24 Hrs at 10 knots on diesel in hybrid mode
Passenger Carrying Capacity	50
Seating	50 (including 4-wheel chair)
Compliment	3 (As per IV rule / IACS Rules)
Deadweight	5 - 10 Tonnes

The above parameters shall be updated based on hull form optimization study, model test results and design progress.

1.3 CLASSIFICATION, RULES AND REGULATIONS, STANDARD OF QUALITY – GENERAL INFORMATION

1.3.1 CLASSIFICATION

The boat shall be built under classification with the IRS notation: IW, IY, ZONE1, Ferry or equivalent of any other IACS Member Class notation and DNV Notation "Battery" (applicable clauses for inland vessels). The Boats shall be registered as an Inland Water Self Propelled passenger vessel with the State IWT / Maritime Board.

1.3.2 RULES AND REGULATIONS

- 1. The Boat should comply with the Rules and regulations as prescribed Inland Vessels Act 2021 as amended and rules thereupon
- 2. Rules and Regulations for the Construction and Classification of Inland Water vessels– Indian Register of Shipping or Equivalent IACS Member Class.
- 3. IMO Resolution A 952 (23) Graphical symbols for shipboard fire control plan
- 4. IMO Resolution A 760 (18) Symbols related to life saving appliances and arrangement
- 5. IMO Resolution A 658 (16) Use and fitting of retro reflective material on life saving Appliances

1.3.3 GUIDELINES AND RECOMMENDATIONS

The boat shall be constructed with the following guidelines & recommendations.

- a) Disability access guidelines issued by International Navigation Association, 2004 and Accessibility guidelines issued by MoPSW, GoI in November 2023.
- b) The Rights of persons with disabilities Act.2016
- c) National Disaster Management guidelines on boat safety, 2017, GoI
- d) IMO Resolution A 756 (18) Guidelines on information to be provided with fire control plan and booklets required by SOLAS regulation II-2/20 and 41-2
- e) IMO Resolution A 468 (XII) Code on Noise levels on board ships 1981 as recommendation shall be followed for keeping the sound levels below the permitted maximum, in the passenger seating area and wheelhouse.
- f) ISO 6954-2000(E) Guidelines for evaluation of vertical & horizontal vibration in Merchant Ships (Peak values) Vibration levels in passenger seating area shall be below the upper limit of the "stipulated zone" in the figure of guideline.
- g) MSC 307 (88)-2010, Fire Test Procedure code.

1.3.4 LIST OF CERTIFICATES AND DOCUMENTS

All certificates shall be delivered in two-sets to the Buyer, unless otherwise stated in the Contract. These certificates shall be submitted to the Buyer prior to delivery of the Boat. The Buyer is responsible for the registration of the Boat, all the necessary documents, as required shall be provided by the Builder. All certificates mentioned below but not limited by this list, shall be delivered:

- a) All certificates required under IV Act 2021 & Classification society
- b) Class Certificates
- c) Builder's Certificate and Deadweight Certificate
- d) Certificate of Class
- e) Type approval Certificates for all equipment (main and auxiliary machineries) including batteries
- f) Type approval Certificates for all LSA and FFA items
- g) Certificates required for lifting appliances
- h) Documents for Certificate of Registry (Statutory Authority)

- i) Certificate for navigation aids, winches, mooring ropes & other mooring equipment, castings and forging etc. (Class/Statutory authority or maker certificate as applicable)
- j) Trim and Stability Booklet (Statutory authority)
- k) Light weight & Deadweight calculation sheet from shipyard.
- 1) Statement of compliance for TBT free anti fouling paint issued by the paint manufacturer and Class in case of paint being applied
- m) Maker's certificate for light and navigation aids
- n) Certificate of Marine equipment issued by notified body
- o) Technical file for all engines, if applicable
- p) All machinery, electrical, communication and navigation equipment; anchors and chain cables; lifesaving and fire-fighting appliances shall be certified by the administration or by an Administration acceptable to them by recognized organization
- q) Any other certificate required by for registration
- r) Asbestos free declaration from the builder
- s) Other certificates not listed in this specification but required by the rules and regulations "as mandatory" shall be supplied.
- t) "As built" and "as fitted" drawings. Four complete sets of paper print and two soft copies on DVD for Hull, Machinery, Deckhouse, Electrical and Auxiliaries.

1.4 REGISTRATION

Boats shall be registered by Competent Authority as prescribed under IV Act 2021. Builder shall provide all necessary documentation and enable the survey requirements for registration and assist Owner suitably.

1.5 HYDRODYNAMIC DESIGN & MODEL TESTING

Boats shall be designed optimally for a very low power system. As far as possible, the design should be based on a proven design.

The CFD analysis to be carried out and report to be submitted covering all aspects of resistance and waves/wake wash for at least 3 depths i.e. 1.5 m, 2.5 m & 3.5 m. along with preliminary design.

The model testing of the Boats shall be performed in facility conforming to ITTC standards of tank testing and extrapolation. The model shall be manufactured to a

geometric scale of the full-scale boat including appendages. The hydrodynamic design facility is to predict the speed performance characteristics of the vessel in deep water and also in shallow water of 1.5 m, 2.5 m and 3.5 m water depth.

Appendages to be designed to minimize the resistance. Anode arrangement shall be preferably flush type to the hull.

The builder shall carry out the following tank tests/ calculations before the final design for the first vessel in an international reputed testing facility and arrive at an optimum design with lowest power and minimum wake wash characteristics.

Proper test report to be submitted to the Owner

- a) Resistance test at the design and light draught for a speed range of 6 to 11 Knots Shallow water test depth of water < 2 m
- b) Paint flow test at design draught at design speed (or this may be substituted by alignment of appendages in CFD analysis)
- c) Propeller open water test for designed propeller
- d) Self-propulsion test with designed propeller at the design and light draught at both 7 knots and 10 knots speed.

Model used for testing shall be preserved and delivered to owner after completion of tests.

Buyer to witness the tests. Soft copy in AUTOCAD format and hard copy (3 sets) of drawings and results of the model test need to be submitted to the Buyer along with lines plan & offset table of the model.

1.6 MODEL SUPPLY

Five models for the series of boats in 1:20 scale to be supplied in illuminated glass-box. Each model to be supplied along with the first vessel of each lot of 5 vessels.

1.7 SPARES AND INVENTORY

Following spares and inventories shall be supplied by the builder (List to be submitted).

Spares as required by classification society/ statutory authority to be supplied by yard.

Maker's standard spares for maintenance for a period of Four years for all equipment.

On-board spares* (Deck, accommodation, engine and electrical inventory) and tools should be provided as per the equipment manufacturer's recommendations for daily and routine maintenance for a period of 4 years.

Suitable minimum tools for maintenance during normal operation shall be provided as per OEM standard scope.

1 Spare anchor, 1 spare Electric podded electric type or electric outboard steerable propellers (complete unit with drive motor) for series of every 5 vessels, supplied with the first vessel of lot

Lifesaving appliances- duly approved by IR class / any other IACS body.

1.8 WEIGHT MONITORING AND CONTROL

The Builder shall religiously exercise weight monitoring and control during the design and construction of the boat to ensure that draft is not exceeded.

A weight management and control programme will be operated by the shipyard

1.9 TRIALS

The inspection, test and trial during construction of the Boat shall be carried out by the Builder on board in accordance with the requirements of the Classification Society and/or the Regulatory Bodies concerned to demonstrate satisfactory workmanship, suitability for the purpose intended and compliance with Technical Specifications in the presence of Buyer's representatives, Classification Society and Statutory Authorities as per the agreed tests and trials plan. Inspection, Tests and Trials shall be intimated 2 weeks in advance to the buyer or buyer's representative. Whenever such trials are planned outside the country, a notice of 60 days shall be given.

The inspections, tests and trials are categorized under four (4) major heads:

- a) Day-to-day inspections at shipyard and at sub-contractors works,
- b) Shop tests and trials of machinery and equipment
- c) On board & Quay side tests and trials of systems after installation,
- d) Sea trials

Programme for trials shall be submitted to the Buyer at least one month before the trials. Trial procedures & protocols shall be submitted to the Buyer two weeks before the trials for approval. Builder shall provide fuel oil, lube oil, fresh water, etc.as required for the trials. Grades of fuel and lube oil used shall be equivalent to Manufacturer's recommendations. Un-broached left-over stocks of fuel and lube oil shall be taken over by Buyer as per contract.

After the completion of all trials, the results of these inspections, tests and trials shall be submitted and confirmed to the Buyer and Class. If the reports are not to the satisfaction of the Buyer as per the Trial procedure, the surveyor of the Classification Society and the Regulatory bodies concerned, the same shall be notified to the Builder and Builder shall arrange another trial after carrying out corrections, if any.

1.9.1 FACTORY ACCEPTANCE TRIALS

A list of major/ important equipment machinery shall be drawn up mutually by the builder &Buyer of the various equipment that shall be inspected in the presence of Buyer & classification society. Trials procedures/ protocol shall be forwarded to the Buyer 2 weeks in advance. Whenever such trials are planned outside the country, a notice of 60 days shall be given.

Following trials shall be attended by Buyer/ Class.

- a) Diesel Generator Load, safety cut out & Governor trials, vibration tests.
- b) Propulsion machinery tests
- c) Battery capacity & Battery Management System trials
- d) Battery safety/ performance /discharging/ charging tests / Main switchboards and Power Management Systems

1.9.2 DOCK/QUAY TRIALS

On completion of installation of machinery & systems, dock/quay trials of such systems shall be conducted in the presence of Buyer/ Class. A schedule of such trials shall be drawn up & submitted to the Buyer/ Class one month in advance. Trial procedure shall be submitted by the builder to the Buyer for approval along with the schedule.

During quay trials/ inspections and tests, following checks shall be made as applicable:

- a) Performance
- b) Capacity measurement
- c) Pipeline for testing and electrical cable inspection
- d) Equipment inspection after testing
- e) Compliance of all rules, regulations etc.
- f) Inspection for easy access and maintenance
- g) Name plates of all equipment's panel boards, fittings, valves etc.
- h) Automation test according to Rules
- i) Functional test of CCTV system, Passenger information display system and passenger announcement system.

j) Functional test of fire detection and extinction equipment and safety equipment to be carried out before sea trial.

1.9.3 INCLINING EXPERIMENT

To determine the stability, lightship weight and coordinates of the centre of gravity, an inclining experiment according to the regulations shall be carried out in the presence of the Buyer's/ Class representatives and representatives of IWAI. This experiment shall be the basis for the final calculations of trim and stability. The inclining test may be omitted for the subsequent Boats and only a light ship weight estimation may be sufficient, subject to compliance with statutory rules and class rules.

1.9.4 SEA TRIALS

Upon completion of dock-trial and inclining experiment the completed Boat is to be presented to the Buyer/Class for a sea trial. Procedure and requirement as per classification to be followed.

Following sea trials shall be conducted:

- a) Propulsion Power measurements (Propulsion power for continuous operation at 7 knots service speed at design draught shall be measured for 3 hrs in two double runs. The measurement shall be affected at input power to Propulsion motors. Reading shall be taken from the HMI screen in the Bridge)
- b) Anchor trials, steering manoeuvres
- c) Manoeuvring tests including turning circle measurement and crash test ahead and astern
- d) Test of navigation and communication equipment.
- e) Speed trials including progressive speed trials at both light weight and design draft conditions.
- f) Functional and safety tests all equipment
- g) Endurance trials for battery for 4 hrs with fast charging after every hour.
- h) DG capacity at design speed
- i) Fuel consumption
- j) Vibration and noise measurements during endurance trials
- k) Battery and system performance trial
- 1) Emergency Steering Test

m) Night navigation and thermal camera tests

Berthing trials at IBP route in Kolkata at selected locations to be finalized by IWAI.

A detailed program for the sea trials shall be agreed upon with the Buyer at least one month before the commencement of trials. The trial procedure shall be submitted by the Builder along with the sea trials programme.

If any basic faults are found during the trials, the Builder shall be obliged to repeat the trials in case of necessity, within a reasonable period.

1.9.5 PROPULSION POWER MEASUREMENTS

Propulsion power for continuous operation at 7 knots service speed at design draught shall be measured for 3 hrs in two double runs in pure diesel mode and in pure battery mode. The measurement shall be affected at input power to Propulsion motors.

The vessel should run for not less than 3 hrs in pure diesel as well as in pure electric mode.

Where in battery mode State of charge of battery after every 1 hr of trial (total trial of 3 hrs) should be between 80% and 20%

1.9.6 SPEED TRIALS

Speed trials shall be conducted at light draught and design draught with clean bottom with wind speeds not exceeding Beaufort Scale 6 in deep water. Speed trials to be done for in two double runs, as mentioned below:

- a) At design speed of 10 knots at design draught conducted in hybrid mode (using both battery & DG Sets)
- b) At service speed of 7 knots at design draught conducted in pure electric mode (using battery only)
- c) At service speed of 7 knots at design draught conducted in DG mode (using DG Sets only)

1.9.7 POWER MANAGEMENT & BATTERY MANAGEMENT SYSTEMS (PMS & BMS)

The power management system shall be tested for optimum and economic operation of generator sets sharing different propulsion loads. The system shall also be tested upon auto starting (on/off) of generators at different conditions.

A detail system philosophy to be submitted for review and approval.

Battery Management System shall be tested for:

a) Cell level Monitoring.

- b) Thermal over runs warning/isolation.
- c) Over Charging and Over Discharging
- d) Over Current
- e) Over Heating and Excessive Cold
- f) Short Circuit
- g) Visual inspection of mechanical damage to cells

1.9.8 BATTERY & SYSTEM PERFORMANCE TRIALS

Following performance trials shall be undertaken by the builder:

- a) Pure electric mode: Battery capacity & system shall be carried out by propelling 50 pax boat continuously on electric motors without using DG (DG shall not be used) at 7 knots for one hour at design draught with al normal auxiliaries including air conditioning running. At the commencement of this trial, the battery state of charge (SOC) shall be around 80%. On completion of one-hour continuous operation at 7 knots, the battery SOC shall be as recommended by the Builder to meet criteria outlined at Section 6.2.1. At the end of the trials, the battery SoC shall not fall below the designed Depth of discharge and in no way drop more than 60% from initial value and in no way below a residual SoC value of 30%. The other boat parameters shall be maintained within safe working limits during performance trials.
- b) Hybrid mode: Performance trials on hybrid mode to be conducted at 10 knots speed at designed draught. During the trials Battery performance measurement and load sharing properties as well as Generator starting /stopping characteristics shall be tested. Continuous operation for one hour shall be checked in hybrid mode.

1.9.9 FINAL DRY DOCKING / BOTTOM INSPECTION

In case the Boat has been afloat for more than 120 calendar days before delivery, the boat has to be dry docked. Deficiencies/ defects, if any, shall be rectified, prior to commencement of sea trials/ delivery. This inspection shall be attended by Buyer & Class.

1.10 WORKMANSHIP

The workmanship and the construction equipment of the Boat are to be of current / very modern in every respect and shall comply with first-class international shipbuilding practices meeting the class requirements.

Materials, machinery, equipment etc. shall be new, of good/reputed makers and shall be generally be in accordance with maker standards, rules and regulations.

1.11 REPORTS AND PHOTOGRAPHS

Builder shall provide construction detailed schedule to buyer and periodic detailed reports on progress of construction with colour photographs to Buyer. These reports shall be on a fortnightly basis. A quality time lapse video for the entire construction of the first vessel to be provided.

1.12 **DELIVERY**

The Boat shall be delivered to the Buyer in ready to operate condition, without any work pending, afloat at Kolkata.

Operation shall be demonstrated, and training provided to IWAI in the channels of Kolkata.

1.13 GUARANTEE / WARRANTY

The Boat including all the machinery and equipment shall be guaranteed for a period of 36 months from the date of delivery. Any defect except normal wear and tear shall be dealt with by the Builders at their cost.

- a) Extended Warranty (Propulsion system): Propulsion system including podded electric type or electric outboard steerable propellers, its controls, power supply systems, Hybrid systems and generator system to be warrantied for an additional period of 36 months from completion of warranty period.
- b) Battery system: The builder shall provide a replacement warranty for 32000 full charging cycles or 7 years whichever occurs later from the date of delivery of the vessel. Partial charging cycle shall be also considered as part thereof to the total cycle based on the approved life cycle curve. The charging cycle is based on an average full cycle operation of 15 cycles per day for a period of 7 years. Full charging cycle means charging discharging between the threshold values of 80% and 30%. The risk of warranty to be covered through a third-party insurance for this warranty.

1.14 DEADWEIGHT, CAPACITIES, STABILITY AND TRIM

1.14.1 DEADWEIGHT

The deadweight measurement shall be carried out as per classification society and administration guidelines in the presence of the Buyer and the Classification Society and Regulatory bodies.

1.14.2 CAPACITIES

- a) Void Tanks (As per general arrangement drawing)
- b) Fuel oil tank

A Fuel tank with total capacity to facilitate 24 hrs of operation at diesel mode shall be provided. Calculation shall be usable storage only. The Fuel tank shall be suitably positioned so that positive suction for DG is ensured. One mechanical means and one digital means of measuring the fuelconsumption shall be provided. High level and Low level alarms to be provided. Display of the alarms and fuel consumption to be available in the wheel house.

The provision to periodically transmit fuel consumption data log to ITMS system shall be provided.

1.14.3 STABILITY AND TRIM

The Boat shall be designed in a way that the Boat will be afloat with appropriate trim, approximately on even keel or slight aft trim at departure condition when loaded to the designed draft conditions. Forward trim is not allowed in any of the operating conditions.

The intact stability and damaged stability criteria shall be as per Classification notes "Stability Requirements for Inland Waterways Passenger Ships". Damaged Stability criteria shall consider battery compartment flooding in the event of major fire or explosion in the battery compartment.

At the primary stage, the Builder shall submit trim calculations and stability data to the Buyer for the main conditions.

1.14.4 LIFTING ARRANGEMENT/ DOCKING PLAN

Builder shall appropriately strengthen the hull (at frames to be indicated & marked by the builder) to facilitate upright lifting of the boat for maintenance/ dry docking using a travel lift facility (details to be provided by Buyer). The arrangement for strengthening of hull to facilitate upright lifting using cradle

The builder shall also have to provide docking plan (versions 1 & 2) along with the list of drawings.

1.14.5 MAKER LIST

Makers list for machineries, Electrical equipments, outfittings etc. shall be approved by owner.

II. HULL AND SUPERSTRUCTURE

2.1 HULL DESIGN PRINCIPLES

Shipbuilding Quality Aluminium (IACS member approved grade 5083 AA or equivalent for plating and 6082 AAs or equivalent for extrusion) shall be used.

All outfitting and fasteners shall be of stainless-steel material (316L).

Necessary Camber shall be provided on main deck to avoid water accumulation Suitable corrosion margins to be assumed in all the critical areas prone to wear/erosion/corrosion.

The structural design of the hull is to be made for avoiding stress concentrations or keep stress as low as possible respectively. Any static and dynamic loads assumed for the structural design to be in accordance with the Rules.

The passenger accommodation is to be made of FRP. The connection between Aluminium hull and FRP Passenger Accommodation is to be in line with best international shipbuilding practice and relevant Rules and Regulations.

Builder should aim at keeping the displacement to a minimum.

2.2 HULL CONSTRUCTION PRINCIPLES

Welding operation is to be carried out under protection from adverse environment in completely enclosed atmosphere. Class approved procedure to be used. Approved weld procedures and list of approved welders shall be furnished.

Sea chests shall be fitted to the bottom of each hull as required. Each inlet box shall have a flushed mounted removable cover for inspection & cleaning. The boat generally shall be of welded construction, in weather and moisture exposed areas (e.g. tanks, void spaces, etc.) continuous welding shall be used. Each sea suction chest shall be fitted with" injection valve" for cleaning the grating cover of sea chest, with compressed air. Strainers/filters shall be provided wherever required for seawater intake from sea chest.

The welding of the hull shall be tested by NDT in compliance with the classification rules.

NDT Reports shall be submitted to class and the Buyer's representative for approval.

2.2.1 SHELL PLATING

Shell plating shall be of welded construction with transverse or longitudinal framing.

Thickness of shell plate and bottom plate shall be of Minimum 8 mm and deck plate of minimum 6mm. Deck shall be finished flat without undulations. Plates in the aft region near the propeller shall be reinforced to reduce transfer of vibrations.

Plates shall be reinforced, if required, in way of lifting area in order to lift at the boat yard with Travel lift.

Generally, non-watertight structural members shall be provided with adequate access/lightening/scallop holes in accordance with the Builder's practice, provided such hole is necessary or practicable and not detrimental to the structural strength of members.

Manhole arrangements shall be according to rules. Manholes and bilge wells shall be accessible.

Survey grade Echo Sounder, GPS, Class-B AIS, IPad pro cellular (duly with charging facility) and antennas shall be arranged at suitable location.

Additional thickness of minimum 2 mm shall be provided in way of sea chest, propeller wake area, areas of hull opening and in way of anchor stowage etc and shell plate up-to one fourth of the vessel length forward area and deck plate forward of superstructure.

2.2.2 DECKS AND BEAMS

All decks are to be of all-welded Aluminium construction.

Decks, in way of winches, fairleads, bollards, other equipment on the deck shall be strengthened adequately.

2.2.3 MACHINERY FOUNDATIONS

Seating of podded electric type or electric outboard steerable propellers as applicable, shall be adequately strengthened as per class rules. DG seating shall be incorporated in the web frame/longitudinal girder structure. Seating of main components shall be in accordance with recommendations of the machinery manufacturers. Pillars in way of passenger spaces shall be fitted as required by rules. Foundation layout shall be in accordance with static and dynamic loads (vibration prevention). Plate in way of DG seating shall be in accordance with OEM's/ maker's recommendations. Vibration mountings may be incorporated in DG seating.

Transition material bi metallic strips (explosive bonding) shall be used wherever there are welded connections of different material to prevent corrosion. Insulated joints and bolts to be used on all foundations with bolting.

Utmost care to be taken to avoid all chances of bimetallic corrosion and galvanic corrosion of all sorts both within the vessel and outside. Detailed plan should be submitted for approval.

2.3 SUPERSTRUCTURE CONSTRUCTION

Superstructure shall be of Sandwich construction and Vacuum infusion technology shall be used for fabrication of the superstructure.

Resin: - Water resistant low viscosity fire retardant Epoxy/Modified epoxy/Epoxy backbone resin suitable for Vacuum infusion technology shall be used.

Fibre: - Stitched multi axial fibres shall be used for construction or Carbon fibre

Core material: - PVC Foam material. The density of bulkheads, super structure, deck floor & stiffeners shall be of 80 Kg/m3 minimum. Plywood shall not be considered

for the construction of vessel without prior owner approval. For securing areas PVC foam to be of minimum density of 350 Kg/m3 (if any).

A layer of surface tissue shall be used with Gel Coat, to get good surface finish, gloss, scratch resistance and anti-graffiti properties. The colour of panels shall not fade or discolour with time or change due to rubbing. Painting of surfaces after curing of FRP panels shall be avoided. Colours shall be achieved by using pigmented Gel Coat in the moulds. Suitable gel coat compatible with epoxy resin layers and UV resistant shall be used for fabrication.

Superstructure shall be free of all probable defects of FRP, especially blistering, Chalking, Crazing, Cracks, Fading and water spotting, Distortion or warping, yellowing etc.

Delta- E value for colour fading as below:

- a) For light colours, variation of Delta of (dE) less than or equal to 4
- b) For Dark colours, variation of Delta (dE) less than or equal to 6.

Extended warranty of five years ensuring quality for covering all aforementioned defects over and above the standard warranty shall be provided.

Samples of FRP to be tested as per class requirements for

- a) Water resistance
- b) UV resistance
- c) Applicable provisions of fire-retardant tests for resins and gel coat
- d) Combustibility testing as per class requirement
- e) ISO standards for Resin curing to be observed

III. OUTFITTING

SS316L shall be used for All external items exposed to sea water such as

- a) Cleats
- b) Door fittings
- c) Hand grabs
- d) Handles
- e) General fittings
- f) Hand rails, Bollards could be SUS 316 L or Aluminium

g) Ramp - Aluminium alloy telescopic with 8m length

3.1 ANCHORING AND MOORING ARRANGEMENT

SHHP/ HHP type Anchor (based on design considerations and class requirements) shall be provided with suitable anchor chain or wire rope.

Anchor stowing arrangement shall avoid chances of bi-metallic/Galvanic corrosion with aluminium hull.

Suitable electrically operated anchor winch as per anchor weight shall be provided at suitable location. Also, to provide manual operation in case of emergency. Provision shall be provided for stowage of anchor chain/rope below deck. Stowage space shall be provided with bilge pumping arrangement.

Bollards and cleats shall be arranged at forward and aft deck. Forward additional bollard/ arrangement to be provided capable of getting towed. Mooring arrangement in the midship area shall include a electric vertical winch for assisting the berthing of the boat. Mooring arrangement of the boat shall be approved by Owners. Polypropylene ropes is to be provided as per rule requirement.

3.2 SEMI AUTOMATIC BERTHING/MOORING ARRANGEMENT

Suitable simple arrangement with minimum manual interference shall be provided to enable mooring/ berthing to the floating jetties with a mechanical arrangement with lock and/ or single mooring line handling the boat. Suitable strengthening below the semi-automatic berthing/ mooring area shall be provided to cater for the berthing load. The locking if provided, shall be remotely released from wheelhouse. The arrangement shall be such that berthing on both Port and Starboard side is made possible. The arrangement shall be Complimentary with existing arrangement on floating pontoons/ Boats and to be to be agreed with Buyer during design phase. The arrangement should include a vertical electric winch at the centre. One set of suitable Boat hook to be provided for handling the mooring rope conveniently.

3.3 OTHER OUT FITTINGS

A total of two aluminum alloy ramps shall be provided at the aft entries/exits (Port and Starboard). The ramps shall cater to unassisted wheelchair passengers. Ramps shall be foldable/retractable using a hinged mechanism. The operation of ramp should be single handed, quick with minimum effort and lockable. Operation through winches are not accepted. At rigged position ramp shall be at grade with deck end and ramp hinges shall not be projecting outside of the side shell.

The underside to be protected by means of hard plastic or other approved means to avoid wear between ramp and pontoon surface. Suitable compact roller arrangement may be provided, it shall have an electronic indication on the wheel house on its status (OPEN /CLOSE) at any point of time. If the ramp is Electrically operated, it shall be

controlled from Wheel house and locally. The side rail of the ramp shall be foldable, the details shall be shared by the buyer during detail design phase.

The Ramp if Electrically operated shall be controlled from Wheel house and locally

Free board of pontoon is 0.4 - 0.7 m

Keep chain arrangement for emergency exit from forward deck. Portable Foldable ramp shall be provided for emergency operation. Safe passenger embarkation and disembarkation shall be provided during double banking.

Hatches and manholes provided shall be flushed type inside passenger cabin area.

Ladders within battery room, generator room tanks shall be made of Aluminium alloy. FRP ladders shall be provided for access to the top of superstructure.

Handrails, grabs, and/or bulwarks shall be fitted on all decks where necessary for the safety of passengers, and for crew access.

D type Rubber fenders of fixed type along with rubber tyres to be fitted shall be provided on all sides of the boat, which shall be positioned such that it won't interfere with the pontoon edges Sufficient foam fenders shall be provided with eye arrangement on deck.

3.4 CONTROL EQUIPMENT FOR STEERING ELECTRIC OUTBOARD

Maneuvering shall be with steering wheel of reasonable size (300mm) positioned comfortably for prolonged use in sitting position ergonomically for continuous operation. The steering of the thrusters shall be synchronized, also capable to be operated individually. Engine speed of the thrusters to be controlled independently and also capable to be synchronized.

Steering motor angle indicators for both port side and starboard side shall be provided in wheelhouse.

A podded electric type or electric outboard steerable propellers thruster indicator for indicating turning angles (both port and Stbd) shall be provided in wheelhouse. Steering arrangement shall cater to manual mode of operation during emergency with local indicators.

3.5 MAST

Mast shall comply with air draft limitations to carry mast head light, anchor lights, towing light and other appropriate navigation lights and shapes as per IV Rules. Mast may be collapsible if not interfering with day to day operation of the vessel.

3.6 LIFE SAVING APPLIANCES

Life-saving appliances shall be as per Classification, subject to minimum 100% life jackets for passengers and crews child jackets to be provided 10%. 10% additional life jackets in each category to be provided and 4 nos infant life jacket to be provided. All lifesaving appliances shall be type approved by Classification society.

All life-saving appliances shall have operation manuals.

3.7 FIRE FIGHTING SYSTEM

Fire fighting system shall be in accordance with IV Rules & Class Rules. For the Battery system, Fire fighting and monitoring shall be incorporated in the module system complying class rules. Cell level monitoring for thermal Run-away to be incorporated. In case of a thermal run-away, by any reason, containment system to be provided, as per class requirement.

3.7.1 BILGE PUMP

Each compartment including void spaces shall be capable of being pumped out with positive type (or submersible centrifugal type) individual bilge pump as per rules, and shall have separate bilge water level sensor for high flood level alarms integrated with bilge pumps. The high-level alarm audio and visual indicators shall be provided in the wheelhouse console. If bilge pump proposed is of replaceable type 10% spare shall be provided for the whole lot. Arrangement shall be as per Class rules.

3.7.2 FIRE DETECTION AND ALARM SYSTEM

One (1) set of fixed fire-detection and alarm system shall be provided.

Sensors are to be provided in unattended spaces and passenger cabins as per class/statutory rules. Both heat and smoke detectors shall be installed in Battery rooms, machinery rooms and all other critical areas The system shall be addressable type and shall be type approved. Risk assessment for Battery room to be carried out by Class and recommended system to be installed.

Interface of the system shall be suitably ensured with the overall system integration and operating requirements. Fire Detection and Alarm System shall be interface with Single Control Unit of Automatic boat location system (ABLS) provided by Buyer.

3.7.3 FIRE PUMP

For fire fighting duties, two vertical self-priming motor driven centrifugal pumps (one standby) and one hand operated pump shall be installed as per IV rules and complying with Class rules. Starting of fire pumps should be interlocked with Generator starting so that Generator comes on load when fire pump is started.

3.7.4 STRUCTURAL FIRE PROTECTION

As per class requirement

3.8 TOOLS

Suitable minimum tools (as per each equipment manufacture requirement, yard standard and general requirement) for maintenance during normal operation shall be provided.

Metallic tools shall be with insulation.

Three sets of lifting cradles to be provided for the series of boats. Each set shall be suitable for lifting a boat with travel lift.

Two sets of Fire sensor testing/calibration tool shall be provided for the series of boats.

IV. ACCOMMODATION

4.1 PASSENGER CABIN

The Builder shall propose boat interior layouts, which incorporate a modern aesthetic approach with considerations to optimise passenger comfort, safety and security as well as to minimize noise in the cabin. For improving aesthetics, exterior stickers can be applied upon which design shall be mutually agreed with the buyer. If livery is applied same shall have a 5 year warranty. Superior quality of metro Interiors are to be matched and transferred to passenger cabin as far as possible. The clear ceiling height from top of deck flooring to ceiling shall not be less than 2.1m.

It shall incorporate wide double-leaf automatic doors at the aft deck, athwart ship face to face seating, centre lobby for standees, stainless steel grab-poles and rails, LED lighting, passenger information displays, public address loud speakers, and passenger alarm to permit passengers to make the boat operator aware of problems.

The body side and roof outer skin shall have a suitable thickness of approved acoustic and thermal insulating material bonded to their interior surfaces.

The design of interior fittings shall be safe under all conditions of passenger impact, during pitching and rolling movements of the vessel.

All non-metallic materials shall satisfy the requirements of flammability, toxicity and smoke emission limitations specified in FTP code to the extent required by class.

All interior surfaces must be finished with good blending and good slow ageing properties to provide a pleasant, high-quality interior and for ease of cleaning and maintenance.

All wall and ceiling panel surfaces shall be smooth finished with modern low flammability, low smoke emission, anti-graffiti, and low toxicity materials. Passage way including outside the passenger cabin shall be provided with ceiling panels. All internal panels shall be resistant to graffiti, scuffing, vandalism and cleaning agents.

Rounded corners or covings shall be provided wherever mutually perpendicular flat plane surfaces abut. Metal sticking strips of 150mm depth with radiused coving are required on all exposed vertical surfaces above floor level. Interior should be superior in quality and aesthetically pleasing. The colour shall not fade/discolour with time or due to rubbing. The Builder shall prepare detail plan during design and seek approval. The panels shall preferably be gel coated to avoid any discolouration. Hand laying process shall not be acceptable unless inevitable due to shape, etc. Contractor shall submit detail plan for approval. Details shall be discussed during design. Alternatively, powder coated Aluminium internal panels of suitable thickness adequately stiffened with superior finish can be considered.

Flooring shall be raised to the level of sill height or suitable provision shall be made for a seamless entry for physically challenging without barrier. Suitable ramps to be provided from deck, incase of raised deck.

All fastening devices, hinges, fixings and securing screws shall be stainless steel material (SUS316L) and not be visible from within the cabin. Gaps between all interior linings, Ceiling, panels kick strips, seat shell, etc shall not exceed 3mm. The effects of thermal expansion shall be taken into account and all unsealed gaps shall not exceed 1mm in depth where feasible. Suitable cushioning at panel joints shall be provided to suppress noise. Equipment cupboards for housing equipment, for which access from the cabin, if necessary, may be provided inside the passenger cabin and wheel house.

Gaps between all interior linings, panels kick strips, seat shell, etc may be sealed suitably. The effects of thermal expansion shall be considered. Suitable cushioning at panel joints shall be provided to suppress noise.

A dedicated space shall be provided near to entry/exit doors to accommodate 04 no. wheelchairs, complete with its occupant. Detailed proposals, including the need for a doorway ramp shall be submitted. Call button to be provided near the wheelchair accommodating area for warning for prolonged stay/ assistance as required.

The Builder shall provide all interior and exterior signage required by the buyer for approval. All interior/ exterior stickers/signage strips/logo, etc. used in any location shall conform to international norms. Marine quality Exterior sticker shall be considered for aesthetic appearance and design to be agreed with buyer. The safety related signage shall be fluorescent. The signage used for marking wheelchair shall be placed on floor as per the standard signage. An Ergonomic study shall be conducted for passenger cabin and wheel house. Study to be conducted on Anthropometric data, Grab Height & positions, Accessibilty for passengers with restricted mobility complying ISO 7176 & Wheelhouse crew seating height, visibility and access to the console.

4.2 WINDOWS

Cabin windows shall be provided and be flush mounted with exterior of superstructure.

Window units shall be modular units and shall be replaceable with minimum disturbance to superstructure.

Large window openings are preferred to permit standing passengers a wider view. The size of the windows shall be subject to review by the owner. Side windows shall be single piece in terms of visual perspective.

All side windows shall transmit less than 10% of the incident ultraviolet radiation and infra-red radiation. All windows shall transmit at least 50% to 55% of incident visible light.

Window seals shall be designed to prevent ingress of water to the inside of walls. Use of rubber extrusions is preferred to make the mounting watertight. The sealing material shall be selected that it lasts at least for the interval between major overhauls of vessel.

Scratch free (bothsides), high quality, lightweight are preferred for the window glasses like Perspex glass/ Plexiglas/toughened glass. Suitable arrangement for non air conditioned operation in case of emergency or pandemic situation to be considered.

4.3 CABIN DOORS

For entry/exit, wide electrically operated-sliding bi-parting doors with glass panels on upper half shall be provided at the aft end of passenger cabin. Passenger access Doors shall be remotely controlled from wheelhouse and in case of an emergency, be controllable locally. The controls of door opening and closing shall be integrated with passenger announcement system. Provision for wheelchair movement across sill to be provided. Standard width shall not be less than 1350 mm.

One no. door shall be provided on port/starboard side of wheelhouse cabin for crew access to bow part of vessel and one sliding door for entry from passenger cabin.

To ensure efficient climate control, passenger cabin doors shall also have provision for "local mode" to be activated from wheelhouse. The status of the door shall also be indicated (Red for open (unsafe) and green for closed (safe)).

- a) REMOTE MODE: The door OPEN/CLOSE shall be operated from wheel house.
- b) LOCAL MODE: The door can be locally opened with a push button and automatically closes after a preset time.
- c) EMERGENCY OPERATION: In case of system Failure / Emergency (Mechanical operation) to be provided.

Arrangement/Make to be approved by owner.

Door Indication lights: An amber colour Door Indication light shall be mounted adjacent to the Passenger Access Door to provide the visual indication of the working of the Door. The Door indication light complies to suitable standard. Doors shall have all safety features to prevent accidental closing while passengers passing including children

The different functions of this indicator will be working as follows

The lamp will be extinguished (Off) when the door is fully closed and locked.

The lamp will be illuminated when the door is in fully open condition or when the locking mechanism has failed, or doors are cut-offs / isolated.

The lamp will be flashing during the door closing or opening

When pulling down the emergency handle the indicator is illuminated. The controls shall be on Emergency (24V DC) System.

4.4 SEATS

Seating arrangement to be provided on either side of boat with alternate rows sitting face to face for ease of commuter movement. Seating to be provided for 50 passengers

+ 04 nos crew and four space for wheelchair to be arranged near to the door with suitable SS handrail/grab and securing arrangement. This area could have automatically folding seats for passengers, when there is no passenger on wheelchair. A curtain arrangement shall be provided in the area to use it for breast feeding. Foldable seats shall be considered in way of wheel chair arrangement.

The seats shall be designed to ensure they are comfortable with lumber support & aesthetically pleasing, easily cleanable, repairable and changeable and will not adversely affected by normal cleaning agents. Durable, lightweight, fire-resistant and vandalism proof mountings shall be capable of withstanding the loads arising in service conditions.

Sitting surface may be FRP or ABS plastic of high-quality finish and endurance.

The colour scheme and design shall be selected by the owner during design.

Seats shall be tested against load test and indentation test as per ISO 2439. The indentation hardness shall be similar to industry standards. The indentation hardness and depth shall be measured first be tested initially and then at 80000 cycle intervals.

Unpainted gel coat seats shall be used. A very high order gloss level shall be achieved and permitted gloss level shall not be less than 80 measured as per relevant standards. The seats shall be subjected to endurance tests up to 300,000 cycles. The seat and its

base shall be provided with suitable arrangement to easily lift it for maintenance purpose.

Pilot seat shall be cushioned, non slippery, ergonomically designed with back/lumber support using non-flammable materials and filling, and fully adjustable in the longitudinal and vertical directions with adjustable footrest. The seat and footrest shall be harmonized and shall be suitable for anthropology of Indians to ensure comfortable seating for the Pilot.

4.5 GRAB POLES AND RAILS

Stainless steel grab poles brush finish (SUS 316L) and rails shall be provided in the standing areas of the cabin for the comfort and safety of standing passengers. The grab rails and poles shall be so secured as to securely last for vessel lifetime.

Grab poles shall have not less than three vertical arms for increased accessibility, however, space between the vertical arms shall be limited to ensure that no entrapment of any part of the body is possible.

Sufficient margin shall be available in the top guide so as to ensure that under extreme loading conditions, the grab poles shall not become free.

Stainless steel grab handles with five handles per meter of grab rail shall be provided. Additional stainless-steel grab rails shall also be provided at the entrance of each door and also near gang way as decided by the Owner.

Grab rails in cabin area shall be brush finished.

Grab handles shall be subjected to "pull off test" to measure the maximum load they can withstand. A minimum load of 2500N shall be achieved. The fully assembled grab handle shall be subjected to an endurance test.

a) Load: 35 kg vertical

b) Cycles: 300,000 cycles minimum

c) Bending angle: ±45 degrees

d) Frequency: Each cycle to consist of movement of the handle from one extreme to another extreme and back within one second.

Any other test considered essential for the grab handle test to be included in the test plan.

4.6 LUGGAGE SPACE AND STORE

A centrally located dedicated luggage space shall be provided inside the cabins for passengers. May be arranged between central row of seats on both sides. A suitable

pantry table cum luggage space shall be provided outside the cabin for use of cargo parcels and use as pantry when vessel is used for alternate purposes.

A suitably sized electric room shall be provided in combination with DG room

Reasonable area aft of accommodation to be arranged as a deck space for use as a tourist deck.

Life jackets for passengers shall be arranged below each seat and within the luggage rack box For the standing passengers, life jackets may be kept on sealing racks on both sides with suitable signage. Child jackets/ infant jackets may be stored separately in a prominent space.

Provision shall be provided for vanity bag hooks and bottle racks.

4.7 LED DESTINATION BOARD

The LED dot matrix display board shall be provided on both sides inside the passenger cabin which shall be visible from outside for displaying the route/destination. The minimum technical features of the LED display board as follows.

Type of Media	LED
Colour	Amber
Light Intensity	>=2000 cd/m2
Resolution	244(w) x16 (h) dots
Dimension	1.4 (w) x 0.96(h) M
Angle of view (vertically)	120 deg. Perpendicular to display
Angle of view (horizontally)	120 deg. Perpendicular to display and shall be legible under direct sunlight, artificial light and darkness
Warranty	50,000 hrs at 70 L

4.8 FLOOR

Inside the cabin shall be arranged with a raised flooring to be even grade with the sill height or otherwise through a suitable arrangement to enable barrier free access for the wheelchairs, without compromising on the aesthetics/ convenience of passengers. The arrangement shall meet class requirement. Suitable Deck coverings shall be

provided for ease of maintenance, ease of maintaining cleanliness levels. The decks shall be smooth and fair when deck coverings are applied.

Outside the cabin, durable, anti-skid vinyl/rubber/WPC flooring suitable for weather deck shall be provided.

Sticking strips of 150mm depth with radiused coving are required on all exposed vertical surfaces above floor level.

The selection of deck covering materials and colours shall take into account the durability, low flame spread, cleanability, and maintainability of such materials. The materials and colours proposed by the Builder shall account for the fact that many areas of the Vessel will be subject to heavy foot traffic. Materials shall be selected based on their ability to hide such heavy use, yet the materials must present a pleasing appearance and contribute to the overall decor. Another factor that shall be considered in the selection and installation methods of these materials is the ease of future replacement as the deck coverings wear out over time. Builder shall submit the details of flooring material for owner approval.

Channels or other fittings used for the movement of chairs/ fixing of tables, if any should not hinder the passenger movement in any way and should be at grade with the flooring. Any such fittings shall be easily cleanable.

Builder shall propose suitable cleaning machines and liquid (with suitable local equivalent) and demonstrate the cleaning procedure to the satisfaction of Owner and without any damage to flooring on long term basis. The cleaning machines shall be easily accessible to clean floors below seats. The flooring shall have high abrasion resistance as per ISO-4649 and shall be compliant with relevant EN/ISO standards.

4.9 WHEEL HOUSE

The wheelhouse shall be located ahead of the passenger space. The wheelhouse shall be provided with shades and wiper. Electrical (24V) Window screen wipers/washers shall be fitted on front window(s) of helm position with synchronized switch panel on the operator's console. Wheel house shall be separated from the passenger area.

Wheelhouse should offer very good all-round visibility for navigation. All start/stop functions for equipment shall be remote controlled. The controls for foldable mast and entry/exit doors shall be provided at wheelhouse. A false-flooring of suitable height shall be provided in the wheel house to provide elevated deck in the wheel house compared to passenger area.

4.10 AIR CONDITIONING SYSTEM FOR PASSENGER CABIN

The passenger cabin and wheel house shall be air conditioned as per designed heat load calculation. The units shall be Roof top mounted and designed ensuring passenger comfort and ease of maintenance.(Provision should be made for easy access

to accommodation top). The duct shall be split from end to end to distribute air evenly throughout the length of the cabin. In the event of failure one HVAC unit conditioned air shall be made available throughout the length of cabin including wheelhouse. Fresh air mode and recirculation mode to be considered for cabin air-conditioning. The system shall use refrigerant of low global warming potential. Heat insulation to be considered as required to ensure 20 deg C. System shall have 2 Nos. of independent rooftop modules each having 70% of the total cooling capacity requirement. For battery bank calculations minimum (5kwh absorbed energy) to be considered. Design of the HVAC shall be submitted for owner approval. Air conditioners for Battery compartment shall be with sufficient redundancy for operation.

V. MACHINERY

The operation of the machinery shall comply with the requirements of the classification Society.

Main Machinery shall be designed on the basis of the following conditions, unless otherwise specified hereafter:

- a) Seawater temperature 32°C
- b) Ambient temperature 45°C
- c) Relative humidity 90%
- d) Salt and chemical corrosion as found in a tropical coastal area
- e) All the machinery/equipment's shall be air cooled or cooled with closed loop chiller system with Air cooled heat exchanger. River/Sea water for cooling shall be avoided as far as possible.
- f) Materials used in electrical components shall be highly resistant to corrosion and erosion on long term, if water cooled system is required.

The plant shall be designed without any operating restrictions in tropical condition.

Special attention shall be paid to a logical arrangement of accessibility of all installations in structural components. Provisions shall be made to ensure that all repairs and maintenance can be performed in the easiest way.

Heat exchangers shall be designed for a seawater temperature of 32°C and for the fresh-water cooling circuit in accordance with the heat balance calculation of various systems. Freshwater coolers shall be designed with the fouling factor of 20%.Low pressure Indicators to be provided for sea water cooling pump.

Machinery monitoring system shall be interfaced with Intelligent Transport Management System provided by Buyer.

5.1 MAIN PROPULSION PODDED ELECTRIC TYPE OR ELECTRIC OUTBOARD STEERABLE PROPELLERS

Two Reversible Permanent magnet synchronous electrical outboard steerable propellers or permanent magnet Podded electric propulsion shall be provided. Provision shall be made to tilt/lift the system in order to get access to propellers for cleaning / removal of entangled weeds etc from the deck/platform The propulsion system shall be suitable for continuous operation without any failure at rated power. A sea margin of 15% to be considered for powering and 10% more propulsion power shall be considered for compensating shallow water effect including battery capacity. Power rating of the propulsion machinery shall have 20% margin.

Low maintenance and high service intervals shall be a criterion in the selection of the propulsion system.

A Power management system shall be installed to manage all types of power requirements. The wheelhouse shall be provided with all remote controls & remote reading gauges for the machinery installed in the engine rooms.

Shipyard shall provide detailed system configuration and operation detail confirming adequacy of technical requirements. Other parameters can be interfaced to main machinery monitoring system.

Standards:

Hazardous Area Classification	Non-hazardous
Standard	IEC
Insulation Class	F
Temp Rise	F

All motor's data is subject to tolerances in accordance with IEC.

Ratings and Performance Data

Numbers	2
Converter Supply	Variable Frequency Drive
Voltage	V 690/any suitable voltage
Frequency	Hz 50/60

5.2 CONTROL INSTRUMENTS

The diesel generators shall be equipped with sensors for alarm monitoring and trips in accordance with the classification society. Safety trips such as Lub oil low Pressure, High Lub oil Temp, High Exh temperature, High cooling liquid temperature etc to be provided as standard.

An Alarm monitoring panel shall be provided with all major alarms of propulsion motors, main generator engines and battery monitoring and management system and hybrid propulsion system. All the parameters of the diesel generator, battery Bank Power Management System and Propulsion machinery shall be interfaced with operation control centre reporting system for monitoring from shore.

There shall be one remote control for start/stop for diesel generator in the wheelhouse and one emergency control at the local position.

Temperature readings of engine room and battery compartment shall be available at the wheelhouse and the local position.

Provision shall be made so that podded electric type or electric outboard steerable propellers should be able to operate independently.

5.3 EXHAUST SYSTEM

Diesel exhaust line gas shall be insulated as per class, lined both sides covered with galvanised steel sheets.

For diesel generator, a separate exhaust-gas pipeline shall be provided. All exhaust pipelines shall be stainless steel and as short as possible and the bends not to be too sharp, to avoid back pressure.

Opening height of the exhaust pipe should be above the top of superstructure, to avoid the exhaust gas entering accommodation in case of change in wind direction.

An expansion joint is to be fitted immediately on the turbocharger exhaust outlet to compensate the thermal expansion.

Generator engines/ compartment to be acoustically insulated to have minimise noise. Noise level to be limited to 85 Db.

5.4 MAIN MACHINERY MONITORING SCREEN

Monitoring screens are to be provided, in the wheelhouse. This shall contain the monitoring and control console with the necessary alarm, shutdown, indicating and monitoring units. The control panel shall be installed according to the class requirements. Separate screen for P&S system of 18-inch screen shall be provided.

5.5 DIESEL GENERATOR SET

One no Diesel generator shall be supplying 415 V or voltage as suitable, 50 Hz 3 phase to cater to the operating requirements of 50 Pax boats.

The fuel used shall be HSD/LS HFHSD and shall be integrated for continuous operation. The DG shall be complying with Tier 3 emission norms/ equivalent. The DG shall be class approved for marine use. Generators shall be preferably radiator cooled or water cooled with closed loop system with air cooled chilling unit.

The alternator shall be continues rated, brush less, Permanent Magnet Generator (PNG), self-exciting, self-regulating with revolving field, in-ventilated, drip-proof design, Automatic voltage Regulation and with damper-windings included.

Generators shall have proper acoustic insulation to ensure minimum vibration and sound. The noise level shall be limited to 80 DB. Where DG set is not covered (package type), scanning of hot spots around DG set shall be done and areas where temperature is more than 220oC and above to be protected for prevention of self-ignition/fire.

The generator capacity shall be optimised for minimum fuel consumption at low loads. Variable speed diesel generator system usage shall be explored. Generator powering shall consider propulsion losses (appx 5 %) & sea margin (appx 10%). The generator power shall be capable of propelling the boat at service speed of 7 knots at 85% MCR rating.

Necessary provisions for interfacing with Power Management System for necessary control (Auto ON/OFF based on load) and monitoring and OCC reporting system on shore shall be provided.

All the critical parameters of the Diesel Generators shall be displayed on local and remote panel. Controls like ON/OFF, Emergency OFF shall also be provided. Minimum parameters shall include Engine speed. RPM, exhaust temperature, cooling temperature, LO temperature and pressure, running hours, fuel consumption power and voltage. An audio-visual alarm shall also be provided with alarm for all critical parameters

The engine to be completed with:

- a) Air filter(s), marine type
- b) Duplex changeover lube. Oil filter with drip tray
- c) Lube oil cooler, if required
- d) Cooling systems with thermostat as applicable
- e) Duplex changeover fuel-oil filter with drip tray
- f) Flameproof flexible fuel-oil hoses

- g) Battery for automatic start shall be maintenance freewith 3 years warranty
- h) Anti-vibrations to ensure minimum vibration
- i) Anti-condensation heater(s) for alternator
- j) Gutters of at least 50 mm height, should be provided all around the generator to avoid the spread of fuel/lub oil in case of any leakage

5.6 FANS AND VENTILATION

5.6.1 ENGINE ROOM FANS

The machinery room shall be ventilated mechanically as per rules. Fans shall be of axial flow reversible type

5.6.2 AIR SUPPLY AND EXHAUST DUCTS

All air-supply and exhaust ducts to the engine room shall be provided with fire-flaps, suitable to be operated from outside complying with rules.

5.7 PIPING AND FITTINGS

5.7.1 GENERAL DESIGN MEASURES

The piping is to be designed on basis of the Piping Diagrams. The actual pipe laying is to be done with shortest ways and best maintenance possibility. As far as possible elbows in the piping system shall be avoided and same shall be replaced with bends. This is to avoid pressure drop and resistance to flow.

Pipes and flanges used shall be approved in accordance with class rules and best shipbuilding practices.

The piping system is to be suitably designed to allow expansion for thermal and/deflection of hull structures by means of expansion joints or bends.

Piping shall be laid as straight as practical with a minimum number of bends and joints.

All the Out-Board Discharge Valves shall be Screw Down Non-Return Valve and the values shall be suitably located to ensure easy operations. All river/sea suction valves located at underdeck shall be operable remotely from deck. Body of the Duplex filters for sea water cooling system, if used, shall be of stainless steel.

Duplex sea water filters shall be with differential pressure indication in wheel house.

5.7.2 AIR AND SOUNDING PIPES

Air and sounding pipes shall be provided for all tanks and void spaces as per IV rules and Class. A striker plate shall be provided below the sounding pipe.

5.7.3 DECK FIRE LINES

Fire-and deck wash lines shall be arranged according to Class. Deck washing connections on deck shall be provided to connect water lines from shore/jetties.

5.7.4 DIESEL EXHAUST GAS LINES

Diesel exhaust-gas lines shall be insulated as per class, lined both sides with galvanized steel sheets.

5.7.5 OTHER PIPES CONVEYING HEATED MEDIUM

Other pipes conveying heated medium to be insulated according to the necessity. In general, also the flanges are to be insulated.

5.7.6 DIESEL FILLING POINTS

Diesel filling points shall be provided on both port and starboard at suitable location for filling up the diesel oil storage tank with secured closing device.

5.7.7 PIPING MATERIALS

Bilge System	UPVC/ ABS Plastic as appropriate
In machinery spaces	316L stainless steel
In other Spaces	316L stainless steel or aluminium or ABS
plastic	as appropriate
Fire-Fighting System	316L stainless steel
Fuel Oil System	316L stainless steel
Domestic freshwater system	ABS plastic/polypropylene
Domestic sea water system	ABS plastic /polypropylene
Sewage and drainage system	UPVC / ABS plastic
Engine cooling system	316L stainless steel
A.C Refrigeration piping	Copper with insulation to prevent
Condensation	Air, filling and sounding piping: n316L stainless steel

VI. ELECTRIC SYSTEM

All the electrical equipment and installation on the Vessels shall comply with the requirements of the Classification Society. Special attention shall be paid to:

- a) Mutual interference of static converters connected to the same bus bar system
- b) Voltage distortion and reacting to other consumers
- c) The selection of the ratio between the sub-transient reactance of the system and the commutating reactance of the static converter
- d) Consideration of reactions from rectifier installations
- e) Influence by harmonics and high-frequency interference
- f) All sea water cooling lines shall be away from electrical systems as much as possible.

6.1 ELECTRICAL NETWORKS

The electrical system shall essentially have 4 type networks:

- a) 700V DC System (or any other suitable voltage)
- b) 415V 50 Hz 3 phase three wire insulated System
- c) 230V 50 Hz 3 phase three wire insulated system.
- d) 24V DC System for emergency backup.

6.1.1 DC SYSTEM

The propulsion system and Battery bank shall be deigned at 700 V DC (or any other suitable voltage). The systems shall be designed such that the port and starboard side propulsion systems shall act independently.

6.1.2 415V 50 HZ 3 PHASE THREE WIRE INSULATED SYSTEM

415V, 3PH, 50 Hz, three-wire insulated system, or any suitable power supply voltage is the Main power source. Power distribution system shall supply power to each consumer's load through main switchboard (MSB) directly and/or through distributing boards.

6.1.3 AC 240V SYSTEM

240 V 3PH, 50 Hz, three-wire insulated system-shall be used for powering all the Main lighting system and other machineries. It shall also act as the power source for 24 V DC rectifier unit.

The system shall be designed with two identical networks including main transformers with power sources isolated from each other. The network shall be able to select the power source using a source selector switch.

6.1.4 24 V DC SYSTEM

The 24 V DC system shall have 2 Nos transformer-based rectifier unit (one for redundancy) which would act as a source for all 24 V DC systems for the boat. The 24V DC System battery charger / rectifier shall be type approved. Emergency battery backup shall be provided for emergency power supply if separate emergency source is considered. The Emergency battery shall be Li Ion maintenance free type, with minimum 5 years warranty. In addition to the class requirement, CCTV cameras, All Navigation and Communication systems, Portable Bilge Pump and ITMS system (300W) shall also be considered in Emergency supply. Parameters like Battery temperature, SoC Voltage Shall be displayed on the Wheel house Console. Battery charging provisions as per Class requirements shall also be provided.

6.2 POWER SOURCE

The power source for propulsion and hotel loads for the vessel is powered by:

- a) Lithium based (LTO) Battery Bank
- b) 415V, 3PH, 50 Hz, Diesel Generator
- c) Shore Charging Point (to be set up by successful bidder at selected location)
- d) Shore Supply Panel

6.2.1 LITHIUM BASED BATTERY BANK

The boat shall be provided with adequate sets of class approved Lithium based battery (chemistries LTO) storage for propulsion. The battery shall meet the best of safety standards and shall be resistant to explosion in a piercing test. The battery shall have a warranty of 32000 full cycles. Suitable depth of discharge shall be recommended by the builder to maximize the said life span. Permitted charging rate shall be 2C and above. De-rating of battery over the life span shall be considered while deciding the battery capacity. Battery chemistry shall permit such charge acceptance rate so that the battery can be brought up to 80% SOC within a period of maximum 15 minutes of shore charging including connection/ disconnection time.

The batteries shall be cooled and space ventilated / air conditioned as required so that the above life expectancy shall be maximized. The cooling circuit temp to be

maintained using an air cooled chiller unit if liquid cooled. System shall be preferably air cooled.

When boat is not in operation the system should be adaptable to complete shutdown including cooling, except safety and monitoring system.

Boat design and battery compartment shall cater for size, weight, volume of the batteries. The propulsion system design shall cater for current marine tested (class approved) and commercially proven chemistries available i.e. The design shall cater for changing over to any proven and tested chemistries at a later date.

Battery state to the cell level shall be monitored continuously and remote state of charge readings shall be available in wheelhouse. Battery temperature shall be monitored and suitable thermal over run warning/isolation shall be provided.

6.2.2 GENERATOR

The generator shall cater for the following:

- a) During normal operation, the ship's power (propulsion and hotel/auxiliary load) shall be supplied from the battery bank for meeting the requirements of 7 knots service speed. When the operating power is to be increased beyond 7 knots, the additional power required beyond 7 knots shall be provided from the generator in Hybrid mode. The battery/power management system shall automatically bring the DGs online for the additional speed as demanded by the operator. The system shall be so designed that the battery and DG shall operate together without any interruption during the period the DG is online. In the event additional power demand is removed (i.e. reduced back to service speed or below), the DG shall go automatically offline and stop subsequently.
- b) In case there is a failure in the battery system or drop in SOC of Battery system beyond the residual capacity of 20%, the power required for meeting the service speed requirement of 7 knots shall be fed from the DG. Battery shall have option of simultaneous charging at 0.25 0.15 C rating
- c) Generator SFOC shall not be more than 240g/KW Hr

6.2.3 SHORE CHARGING POINT

Standard EV (Electric Vehicle Supply E) chargers with CCS 2 protocol are to be provided by successful bidder for the charging system on the shore. The shore connection will be provided by IWAI, RO Kolkata. The bidder has to ensure its connectivity to the shore connection by IWAI.

The charging current shall be limited to 150-200 A. The battery system voltage to be selected accordingly, inter-operability between battery bank and charger shall be proven.

The Vessel shall have Shore Charging panels/terminal suitably located on the port and starboard side for both banks for easily connecting it to the shore facility. The shore charging arrangement shall be simple and easily handled for achieving connection (manual) in a very short span of time and positioned such that the Vessels could be connected in fore and aft berthing directions on the same jetty.

6.2.4 SHORE SUPPLY PANEL

Shore supply panel shall be provided for Supply of 415V 50 Hz 3 phase Supply from Shore to cater the Hotel loads and Slow charging of Battery Bank. The battery shall be charged at 0.25 C rating during slow charging. Necessary Interlocks with other systems shall be provided to ensure safe operation. The panel board shall be equipped with voltmeters, ammeters, watt meters (KVAR and KW), power factor meter, frequency meters, Phase sequence Meter's etc.

6.3 ELECTRIC INSTALLATION

6.3.1 ELECTRICAL PANEL/ SWITCH BOARDS

Each Main distribution panel shall be equipped with voltmeters, ammeters, watt meters (KVAR and KW), power factor meter, frequency meters etc. At least two no's spare breaker capacity shall be provided. All the measuring instruments, changeover switches, buttons and pilot lamps, shall be as per class requirement. All the main panels or relay contact for emergency operation shall be located in the wheelhouse.

All the bus bars shall be of copper, and designed laying emphasis on maximum system operational efficiency & minimizing components, weights. The bus bars shall have proper insulation cover wherever required.

The MSB shall be located at the Electrical/DG room. All the main protection/control of the electrical system of all the major equipment's of the boat shall be in the MSB. Boat's equipment, which are installed in pairs (redundancy), shall be connected to separate sections so that uninterrupted operation is assured.

The determination of setting value for protection devices of distribution circuit shall be on the principle of "at first to cut-off protection circuit closest to consumer load" and then "to cut-off protection circuit closest to MSB". Power distributing boxes and lighting distributing boxes shall be suitably located. All light operating switches shall be provided on wheelhouse.

6.3.2 ELECTRICAL CABLES

Installation of all cables and electric equipment shall be carried out according to Class requirements and owner satisfaction. All cables shall be class approved. All the cabling and wiring installations in the wheelhouse and passenger area to be concealed type. Each cable shall be marked with indelible and non-corrosive cable markers for easy identification. The cable markers shall be clearly visible after cleating and

strapping. Individual conductors and wires shall be identified with a closed, slip-on type plastic cable identification ferrule, carrying a number identical to the terminal. This is also applicable to the single conductors for cross-wiring between the terminal blocks in panels. Multi cable transits (MCT) and cable tray material shall be selected such as to avoid eddy current and galvanic corrosion. No electrical equipment, components or cables shall run through or be installed inside tanks' compartments.

6.3.3 MOTOR

All motors of the ship shall be of marine type. The protection class of the motor on the main deck is IP56 and that of the machinery room shall be IP44.

6.3.4 CONTROL AND MONITORING EQUIPMENT

All the control and monitoring devices shall be according to the power and operational requirements. Necessary automatic/manual control change over switches and local/remote control change over switches shall be provided. The Main switchboard and all other main electrical panels shall be located above the deck inside the electrical room/cabin.

The propulsion motor shall have individual variable frequency control system for speed control of each motor. Each power source (Battery and Genset) and propulsion motor shall have control and monitoring console in the wheelhouse. All the major parameters like motor RPM, Input power, temperature, Efficiency shall be provided.

The battery management system and power management system shall monitor and control all the operations for the boat. Relevant data required for live monitoring and analysis shall be stored and transmitted to OCC via WiFI/ GSM network (ITMS system). The builder shall provide the necessary backend software for OCC to monitor the transmitted data.

6.3.5 WHEELHOUSE CONSOLE

The central boat control console shall be located in the wheelhouse. The central boat console shall have all the required control and monitoring devices including alarms, cut outs and emergency stops for all the electrical installations of the vessel.

The navigation and communication aids of the vessel shall also be located on the wheelhouse console. The console shall be user friendly and ergonomically designed. The design and positioning of displays shall be such that glare on the panels are minimized and clear visibility ensured in under bright sun day. The detail arrangement of the same to be approved by the owner.

6.3.6 LIGHTING SYSTEM

The Builder is to ensure that the power consumption for lighting shall be kept to a minimum whilst adhering to the Lux levels. The lighting circuit shall be designed such that the lux level inside the passenger cabin can be controlled from the wheelhouse.

The passenger saloon lighting shall consist of LED lamps powered by 230 V AC and interleaved into longitudinal strips located in the ceilings, arranged in such a way that it extends along the full length of the saloon ceiling. The lighting installation shall be done in a manner which provides uniform lighting level and avoids glares that could disturb passenger vision.

The lighting aesthetics shall be attained either by using LED strips with Opal acrylic diffuser or with individual LED Fixtures integrated and arranged linearly. If LED strip with diffuser arrangement are used, the designer shall ensure that the luminaire shall be continues (shadow-dot free). The surrounding temperature in the saloon shall not to be significantly increased and the noise generated by lighting component shall be inaudible.

The luminaire shall be designed with Snap fit diffuser for easy maintenance from bottom without removing the full fixture, housing shall be made of either aluminium or polycarbonate. The LED drivers shall be vibration proof and shall be suitable for transportation applications. Housing of the lighting devices shall prevent entry of dirt, moisture and insects.

The saloon interior lighting intensity shall be uniformly distributed, and the lighting level will not be less than 150 lux at any point in the saloon during the normal operation. A constant level of illumination of 250 lux at a height of 1 m above floor level shall be maintained. The Energy Calculations for LTO battery bank calculations shall be done on a constant level of illumination of 250 lux at a height of 1 m above floor level.

The lighting control switch is located in the Cab. The switch shall be manually engaged to control the illumination level in the saloon. The lighting shall be controlled to different level viz.: 150-200-250-300 Lux. preferably using DALI.

The saloon lighting lines are to protect by circuit breaker. The system design shall ensure that single point failure in network shall not result in complete failure of lighting system. In case of total failure of main power (emergency mode) the lighting of the cabin shall be maintained at a reduced illumination as per class requirements.

The luminaire shall have Burning hours: 50,000 Hrs at 70 L Standard Deviation Colour Matching (SDCM) shall be less than 5. The Luminaire System efficacy shall be greater than 110 lm/W.

The system shall be easily replaceable, and spares shall be readily available. The Light design details and fixture details shall be submitted to client for approval.

All the light fixtures shall be LED type arranged aesthetically well ensuring maximum comfort to the crew and passengers. Illumination level measurement shall be taken at

a height of 1 M above the deck/ grating level. Minimum two no of light fixtures shall be placed in all compartments of the demi hull. Explosion-proof lighting fixtures shall be provided in dangerous areas viz battery room etc. Average illumination shall be as follows in the compartments indicated below:

- a) Engine/ Generator Room 300 Lux
- b) Propulsion Equipment Room 300 Lux
- c) Machinery Rooms --- 300 Lux
- d) Battery Room 300 Lux
- e) Switchboard Room --- 300 Lux
- f) Wheelhouse 300 Lux (10 lux special red light required for night operations)

The emergency lighting system shall be 24 V DC supply, automatically triggered in case of failure of main lighting system. The emergency lighting system shall cater to the necessary lux level defined at various locations and all the navigation lights of the boat.

6.3.7 LED DISPLAY SYSTEM

- a) To provide LED display system showing "ENTRY" and "EXIT"
- b) To provide LED display showing "EMERGENCY EXIT"
- c) To provide LED display showing route map indicating the boat jetty names

6.3.8 POWER POINTS

USB Charging sockets: USB charging ports shall be provided in the cabin for passenger usage common for each double row on both sides.

6/ 16 A Sockets: Passenger Cabin should also have 8 nos. (4 each on fore and aft) 6/16A socket with suitable protection and secured from passenger access to be used for cabin cleaning machines, etc. At least 2 Nos in wheelhouse to be provided additionally for crew. A total of 4 Nos to be provided on Aft open Space. The total absorbed energy shall be considered as 3000 watts per Hr.

6.4 NAVIGATION AND COMMUNICATION SYSTEM

All navigation and communication equipment are to be fitted as per IV rules and Class Requirement. The equipment's shall be suitable for use in Marine Environment including on board vibrations rolling, pitching and at ambient temperature of 40-degree C. Essentially following items are to be provided.

6.4.1 CCTV SURVEILLANCE SYSTEM

The CCTV surveillance system should consist of IP HD night vision cameras with minimum 8MP with IR, software, servers (NVR), power supply and communication cables PoE switches and other required materials to complete the project. The system / equipment's shall be rugged, durable, compact and suitable for marine application. These cameras shall be designed to work on 24 X 7 basis and shall capture the video feeds at 25 FPS 1920 x 1080 p resolution during entire duration of day. The system shall have live monitoring in the wheelhouse console and have a storage capacity for 30 days (cloud /in-house storage) with the provision to automatically overwrite the data in chronological order.

Recording frame rate and resolution in respect of individual channel shall be programmable. A Video stabilization module shall be installed to ensure the quality of CCTV camera outputs. The display screen size shall be minimum 32". The Surveillance coverage scheme/Field of View shall be submitted for owner approval. The system design shall ensure no single point of failure. Suitable redundancy shall be provided for all the functionalities in the system to include Recording and Management Server System. The data shall be transmitted via WiFi / GSM network for remote monitoring and storing in Operation Control Centre.

Coverage area Surveillance: 100% coverage on All machinery space, Passenger Cabin & Wheelhouse

Navigation Aid: Port, Stbd, Fore and Aft of the boat covering areas for reversing, and mooring area.

6.4.2 PASSENGER ANNOUNCEMENT SYSTEM

All equipment shall be of modern design, fully electronic, modularly constructed and with a proven field performance. An Amplifier, 'gooseneck' microphone with 'Pressto-Talk' button shall be provided on Console and a handset/headset microphone socket (all accessories — Bose) shall be provided on Wheel house and Passenger area with sufficient power capacity. The ambient noise-sensing device shall be provided for automatic level control shall be of continuous monitoring type based on time averaging. Multiple noise sensing device may be provided if wide noise level differentials exist in a zone.

Speakers shall be installed at passenger areas/crew area (as applicable), cabins, and engine room. The design of the PAS shall include a method of spectral enhancement to improve the clarity with which broadcasts are heard by those with frequency selective impaired hearing (i.e. reduced frequency discrimination in background noise). PA System shall be designed in such a way that zone wise speaker evenly distributed to the amplifier so that one amplifier fault shall not contribute to entire PA system faulty.

The Systems shall be so designed as to have a minimum expandable capacity of 20% without any modification in software, software license, Ports etc.

All equipment's shall be made up of either Aluminum or non-corrosive materials considering marine environment.

All elements of the PAS shall comply with the requirements of BS5839-8. The PA speech broadcast quality shall achieve a minimum Speech Transmission Index (STIPA) of 0.5 of the space in accordance with BSEN 60849 as embraced by BS5839-8. The Loudspeakers shall comply with all relevant provisions of IEC 60065.

The yard shall submit a detail report, for approval of the Employer, on the STIPA performance and same to be verified after installation.

Further detailing shall be done during detail design phase. The Passenger Announcement system in the boat shall interface with ITMS system to provide information to passengers. A detail interface document shall be prepared in consultation with owner.

6.4.3 COMBINED NAVIGATION SYSTEM

Combined Navigation System shall have below listed Items:

a) Multi-Function Display

The boat shall have a Multi-Function Console with the capability to providing a chart overlay and integrate data from Radar GPS, Echo sounder, AIS. Compass and thermal camera. The display shall have a minimum of 18" size with display resolution of 1280 x 800 WXGA or higher and clearly readable on daylight. The viewing angle shall be minimum 80 degree Top, Bottom, Left and Right. Data input/MFC functional control should be available in form of multi touch screen and a mini flushed keyboard or a joystick device, which should be able to withstand marine environment with IP 67/ IPX7. The screen should be able to be split thereby displaying thermal camera images, radar, chart and navigational data simultaneously.

b) GPS/Compass

The MFD shall give GPS co-ordinates of the vessel with accuracy better than 1m. The MFD shall display the speed of the boat, plot routes on the maps and show the boat heading direction with solid-state gyro electronic compass. The data shall be collected with built in or external transducers.

c) Doppler Radar

The navigational radar shall display clear raw indication of all moving and stationary contacts, coastline, creeks, navigational buoys and marks, harbour installations, jetties etc. and should be able to track at least 10 contacts at a time with the distance from the boat. The transmitter shall have X-band solid state transmitter with pulse compression technology. Frequency of

transmission between 9200 to 9500 MH. For Radome IP 66/ IPX6 or higher ingress protection standards shall be provided.

d) Thermal camera

The camera shall be Pan tilt zoom type. The detector type shall be 320×240 VOx Micro bolometer with 4x E-Zoom with a minimum field view of min $24^{\circ} \times 18^{\circ}$ and 320×240 Resolution. The Video Output shall be minimum H264 IP Video stream. The equipment shall be as per IEC 60945 with IPX6 rating. The system shall have object and range identification for easily identifying objects at night. Provision for displaying and storing the images for 30 days on CCTV SYSTEM with the provision to automatically overwrite the data in chronological order.

e) Sonar: Echo Sounder

The echo sounder shall be capable of operating in both shallow and deep water of inland waterways. The transmitter shall have the capability to tabulate all depth from the bottom of the hull. The depth shall be shown in the MFD with alarms on pre-set minimum depth.

f) VHF Marine Mobile Band (MMB)

The fixed VHF MMB shall ability to store at least 05 channels with one touch access to Ch 16 and should be able to operate for at least 8 hours continuously. It should have a speaker for reception audio output and have a flexi-cord attached with the microphone. The microphone should have a PTT (Press totalk switch).

6.4.4 LIGHT SOUND SHAPES

The LSS requirements shall be as per Class and IV - Statutory requirements.

a) Navigational Lights

Navigation lighting and signaling fixtures shall be provided as per requirement of the Rules and Regulations. Fail safe panel shall be provided for navigational lights. The bulb/driver shall be replaceable, and spares shall be provided for each navigation light.

b) Electric Horn

One No. 24V DC electric horn capable of making sound signals shall be provided. The driver unit shall be totally enclosed marine type with sufficient IP for outdoor installation. It shall operate from wheelhouse using push button

c) Search Light

The Search light shall be provided at a suitable location on the forward side of superstructure capable of 360 degrees' rotation (electrically) remotely operated from wheelhouse. Spare bulb shall be provided.

d) Loud Hailer

One portable battery-operated loud hailer with minimum power output of 5W rated and 10W maximum shall be provided

e) Binocular along with storage box

6.4.5 ITMS SYSTEM

The Buyer's appointed ITMS contractor shall supply the following ITMS (Intelligent Transportation Management System) items for installation in the boats,

- a) ABLS (Automatic Boat location system)
- b) Passenger Announcement system
- c) Passenger Information Display system
- d) Fixed Digital Mobile Radio (VHF) set
- e) Handheld Digital Mobile Radio (VHF) set

The power and data cables between buyer supplied equipment shall be supplied by buyer (ITMS contractor). All the cables, power source, distribution panel, etc. if required, for installation of ITMS systems except the ITMS contractor supplied cables, shall be supplied by builder for successful installation, completion and functionality checks.

Installation of all buyer supplied equipment, cabling, supports etc. shall be in the scope of builder. Cabling shall be concealed (to the extent possible). All the mounting accessories shall be compatible with boat material, corrosion resistant, aesthetically designed to match with all architectural finishes and of sufficient strength to mount the equipment securely. Testing of all cables including ITMS supplied cables, shall be done by the builder along with ITMS contractor.

The termination, Testing and Commissioning of the buyer ITMS equipment shall be in the scope of buyer appointed ITMS contractor. The builder supplied equipment/cables etc. termination, testing and commissioning shall be in the scope of builder.

Fire Detection and Alarm System shall interface with VMU (Vessel Mounted Unit) of ABLS (Automatic Boat location system). Critical boat machinery equipment parameters are to be integrated into the VMU for further transmission to the operation control center. Approximate 20 boat parameter shall be considered.

The overall schematic drawing and interface termination drawing shall be supplied by ITMS contractor for ITMS supplied items. The boat builder supplied equipment interface engineering, buyer equipment layout, co-ordinated drawing etc. shall be in the scope of builder. The boat builder shall supply installation method statement to buyer/ buyer appointed ITMS contractor.

The detail interface document shall be prepared by the builder together with ITMS contractor during detail design stage. There shall be a prototype sign off between boat builder and buyer's ITMS contractor for entire construction, testing and commissioning procedures for ITMS equipment.

The builder is required to install the above items as per the specifications to be provided by Buyer during execution. For boat design purpose, approximate weight of the ITMS systems equipment shall be considered as 150 kg.

Inland (Minist	Waterways Authority of India ry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
S	ECTION - VII: GENERAL CONI (GCC)	DITIONS OF CONTRACT
	(300)	
	124	

1. GENERAL PROVISIONS

1.1. Definitions

- 1.1.1. Employer means the Chairman, Inland Waterways Authority of India and his successors
- 1.1.2 Authority/
 Department/
 Owner

"Authority/ Department/ Owner" shall mean the Inland Waterways Authority of India, which invites tenders on behalf of the Chairman, IWAI and includes therein-legal representatives, successors, and assigns.

- 1.1.3 **Chairman** Chairman means Chairman of Inland Waterways Authority of India.
- 1.1.4 **Chief Engineer** Chief Engineer means the Chief Engineer of the Authority.
- 1.1.5 **Director** Director means the Director of the Authority, as the case may be.
- 1.1.6 **Deputy Director** Deputy Director means the Deputy Director of the Authority, as the case may be.
- 1.1.7 **Contractor** Contractor means the successful Bidder who is awarded the contract to perform the work covered under this tender document and shall be deemed to include the Contractor's successors, executors, representatives approved by the Engineer-in-charge.
- 1.1.8 Contractor's Representative

"Contractor's Representative" means the person named by the Contractor in the Contract or appointed from time to time by the Contractor, who acts on behalf of the Contractor.

1.1.9 Employer's Personnel

"Employer's Personnel" means the Engineer, the assistants and all other staff, labour and other employees of the Engineer and of the Employer; and any other personnel notified to the Contractor, by the Employer or the Engineer, as Employer's Personnel

1.1.10 Contractor's Personnel

"Contractor's Personnel" means the Contractor's Representative and all personnel whom the Contractor utilises on Site, who may include the staff, labour and other employees of the Contractor and of each Subcontractor; and any other personnel assisting the Contractor in the execution of the works.

1.1.11 Engineer-In-Charge/Engineer

Engineer-In-Charge (EIC) or Engineer means the Employers personnel authorized to direct, supervise and be in-charge of the works on behalf of the Employer.

1.1.12 "Subcontractor" "Subcontractor" means any person named in the Contract as a Subcontractor for a part of the Works or any person to whom a part of the Works has been subcontracted by the Contractor with the consent of the EIC and the legal successors in title to such person, but not any assignee of any such person. 1.1.13 Engineer-in- charge Engineer-in-charge representative/assistant shall mean any representative/ as officer of the Authority nominated by the Engineer-in-charge for assistant day-to-day supervision, checking, taking measurement, checking bills, ensuring quality control, inspecting works and other related works for completion of the project. "Contract" means the Contract Agreement, the Letter of Award, 1.1.14 "Contract" the Form of Tender, the Conditions (GCC & SCC), the ToR, the Specifications, the Drawings, and the further documents (if any) which are listed in the Contract Agreement or in the Letter of Award. 1.1.15 "Specification" "Specification" means the technical specifications and special conditions of the Works included in the Contract and any modification thereof or addition thereto made or submitted by the Contractor and approved by the Engineer. 1.1.16 **Party** Party means either the employer or the contractor as the case may be and "parties" means both of them. "Commencement Date" means the date notified under clause 8.1 1.1.17 Commencement of GCC. Date 1.1.18 for"Time for Completion" means the time for completing the Time **Completion** execution of and passing the Tests on Completion of the Works or any Section or part thereof as stated in the Contract(or as extended under Clause 8.6 of GCC) calculated from the Commencement Date 1.1.19 Day "Day" means a calendar day and "Year" means 365 days. 1.1.20 "GCC" means the General Conditions of Contract. GCC 1.1.21 "SCC" means the Special Conditions of Contract. SCC 1.1.22 **Work Order** "Work Order" means a letter from the Authority conveying the acceptance of the tender/offer subject to such reservations as

may have been stated therein.

1.1.23	Letter of Award	"Letter of Award" means the formal letter issued by the employer to successful bidder to notify award of contract and shall specify including other terms and conditions, total contract price in consideration of execution & completion of works and the remedying of any defect therein by the contractor in accordance with the terms and conditions of the contract.
1.1.24	Bill of Quantities	"Bill of Quantities" means the priced and completed bill of quantities forming part of the Tender.
1.1.25	Tests on Completion	"Tests on Completion" means the tests specified in the Contract or otherwise agreed by the Engineer and the Contractor which are to be made by the Contractor before the Works or any Section or part thereof are taken over by the Employer.
1.1.26	Interim Completion Certificate	"Interim Completion certificate" means a certificate issued on completion of work.
1.1.27	Final Completion Certificate	"Final Completion Certificate" means a certificate issued after completion of work as per the SOW
1.1.29	Cost	"Cost" means all expenditure reasonably incurred (or to be incurred) by the Contractor, whether on or off the Site, including overhead and similar charges, but does not include profit.
1.1.30	Accepted contract Amount	"Accepted contract Amount" means the amount accepted in the Letter of Award for the execution and completion of the
		Works including remedying of any defects.
1.1.31	Final Payment Certificate	"Final Payment Certificate" means the payment certificate issued under Sub-Clause 14.10 of GCC.
1.1.32	Goods	"Goods" means Contractor's Equipment, Materials, Plant and Temporary Works, or any of them as appropriate.
1.1.33	Contractor's Equipment	"Contractor's Equipment" means all apparatus, machinery, vehicles, and other things required for the execution and completion of the works and the remedying of any defects However, Contractor's Equipment excludes Temporary Works, Employer's Equipment (if any), Plant, Materials and any other things intended to form or forming part of the Permanent Works.
1.1.34	Site	"Site" means the places where the Permanent Works are to be executed and to which Plant and materials are to be delivered, and any other places as may be specified in the Contract as

forming part of the Site.

- 1.1.35 **Permanent Works** "Permanent Works" means the permanent works to be executed by the Contractor under the Contract.
- 1.1.36 **Temporary Works** "Temporary Works" means all temporary works of every kind (other than contractor's Equipment) required on Site for the execution and completion of the Permanent Works and the remedying of any defects.
- 1.1.37 **Applicable Law** "Applicable Law" means the laws and any other instruments having the force of law in India for the time being.
- 1.1.38 **Approval** "Approval" means consent by the employer in writing
- 1.1.39 **Drawings** "Drawing" means the drawings and plans specified in the specifications: The expression "Work" means all the works specified or set forth and required in and by the said specifications, drawing and schedule or to be implied there from or incidental thereto or to be hereafter specified or required in such explanatory instructions and drawings (being in conformity with the said original specifications, drawing and schedule) and also in such additional instructions and drawings not being in conformity as aforesaid as shall from time to time, during the progress of the work hereby contracted for, be supplied by the owner.

1.2. Interpretations

- 1.2.1 Where the contract so requires, words imparting the singular only shall also include the plural and vice versa. Any reference to masculine gender shall whenever required include feminine gender and vice versa.
- 1.2.2 Heading and marginal notes in these General Conditions shall not be deemed to form part thereof or be taken into consideration in the interpretation of construction thereof of the contract.
- 1.2.3 Wherever in the Contract provision is made for the giving or issuing of any notice, consent, approval, certificate, or determination by any person, unless otherwise specified such notice, consent, approval, certificate, or determination shall be in writing and the words "notify," "certify" or "determine" shall be construed accordingly. Any such consent, approval, certificate, or determination shall not unreasonably be withheld or delayed.

1.3. Law and Language

The Contract shall be governed by Indian Law.

The language for communications shall be English as stated in SCC.

1.4. Priority of Documents

The documents forming the Contract are to be taken as mutually explanatory of one another. For the purposes of interpretation, the priority of the documents shall be in accordance with the following sequence:

- a) The contract Agreement (if completed),
- b) The Letter of Award/Firm work order
- c) Special conditions of contract
- d) The priced bill of Quantities
- e) General conditions of contract
- f) The Drawings and Annexures.
- g) The Technical Specification
- h) Post bid correspondences and any other document forming part of contract.
- i) If any ambiguity or discrepancy is found in the documents, the EMPLOYER

ENGINEER shall be the sole authority to issue any necessary clarification or instruction in this regard.

1.5. Contract Agreement

In terms of ITB 6.9, Section II of the Tender Document pertaining to JV / Consortium, the registration under Companies Act 2013 is mandatory after award of work and before signing of Agreement. Hence, the formality of incorporation of company under Registrar of Companies (as per the mission of Letter of Intent for JV / Consortium) including submission of Performance Bank Guarantee, PAN, GST registration & any other required additional document in favour of the registered company shall be submitted to the Employer within 30 days from the date of issuance of Letter of Award.

The Parties shall enter into a Contract Agreement within 45 days after the issue of the Letter of Award. The Contract Agreement shall be in the format annexed to the bid. The costs of stamp duties and similar charges (if any) imposed by law in connection with entry into the Contract Agreement shall be borne by the Bidder.

1.6. Assignment/sub-contracting

The Contractor shall not assign the whole or any part of the work to any other Party/Firm/Individual without prior written consent of the Employer.

1.7. Employer's Use of Contractor's Documents

- a) The Specification and Drawings shall be in the custody and care of the Employer. Unless otherwise stated in the Contract, four copies of the Contract and of each subsequent Drawing shall be supplied to the Contractor, who may make or request further copies at the cost of the Contractor. Each of the Contractor's Documents shall be in the custody and care of the Contractor, unless and until taken over by the Employer. Unless otherwise stated in the Contract, the Contractor shall supply to the EMPLOYER/ENGINEER six copies of each of the Contractor's Documents.
- b) The Contractor shall keep on the Site, a copy of the Contract, publications named in the Specification, the Contractor's Documents (if any), the Drawings and Variations and other communications given under the Contract. The Employer's Personnel shall have the right of access to all these documents at all reasonable times. If a Party becomes aware of an error or defect of a technical nature in a document, which was prepared for use in executing the Works, the Party shall promptly give notice to the other Party of such error or defect.
- c) As between the Parties, the Contractor shall retain the copyright and other intellectual property rights in the Contractor's Documents and other project documents made by/or on behalf of the Contractor. The Contractor shall be deemed (by signing the Contract) to give to the Employer a non-terminable transferable non-exclusive royalty-free License to copy, use and Communicate the Contractor's Documents, including making and using modifications of them for the same work or other works under IWAI.

This License shall:

- 1) Apply throughout the actual or intended working life (whichever is longer) of the relevant parts of the Works,
- 2) Entitle any person in authorized possession of the relevant part of the Works to copy, use and communicate the Contractor's Documents for the purposes of completing, operating, maintaining, altering, adjusting, repairing and demolishing the Works,
- 3) In the case of Contractor's Documents, which are in the form of computer programs and other software, permit their use on any computer on the Site and other places as envisaged by the Contract, including replacements of any computers supplied by the Contractor.

The Contractor's Documents and other documents made by (or on behalf of) the Contractor shall not, without the Contractor's consent, be used, copied or

communicated to a third party by (or on behalf of) the Employer for purposes other than those permitted under this Sub-Clause.

1.8. Contractor's use of Employer's Documents

As between the Parties, the Employer shall retain the copyright and other intellectual property rights in the Specification, the Drawings and other documents made by (or on behalf of) the Employer. The Contractor may, at his cost, copy, use, and obtain communication of these documents for the purposes of the Contract. They shall not, without the Employer's consent, be copied, used or communicated to a third party by the Contractor, except as necessary for the purposes of the Contract.

1.9. Disclosure of Confidential Details

The Contractor shall disclose all such confidential and other information as EMPLOYER may reasonably require in order to verifying the Contractor's compliance With the Contract.

1.10. Compliance with Laws

The Contractor shall, in performing the Contract, comply with applicable Laws. Unless otherwise:

- a) the Employer shall have obtained (or shall obtain) the planning, zoning or similar permission for the Permanent works, and any other permissions described in the Specification as having been (or being) obtained by the Employer; and the Employer shall indemnify and hold the Contractor harmless against and from the consequences of any failure to do so; and
- b) the Contractor shall give all notices, pay all taxes, duties and fees, and obtain all permits, licenses and approvals for his equipment and manpower as required by the Laws in relation to the execution and completion of the Works and the remedying of any defects; and the Contractor shall indemnify and hold the Employer harmless against and from the consequences of any failure to do so.

1.11. Joint and Several Liability

If the Contractor constitutes (under applicable Laws) a joint venture, consortium or other unincorporated grouping of two or more persons/ companies

- a) each such persons/ companies shall be deemed to be jointly and severally liable to the Employer for the performance of the Contract;
- b) each such persons/ companies shall notify the Employer of their leader who shall have authority to bind the Contractor and each of these persons/

companies shall provide a parent company guarantee as a part of bid submission.

c) The Contractor shall not alter its composition or legal status without the prior consent of the Employer.

1.12. Contract Details to be Confidential

The Contractor shall treat the details of the contract as private and confidential, except to the extent necessary to carry out obligations under it or to comply with applicable laws. The Contractor shall not publish, permit to be published or disclose any particulars of the works in any trade or technical paper of elsewhere without written approval of the Employer.

2. Employer

2.1. Employer's responsibility

The Employer shall ensure that the work carried out by the Contractor on the Site with:

- (a) full regard to the safety of all persons entitled to be upon the Site, and
- (b) keeping the Site in an orderly state appropriate to the avoidance of danger to such persons.

If under Clause 4.29 and 4.30 the Employer shall employ other contractors on the Site, he shall require them to have the same regard for safety and avoidance of danger.

2.2. Employer's Risks

The Employer's risks are, in so far as they directly affect the execution of the Works in the country where the permanent works are to be executed:

- (a) war, hostilities (whether war be declared or not), invasion, act of foreign enemies,
- (b) rebellion, revolution, insurrection, or military or usurped power, or civil war,
- (c) ionizing radiations, or contamination by radioactivity from any nuclear fuel, or from any nuclear waste from the combustion of nuclear fuel, radio-active toxic explosive or other hazardous properties of any explosive nuclear assembly or nuclear component thereof,
- (d) pressure waves caused by aircraft or other aerial devices travelling at sonic or supersonic speeds,

- (e) riot, commotion or disorder, unless solely restricted to employees of the Contractor or of his Subcontractors and arising from the conduct of the Works,
- (f) loss or damage due to the use or occupation by the Employer of any Section or part of the Permanent Works, except as may be provided for in the Contract,
- (g) loss or damage to the extent that it is due to the design of the Works, and
- (h) any operation of the forces of nature (insofar as it occurs on the Site) that an experienced contractor:
 - (i) could not have reasonably foreseen, or
 - (ii) could reasonably have foreseen, but against which he could not reasonably have taken at least one of the following measures:
- (a) prevent loss or damage to physical property from occurring by taking appropriate measures, or
- (b) insure against such loss or damage.

2.3. Right of Access to the Site

The Employer shall give the Contractor right of access to and possession of all parts of the Site within the time (or times) stated in the SCC. The right and possession may not be exclusive to the Contractor. If, under the Contract, the Employer is required to give (to the Contractor) possession of any foundation, structure, plant or means of access, the Employer shall do so in the time and manner stated in the Specification.

However, the Employer may withhold any such right or possession until the Performance Security has been received

If no such time is stated in the SCC, the Employer shall give the Contractor right of access to, and possession of, the Site within such times as may be required to enable the Contractor to proceed in accordance with the programme submitted. The same shall be handed over to the employer

If the Contractor suffers delay and/or incurs Cost as a result of a failure by the Employer to give any such right or possession within such time, the Contractor shall give notice to the EMPLOYER/ ENGINEER and shall be entitled to

- a. an extension of time for any such delay, if completion is or will be delayed
- b. Payment of any such Cost-plus reasonable profit, which shall be included in the Contract Price.

After receiving this notice, the EMPLOYER/ENGINEER shall proceed to agree or determine these matters.

However, if and to the extent that the Employer's failure was caused by any error or delay by the Contractor, including an error in, or delay in the submission of, any of the Contractor's Documents, the Contractor shall not be entitled to such extension of time Cost or profit.

2.4. Permits, Licenses or Approvals

The Employer shall (where he is in a position to do so) provide reasonable assistance to facilitate the Contractor at the request of the Contractor but without holding any liability to do so towards the contractor:

- a. by obtaining copies of the Laws of the Country which are relevant to the Contract but are not readily available, and
- b. for the Contractor's applications for any permits, Licenses or approvals required by the Laws of the Country:
- c. for the delivery of Goods, including clearance through customs, and
- d. for the export of Contractors Equipment when it is removed from the Site.

All cost related to obtaining of such permits, licenses and approvals shall be borne by the contractor.

2.5. Employer's Claims

If the Employer considers himself to be entitled to any payment under any Clause of these Conditions or otherwise in connection with the Contract, the Employer or the Employer shall give notice and particulars to the Contractor. However, notice is not required for payments due. The notice shall be given as soon as practicable after the Employer became aware of the event or circumstances giving rise to the claim.

The particulars shall specify the Clause or other basis of the claim and shall include substantiation of the amount and/or extension to which the Employer considers himself to be entitled in connection with the Contract. Employer shall then proceed in accordance with the amount (if any) which the Employer is entitled to be paid by the Contractor. This amount may be included as a deduction in the Contract Price and Payment Certificates. The Employer shall only be entitled to set off against or make any deduction from an amount certified in a Payment Certificate, or to otherwise claim against the Contractor.

3. Engineer / Engineer-in-charge / EIC

3.1. Engineer's Duties and Authority

The Employer shall appoint the Engineer who shall carry out duties assigned to him in the Contract. The Engineer's staff shall include suitably qualified engineers and other professionals who are competent to carry out his duties. The Engineer shall have no authority to amend the Contract. The Engineer shall have the following roles and responsibilities.

- 3.1.1. The Engineer shall carry out the duties specified in the Contract.
- 3.1.2. The Engineer may exercise the authority specified in or necessarily to be implied from the Contract. However, he should obtain the specific approval of the Employer before exercising any such authority; particulars of such requirements are as follows: -
- 3.1.2.1. consenting to the sub-contracting of any part of the Works under Clause 5;
- 3.1.2.2. certifying additional cost determined under Clause 4.7;
- 3.1.2.3. determining an extension of time under sub clause 8.6, 8.7 and 8.8 of Clause 8;
- 3.1.2.4. issuing a variation under sub clause 12.1 and 12.2 of clause 12, except:
 - (i) in an emergency situation, as reasonably determined by the Engineer; or
 - (ii) if such variation would increase or decrease the quantity of allied items by less than the percentage permitted as per SCC or;
- 3.1.2.5. Fixing rates or prices under sub clause 12.3, 12.4 and 12.5 of clause 12.

Provided further that any requisite approval shall be deemed to have been given by the Employer for any such authority exercised by the Engineer.

3.1.2.6. Except as expressly stated in the Contract, the Engineer shall have no authority to relieve the Contractor of any of his obligations under the Contract.

3.2. Engineer's Representative

The Engineer's Representative shall be appointed by and be responsible to the Engineer and shall carry out such duties and exercise such authority as may be delegated to him by the Engineer under Sub-Clause 3.3.

3.3. Engineer's Authority to Delegate

Any communication given by the Engineer's Representative to the Contractor in accordance with such delegation shall have the same effect as though it had been given by the Engineer. Provided that:

(a) Any failure of the Engineer's Representative to disapprove any work, materials or Plant shall not prejudice the authority of the Engineer to

disapprove such work, materials, or Plant and to give instructions for the rectification thereof, and

(b) If the Contractor questions any communication of the Engineer's Representative he may refer the matter to the Engineer who shall confirm, reverse, or vary the contents of such communication.

3.4. Appointment of Assistants

The Engineer or the Engineer's Representative may appoint any number of persons to assist the Engineer's Representative in the carrying out of his duties under Sub-Clause 3.2. He shall notify to the Contractor the names, duties and scope of authority of such persons. Such assistants shall have no authority to issue any instructions to the Contractor save in so far as such instructions may be necessary to enable them to carry out their duties and to secure their acceptance of materials, Plant, or workmanship as being in accordance with the Contract, and any instructions given by any of them for those purposes shall be deemed to have been given by the Engineer's Representative.

3.5. Instructions in Writing

Instructions given by the Engineer shall be in writing, provided that if for any reason the Engineer considers it necessary to give any such instruction orally, the Contractor shall comply with such instruction. Confirmation in writing of such oral instruction given by the Engineer, whether before or after the carrying out of the instruction, shall be deemed to be an instruction within the meaning of this Sub-Clause. Provided further that if the Contractor, within 7 days, confirms in writing to the Engineer any oral instruction of the Engineer and such confirmation is not contradicted in writing within 7 days by the Engineer, it shall be deemed to be an instruction of the Engineer.

3.6. Engineer to Act Impartially

Wherever, under the Contract, the Engineer is required to exercise his discretion by:

- (a) Giving his decision, opinion or consent,
- (b) Expressing his satisfaction or approval,
- (c) Determining value, or
- (d) Otherwise taking action which may affect the rights and obligations of the Employer or the Contractor

He shall exercise such discretion impartially within the terms of the Contract and having regard to all the circumstances. Any such decision, opinion, consent,

expression of satisfaction, or approval, determination of value or action may be opened up, reviewed or revised as provided in Clause 15.3.1.

3.7. Engineer at liberty to object

The Engineer shall be at liberty to object to and require the Contractor to remove forthwith from the Works any person provided by the Contractor who, in the opinion of the Engineer, misconducts himself, or is incompetent or negligent in the proper performance of his duties, or whose presence on Site is otherwise considered by the Engineer to be undesirable, and such person shall not be again allowed upon the Works without the consent of the Engineer. Any person so removed from the Works shall be replaced as soon as possible.

4. Contractor

4.1. Contractor's General Responsibilities

The Contractor shall, with due care and diligence, execute and complete the Works and remedy any defects therein in accordance with the provisions of the Contract. The Contractor shall provide all superintendence, labour, materials, Plant, Contractor's Equipment, and all other things, whether of a temporary or permanent nature, required in and for such execution, completion, and remedying of any defects, so far as the necessity for providing the same is specified in or is reasonably to be inferred from the Contract.

The Contractor shall give prompt notice to the Engineer, with a copy to the Employer, of any error, omission, fault or other defect in the design of or Specification for the Works which he discovers when reviewing the Contract or executing the Works.

4.2. Site Operations and Methods of Construction

The Contractor shall take full responsibility for the adequacy, stability and safety of all Site operations and methods of construction. Provided that the Contractor shall not be responsible (except as stated hereunder or as may be otherwise agreed) for the design or specification of Permanent Works, or for the design or specification of any Temporary Works not prepared by the Contractor. Where the Contract expressly provides that part of the Permanent Works shall be designed by the Contractor, he shall be fully responsible for that part of such Works, notwithstanding any approval by the Engineer.

4.3. Contract Agreement

4.3.1. The Contractor shall, if called upon so to do, enter into and execute the Contract Agreement, to be prepared and completed at the cost of the Employer, in the form annexed to these Conditions with such modification as may be necessary. The

Parties shall enter into a Contract Agreement within 45 days after the date of issue of the Letter of Award.

- 4.3.2. The Contractor shall be furnished free of charge certified true copy of the contract document.
- 4.3.3. A copy of the Contract Document furnished to the Contractor as aforesaid shall be kept by the Contractor on the Site in good condition and the same shall at all reasonable times be available for inspection and use by the Engineer-in-Charge, his representatives or by other Inspecting officers of the Authority.
- 4.3.4. None of these Documents shall be used by the Contractor for any purpose other than that of this contract.

4.4. Performance Guarantee and security deposit

- 4.4.1. Performance Guarantee
- 4.4.1.1. For all Bidders including MSME Registered Firms

The contractor shall be required to deposit an amount equal to 5% of the contract value of the work as performance guarantee in the form of irrevocable bank guarantee bond of any scheduled bank or State Bank of India in accordance with the form prescribed within 21 days of the issue of the work order/LOA. The performance guarantee shall be issued by a scheduled or nationalized bank in India.

- 4.4.1.2. Without limitation to the provisions of the preceding paragraph, whenever the Engineer determines an addition to the Contract Price as a result of a change in cost and/or legislation or as a result of a variation amounting to more than 20 percent of the portion of the Contract Price, the Contractor, at the Engineer's written request, shall promptly increase the value of the performance security proportionately and furnish additional performance security, if any, arising out of provisions mentioned herein. The cost of complying with the requirements of this clause shall be borne by the contractor.
- 4.4.1.3. For Tenders having financial quote up to 5% lesser than the estimated contract value, no additional security deposit is required. But for tenders having financial quote less by more than 5% of the estimated contract value, the difference between the quoted amount and 95% of estimated contract value in the form of BG/DD shall be submitted by the bidder within 21 days of the issuance of the work order. Similarly, for tenders having financial quote more than 5% of the estimated contract value, the difference between the quoted amount and the amount more than 5% of the estimated cost, up to a maximum of 10% more than estimated contract value in the form of BG/DD shall be submitted by the bidder within 21 days of the issuance of the work order. The additional BG/DD defined

herein shall be over and above the performance security and shall be returned along with the performance security to the contractor.

4.4.1.4. The Performance Guarantee shall be valid one month beyond the work period until the Contractor has executed & completed the Works, remedied any defects therein in accordance with the Contract. No claim shall be made against such security and such security shall be returned to the Contractor within 21 days after completion of Defect Liability Period. In case the extension of time for completion with or without LD, the contractor shall get the validity of the performance guarantee extended to cover such extension of time of the work.

4.4.2. Security Deposit

- 4.4.2.1 A sum @ 10% of the gross amount of the bill shall be deducted from each running bill of the contractor till the sum along with the sum already deposited in cash/RTGS as the earnest money, will amount to security deposit of 5% of the contract value of work. The earnest money deposited in the form of Bank guarantee shall be returned to the contractor after receiving the Performance Bank Guarantee.
- 4.4.2.2 In case of MSME registered firms, deduction of a sum of 10% of the gross amount of the bill from each running bill till that amount reaches 5% of the contract value of the work as Security Deposit will also be made.

The total security deposit shall remain with IWAI till the completion of the contract or the payment of the final bill payable in accordance with agreement condition whichever is later, provided the Engineer-in-Charge is satisfied that there is no demand outstanding against the contractor.

- 4.4.2.3 No interest will be paid on security deposit.
- 4.4.2.4 If the contractor neglects to observe or fails to perform any of his obligations under the contract, it shall be lawful for the Employer/ EIC to forfeit either in whole or in part, the security deposit furnished by the contractor. However, if the contractor duly performs and completes the contract in all respects and presents in absolute "NO DEMAND CERTIFICATE" in the prescribed form, IWAI shall refund the security deposit to the contractor after deduction of cost and expenses that the Authority may have incurred and other money including all losses and damages which the Authority is entitled to recover from the Contractor.
- 4.4.2.5 In case of delay in the progress of work, the Engineer- in-Charge shall issue to the contractor a memo in writing pointing out the delay in progress and calling upon the contractor to explain the causes for the delay within 3 days of receipt of the memo and 10 days from issuance of memo whichever is earlier. If the Engineer-in-Charge is not satisfied with the explanations offered, he may forfeit the security deposit and / or withhold payment of pending bills in whole or in part and/ or get

the measures of rectification of progress of work accelerated to the pre-defined level at the risk and cost of the contractor.

- 4.4.2.6 All compensation or other sums of money payable by the contractor under the terms of the contract or any other contract or on any other account whatsoever, may be deducted from or paid by the sale of a sufficient part of his security or from any sums which may be due or may become due to the contractor by the Authority on any account whatsoever. Also, in the event of the contractor's security deposit being reduced by reasons of such deductions or sale, as aforesaid the contractor shall, within 14 days of receipt of notice of demand from the Engineer-in-Charge make good the deficit in his security deposit.
- 4.4.2.7 In case the contractor fails to commence the work or when the contractor has defaulted for more than thirty days or when any amount is to be recovered from the contractor as penalty or deduction and the contractor fails to remit such amount even after due notice is given to him in this regard, IWAI at its own discretion may encash the Bank Guarantee furnished as performance guarantee and /or recover the sum from his security deposit.

4.5. Inspection of Site

The Employer shall have made available to the Contractor, before the submission by the Contractor of the Tender, such available data on hydrological and hydrographical, results of rock testing, surface soil testing, bore hole data at the proposed terminal locations, as have been obtained by or on behalf of the Employer from investigations undertaken relevant to the Works but the Contractor shall be responsible for his own interpretation thereof.

The Contractor acknowledges that it has inspected and examined the Site and its surroundings and information available in connection therewith and to have satisfied himself (so far as is practicable, having regard to considerations of cost and time) before submitting his Tender, as to:

- a) The form and nature thereof, including the sub-surface conditions,
- b) The hydrological, hydrographical, and climatic conditions,
- c) The extent and nature of work and materials necessary for the execution and completion of the works and the remedying of any defects therein, and
- d) The means of access to the site and the accommodation he may require,
- e) Power, water, fuel, and other supplies he may require
- f) The infrastructure for transportation of men and material along the project corridor
- g) Market rates of the items in the working country

- h) Medical facilities and
- i) General safety & security of project area and
- j) The Laws, procedures, and labour practices of the Country

and, in general, shall be deemed to have obtained all necessary information, subject as above mentioned, as to risks, contingencies and all other circumstances which may influence or affect his Tender. The Contractor shall be deemed to have based his Tender on the data made available by the Employer and on his own inspection and examination, all as aforementioned.

4.6. Sufficiency of Tender

The Contractor shall be deemed to have satisfied himself as to the correctness and sufficiency of the Tender and of the rates and prices stated in the Bill of Quantities, all of which shall, except insofar as it is otherwise provided in the Contract, cover all his obligations under the Contract (including those in respect of the supply of goods, materials, Plant or services or of contingencies) and all matters and things necessary for the proper execution and completion of the Works and the remedying of any defects therein.

4.7. Unforeseeable Physical Obstructions or Conditions

If, however, during the execution of the Works the Contractor encounters physical obstructions or physical conditions, other than climatic, sea, river conditions on the Site, such obstructions or conditions were, in his opinion, not foreseeable by an experienced contractor, the Contractor shall forthwith give notice thereof to the Engineer, with a copy to the Employer. On receipt of such notice, the Engineer shall, if in his opinion such obstructions or conditions could not have been reasonably foreseen by an experienced contractor, after due consultation with the Employer and the Contractor, determine

- a. Any extension of time to which the Contractor is entitled under sub clause 8.6, 8.7 and 8.8 of clause 8, and
- b. The amount of any costs which may have been incurred by the Contractor by reason of such obstructions or conditions having been encountered, which shall be added to the Contract Price, and shall notify the Contractor accordingly, with a copy to the Employer. Such determination shall take account of any instruction which the Engineer may issue to the Contractor in connection therewith, and any proper and reasonable measures acceptable to the Engineer which the Contractor may take in the absence of specific instructions from the Engineer.

4.8. Work to be in Accordance with Contract

Unless it is legally or physically impossible, the Contractor shall execute and complete the Works and remedy any defects therein in strict accordance with the Contract to the satisfaction of the Engineer. The Contractor shall comply with and adhere strictly to the Engineer's instructions on any matter, whether mentioned in the Contract or not, touching or concerning the Works. The Contractor shall take instructions only from the Engineer (or his Representatives or Assistants)

4.9. Contractor's Programme

4.9.1. The Contractor shall, within the time stated in the SCC, after the date of issue of the Letter of Award, submit to the Engineer for his consent a programme, in such form and detail as the Engineer shall reasonably prescribe, for the execution of the Works. The Contractor shall, whenever required by the Engineer, also provide in writing for his information a general description of the arrangements and methods which the Contractor proposes to adopt for the execution of the Works.

4.9.2. Revised Programme

If at any time it should appear to the Engineer that the actual progress of the Works does not conform to the programme to which consent has been given under Sub- Clause 4.9.1, the Contractor shall produce, at the request of the Engineer, a revised programme showing the modifications to such programme necessary to ensure completion of the Works within the Time for Completion.

4.9.3. Cash Flow Estimate to be submitted

The Contractor shall, within the time stated in the SCC, after the signing of Agreement, provide to the Engineer for his information a detailed 3 months rolling cash flow, of all payments to which the Contractor will be entitled under the Contract and the Contractor shall subsequently supply revised cash flow estimates at quarterly intervals, if required to do so by the Engineer.

4.9.4. Contractor not relieved of Duties or Responsibilities

The submission to and consent by the Engineer of such programmes or the provision of such general descriptions or cash flow estimates shall not relieve the Contractor of any of his duties or responsibilities under the Contract.

4.10. Contractor's Superintendence

The Contractor shall provide all necessary superintendence during the execution of the Works and as long thereafter as the Engineer may consider necessary for the proper fulfilling of the Contractor's obligations under the Contract. The Contractor, or a competent and authorized representative approved of by the Engineer, which approval may at any time be withdrawn, shall give his whole time to the superintendence of the Works. Such authorized representative shall receive, on behalf of the Contractor, instructions from the Engineer.

If approval of the representative is withdrawn by the Engineer, the Contractor shall, as soon as is practicable, having regard to the requirement of replacing him as hereinafter mentioned, after receiving notice of such withdrawal, remove the representative from the Works and shall not thereafter employ him again on the Works in any capacity and shall replace him by another representative approved by the Engineer.

4.11. Contractor's Employees

The Contractor shall provide on the Site in connection with the execution and completion of the Works and the remedying of any defects therein

- (a) only such technical personnel as are skilled and experienced in their respective callings and such foremen and leading hands as are competent to give proper superintendence of the Works, and
- (b) such skilled, semi-skilled and unskilled labour as is necessary for the proper and timely fulfilling of the Contractor's obligations under the Contract.

4.12. Setting out

The Contractor shall be responsible for:

- (a) the accurate setting-out of the Works in relation to original points, lines and levels of reference given by the Engineer in writing,
- (b) the correctness, subject as above mentioned, of the position, levels, dimensions and alignment of all parts of the Works, and
- (c) the provision of all necessary instruments, appliances and labour in connection with the foregoing responsibilities.

If, at any time during the execution of the Works, any error appears in the position, levels, dimensions, or alignment of any part of the Works, the Contractor, on being required so to do by the Engineer, shall, at his own cost, rectify such error to the satisfaction of the Engineer.

The checking of any setting-out or of any line or level by the Engineer shall not in any way relieve the Contractor of his responsibility for the accuracy thereof and the Contractor shall carefully protect and preserve all benchmarks, sight-rails, pegs and other things used in setting-out the Works.

4.13. Safety and Security

The Contractor shall, throughout the execution and completion of the Works and the remedying of any defects therein:

(a) have full regard for the safety of all persons entitled to be upon the Site and keep the Site (so far as the same is under his control) and the Works

(so far as the same are not completed or occupied by the Employer) in an orderly state appropriate to the avoidance of danger to such persons,

(b) provide and maintain at his own cost all lights, guards, fencing, warning signs and watching, when and where necessary or required by the Engineer or by any duly constituted authority, for the protection of the Works or for the safety and convenience of the public or others, and

4.14. Electricity, Water and Gas

The Contractor shall be responsible for the provision of all power, water and other services he may require for his execution of the Works.

4.15. Protection of the Environment

The Contractor shall take all reasonable steps to protect the environment (both on and off the Site) and to limit damage and nuisance to people and property resulting from pollution, noise and other results of his operations.

The Contractor shall ensure that emissions, surface discharges and effluent from the Contractor's activities shall not exceed the permissible / acceptable values and shall not exceed the values prescribed by applicable Laws. The Contractor is required to establish and operate with suitably qualified personnel a system for monitoring the environmental factors that may be directly impacted by the Works and make available such records for inspection of Engineer or duly authorized representatives of Government of the country in which Works are executed as and when so required.

4.16. Care of Works

The Contractor shall take full responsibility for the care of the Works and materials and Plant for incorporation therein from the Commencement Date until the date of issue of the Completion certificate for the whole of the Works, when the responsibility for the said care shall pass to the Employer. Provided that:

- (a) if the Engineer issues a Completion certificate for any Section or part of the Permanent Works, the Contractor shall cease to be liable for the care of that Section or part from the date of issue of the Completion certificate, when the responsibility for the care of that Section or part shall pass to the Employer, and
- (b) The Contractor shall take full responsibility for the care of any outstanding Works and materials and Plant for incorporation therein which he undertakes to finish during the maintenance period until such outstanding Works have been completed pursuant to Clause 10.

4.17. Responsibility to Rectify Loss or Damage

If any loss or damage happens to the Works, or any part thereof, or materials or Plant for incorporation therein, during the period for which the Contractor is responsible for the care thereof, from any cause whatsoever, other than the risks defined in Sub- Clause 2.2, the Contractor shall, at his own cost, rectify such loss or damage so that the Permanent Works conform in every respect with the provisions of the Contract to the satisfaction of the Engineer. The Contractor shall also be liable for any loss or damage to the Works occasioned by him in the course of any operations carried out by him for the purpose of complying with his obligations under Clauses 11.

4.18. Loss or Damage Due to Employer's Risks

In the event of any such loss or damage happening from any of the risks defined in Sub-Clause 2.2, or in combination with other risks, the Contractor shall, if and to the extent required by the Engineer, rectify the loss or damage and the Engineer shall determine an addition to the Contract Price in accordance with Clause 12.3, 12.4 and 12.5 and shall notify the Contractor accordingly, with a copy to the Employer. In the case of a combination of risks causing loss or damage any such determination shall take into account the proportional responsibility of the Contractor and the Employer.

4.19. Compliance with Statutes, Regulations

The Contractor shall conform in all respects, including by the giving of all notices and the paying of all fees, with the provisions of:

- (a) any National or State Statute, Ordinance, or other Law, or any regulation, or bye-law of any local or other duly constituted authority in relation to the execution and completion of the Works and the remedying of any defects therein, and
- (b) the rules and regulations of all public bodies and companies whose property or rights are affected or may be affected in any way by the Works, and the Contractor shall keep the Employer indemnified against all penalties and liability of every kind for breach of any such provisions. Provided always that the Employer shall be responsible for obtaining any planning, zoning or other similar permission required for the Works to proceed and shall indemnify the Contractor in accordance with Sub-Clause 20.7.

4.20. Fossils

The Contractor shall take reasonable precautions to prevent his workmen or any other persons from removing or damaging all fossils, coins, articles of value or antiquity and structures and other remains or things of geological or archaeological interest discovered on the Site. The Contractor immediately upon discovery thereof and before removal, acquaint the Engineer of such discovery

and carry out the Engineer's instructions for dealing with the same. If, by reason of such instructions, the Contractor suffers delay and/or incurs costs then the Engineer shall, after due consultation with the Employer and the Contractor, determine:

- (a) any extension of time to which the Contractor is entitled under sub clause 8.6, 8.7 and 8.8 of clause 8, and
- (b) the amount of such costs, which shall be added to the Contract Price, and shall notify the Contractor accordingly, with a copy to the Employer.

4.21. Patent Rights

The Contractor shall save harmless and indemnify the Employer from and against all claims and proceedings for or on account of infringement of any patent rights, design trademark or name or other protected rights in respect of any Contractor's Equipment, materials or Plant used for or in connection with or for incorporation in the Works and from and against all damages, costs, charges and expenses whatsoever in respect thereof or in relation thereto, except where such infringement results from compliance with the design or Specification provided by the Engineer.

4.22. Copy Right

The copyright in all drawings, documents and other materials containing data and information furnished to the Employer by the Contractor herein shall remain vested in the Contractor or, if they are furnished to the Employer directly or through the Contractor by any third party, including suppliers of materials, the copyright in such materials shall remain vested in such third-party indemnity of the employer.

4.23. Royalties

Except where otherwise stated, the Contractor shall pay all tonnage and other royalties, rent and other payments or compensation, if any, for getting stone, sand, gravel, clay or other materials required for the Works as per governing laws.

4.24. Interference with Traffic and Adjoining Properties

All operations necessary for the execution and completion of the Works and the remedying of any defects therein shall, so far as compliance with the requirements of the Contract permits, be carried on so as not to interfere unnecessarily or improperly with:

(a) the convenience of the public, or

(b) the access to, use and occupation of public or private roads and footpaths to or of properties whether in the possession of the Employer or of any other person.

The Contractor shall save harmless and indemnify the Employer in respect of all claims, proceedings, damages, costs, charges and expenses whatsoever arising out of, or in relation to, any such matters insofar as the Contractor is responsible therefor.

4.25. Avoidance of Damage to Roads and navigation aids, temporary structure/permanent structure.

The Contractor shall use every reasonable means to prevent any of the roads or bridge jetties, wharfs or river channels, temporary berths for inland vessels, communicating with or on the routes to the Site from being damaged or injured by any traffic of the Contractor or any of his Subcontractors and, in particular, shall select routes, choose and use vehicles and restrict and distribute loads so that any such extraordinary traffic as will inevitably arise from the moving of materials, Plant, Contractor's Equipment or Temporary Works from and to the Site shall be limited, as far as reasonably possible, and so that no unnecessary damage or injury may be occasioned to such roads and bridges or other structures. However, if any claim arises regarding damage, maintenance/repairs/reconstruction/replacement would be Contractor's liability.

4.26. Transport of Contractor's Equipment or Temporary Works

Save insofar as the Contract otherwise provides, the Contractor shall be responsible for and shall pay the cost of strengthening any bridges, jetties, wharfs or altering or improving any road or river channels, temporary berths for inland vessels, communicating with or on the routes to the Site to facilitate the movement of Contractor's Equipment or Temporary Works and the Contractor shall indemnify and keep indemnified the Employer against all claims for damage to any such road or bridge or other structures caused by such movement, including such claims as may be made directly against the Employer, and shall negotiate and pay all claims arising solely out of such damage, the costs of which are deemed to be included in the contract price.

The Contractor shall be responsible for packing, loading, transporting, receiving, unloading, storing and protecting all Goods and other things required for the Works.

4.27. Transport of Materials or Plant

If, notwithstanding Sub-Clause 4.25, any damage occurs to any bridge or road or structure jetties, wharfs or river channels, temporary berths for communicating with or on the routes to the Site arising from the transport of materials or Plant, the

Contractor shall notify the Engineer with a copy to the Employer, as soon as he becomes aware of such damage or as soon as he receives any claim from the authority entitled to make such claim. In such cases the Contractor shall negotiate the settlement of and pay all sums due in respect of such claim and shall indemnify the Employer in respect thereof and in respect of all claims, proceedings, damages, costs, charges and expenses in relation thereto. Provided that if and so far as any such claim or part thereof is, in the opinion of the Engineer, due to any failure on the part of the Contractor to observe and perform his obligations under Sub-Clause 4.25, and the Contractor fails to take adequate steps to settle any claims arising, the Engineer will be at liberty to negotiate with such claimant and pay for damages which shall be recoverable from the Contractor by the Employer and may be deducted by the Employer from any monies due or to become due to the Contractor and the Engineer shall notify the Contractor accordingly, with a copy to the Employer. The Contractor shall give the Employer not less than 21 days' notice of the date on which any Equipment, Plant or a major item of other Goods will be delivered to the Site;

4.28. Waterborne Traffic

Where the nature of the Works is such as to require the use by the Contractor of waterborne transport, the foregoing provisions of this Clause shall be construed as though "road" included a lock, dock, sea wall or other structure related to a waterway and "vehicle" included craft and shall have effect accordingly.

4.29. Opportunities for Other Contractors

The Contractor shall, in accordance with the requirements of the Engineer, afford all reasonable opportunities for carrying out their work to:

- (a) any other contractors employed by the Employer and their workmen
- (b) the workmen of the Employer, and
- (c) the workmen of any duly constituted authorities who may be employed in the execution on or near the Site of any work not included in the Contract or of any contract which the Employer may enter into in connection with or ancillary to the Works.

4.30. Facilities for Other Contractors

If, however, pursuant to Sub-Clause 4.29 the Contractor shall, on the written request of the Engineer:

(a) make available to any such other contractor, or to the Employer or any such authority, any roads, or ways for the maintenance of which the Contractor is responsible,

- (b) permit the use, by any such, of Temporary Works or Contractor's Equipment on the Site, or
- (c) provide any other service of whatsoever nature for any such, the Engineer shall determine an addition to the Contract Price in accordance with Clauses 12.3,12.4 and 12.5 and shall notify the Contractor accordingly, with a copy to the Employer.

4.31. Contractor to Keep Site Clear

During the execution of the Works the Contractor shall keep the Site reasonably free from all unnecessary obstruction and shall store or dispose off any Contractor's Equipment and surplus materials and clear away and remove from the Site any wreckage, rubbish or Temporary Works no longer required.

4.32. Clearance of Site on Completion

Prior to the issue of any Completion certificate the Contractor shall clear away and remove from that part of the Site to which such Completion certificate relates all Contractor's Equipment, surplus material, rubbish, and Temporary Works of every kind, and leave such part of the Site and Works clean and in a workmanlike condition to the satisfaction of the Engineer. Provided that the Contractor shall be entitled to retain on Site, until the end of the maintenance period, such materials, Contractor's Equipment and Temporary Works as are required by him for the purpose of fulfilling his obligations during the maintenance period.

4.33. Language Ability of Superintending Staff

A reasonable proportion of the Contractor's superintending staff shall have a working knowledge of the local language of the Site and English language, or the Contractor shall have available on site at all times a sufficient number of competent interpreters to ensure the proper transmission of instructions and information.

4.34. Employment of Local Personnel

The Contractor is encouraged, to the extent practicable and reasonable, to employ staff and labour with appropriate qualifications and experience from sources within the country of the Employer and the country/place where the Work is executed.

4.35. Boreholes and Exploratory Excavation

In respect of Works which the Contractor is required to execute, install, and construct to the requirements of Employer, any boreholes or exploratory excavation that may be required to be undertaken before commencement or during execution of Works will have to be carried out by the Contractor at no extra cost.

4.36. Contractor's Equipment, Temporary Works, and Materials

4.36.1. Contractor's Equipment, Temporary Works, and Materials; Exclusive Use for the Works

All Contractor's Equipment, Temporary Works and materials provided by the Contractor shall, when brought on to the Site, be deemed to be exclusively intended for the execution of the Works and the Contractor shall not remove the same or any part thereof, except for the purpose of moving it from one part of the Site to another, without the consent of the Engineer. Provided that consent shall not be required for vehicles, floating crafts engaged in transportation of staff, labour, Contractor's Equipment, Temporary Works, Plant, or materials to or from the Site.

4.36.2. Employer not Liable for Damage

The Employer shall not at any time be liable, as mentioned in Clauses 2.2, 4.16,4.17,4.18 and 13, for the loss of or damage to any of the said Contractor's Equipment, Temporary Works, or materials.

4.36.3. Customs Clearance

The Employer will use his best endeavours in assisting the Contractor, where required, in obtaining clearance through the Customs of Contractor's Equipment, materials and other things required for the Works. The provisions contained in the Framework Agreement shall apply.

4.36.4. Re-export of Contractor's Equipment

In respect of any Contractor's Equipment which the Contractor has imported for the purposes of the Works, the Employer will use his best endeavours to assist the Contractor, where required, in procuring any necessary Government consent to the re- export of such Contractor's Equipment by the Contractor upon the removal thereof pursuant to the terms of the Contract and in accordance with the provisions contained in the Framework agreement.

4.36.5. Conditions of Hire of Contractor's Equipment

With a view to securing, in the event of termination under Clause 16, the continued availability, for the purpose of executing the Works, of any hired Contractor's Equipment, the Contractor shall not remove or withdraw any of such equipment, materials, temporary works or plant from the site without obtaining written permission to do so issued by the Engineer. Employer shall be entitled to permit the use thereof by any other contractor employed by him for the purpose of executing and completing the Works and remedying any defects therein, under the terms of the said Clause 16.

4.36.6. Costs for the Purpose of Clause 16

In the event of the Employer invoking provisions of sub clause 4.36.5, for the purpose of executing and completing the Works and the remedying of any defects therein for the purpose of clause 16, the costs thereof will be determined by the Engineer in consultation with the employer and same shall form the cost of executing and completing same as incomplete obligation and liability of the Contractor and notice to that effect will be issued to the Contractor by the Engineer.

4.36.7. Incorporation of Clause in Subcontracts

The Contractor shall, where entering into any subcontract for the execution of any part of the Works, incorporate in such subcontract (by reference or otherwise) the provisions of sub clauses 4.36.5 and 4.36.6 in relation to Contractor's Equipment, Temporary Works or materials brought on to the Site by the Subcontractor and employer shall be kept indemnified against any claim arising out of such subcontractor.

4.36.8. Co-operation

The Contractor shall, as specified in the Contract or as instructed by the Engineer, allow appropriate opportunities for carrying out work to:

- (a) the Employer's Personnel,
- (b) any other contractors employed by the Employer, and
- (c) the personnel of any legally constituted public authorities,

who may be employed in the execution on or near the Site of any work not included in the Contract. Any such instruction shall constitute a Variation if and to the extent that it causes the Contractor to incur Unforeseeable Cost. Services for these personnel and other contractors may include the use of Contractor's Equipment, Temporary Works or access arrangements, which are the responsibility of the Contractor.

If, under the Contract, the Employer is required to give to the Contractor possession of any foundation, structure, plant or means of access in accordance with Contractor's Documents, the Contractor shall submit such documents to the EMPLOYER/ENGINEER in the time and manner stated in the Specification.

5. Assignment and Subcontracting

5.1. Assignment of Contract

The Contractor shall not, without the prior consent of the Employer (which consent shall be at the sole discretion of the Employer), assign the Contract or any part thereof, or any benefit or interest therein or thereunder.

5.2. Subcontracting

The Contractor shall not subcontract the whole of the Works. Except where otherwise provided by the Contract, the Contractor shall not subcontract any part of the Works without the prior consent of the Employer conveyed through the Engineer. A subcontractor under the Contract must be a firm or person or entity that satisfies the relevant provisions of the Framework Agreement. Any such consent shall not relieve the Contractor from any liability or obligation under the Contract and he shall be responsible for the acts, defaults and neglects of any Subcontractor, his agents, servants or workmen as fully as if they were the acts, defaults or neglects of the Contractor, his agents, servants or workmen. The work sub-contracted shall not be more than 15% of the contract value. If the Engineer gives his consent to the contractor to subcontract part of the work and the value of such work is more than 10% of the contract amount, the sub-contractor shall have to submit all relevant documents, as stipulated in ITB of this tender document which in turn shall be evaluated to check the capacity/capability of the proposed sub-contractor to execute the work assigned to him. Failing to do so, the Engineer may, at his discretion, reject contractor's claim to subcontract the work to such sub-contractor.

Provided that the Contractor shall not be required to obtain such consent for:

- (a) The provision of labour,
- (b) The purchase of materials which are in accordance with the standards specified in the Contract, or
- (c) The subcontracting of any part of the Works for which the Subcontractor is named in the original Contract.

In all other cases, the Contractor shall give the Employer not less than 14 days" notice of;

- (a) The intended appointment of the Subcontractor, with detailed particulars which shall include his relevant experience,
- (b) The intended commencement of the Subcontractor's work, and
- (c) The intended commencement of the Subcontractor's work on the site.

5.3. Assignment of Subcontractors' Obligations

In the event of a Subcontractor having undertaken towards the Contractor in respect of the work executed, or the goods, materials, Plant, or services supplied by such Subcontractor, any continuing obligation extending for a period exceeding that of the maintenance period under the Contract, the Contractor shall at any time, after the expiration of such Period, assign to the Employer, at the Employer's request and cost, the benefit of such obligation for the unexpired duration thereof.

5.4. Nominated Subcontractors

5.4.1. Definition of "Nominated Subcontractors"

All specialists, merchants, tradesmen and others executing any work or supplying any goods, materials, Plant or services included in the Contract, who may have been or be nominated or selected or approved by the Employer or the Engineer, for the execution of such work or the supply of such goods, materials, Plant or services, be deemed to be subcontractors to the Contractor and are referred to in this Contract as "nominated Subcontractors".

5.4.2. Nominated Subcontractors; Objection to Nomination

The Contractor shall not be required by the Employer or the Engineer, or be deemed to be under any obligation, to employ any nominated Subcontractor against whom the Contractor may raise reasonable objection or who declines to enter into a subcontract with the Contractor containing provisions:

- (a) that in respect of the work, goods, materials, Plant or services the subject of the subcontract, the nominated Subcontractor will undertake towards the Contractor such obligations and liabilities as will enable the Contractor to discharge his own obligations and liabilities towards the Employer under the terms of the Contract and will save harmless and indemnify the Contractor from and against the same and from all claims, proceedings, damages, costs, charges and expenses whatsoever arising out of or in connection therewith, or arising out of or in connection with any failure to perform such obligations or to fulfil such liabilities, and
- (b) that the nominated Subcontractor will keep harmless and indemnify the Contractor from and against any negligence by the nominated Subcontractor, his agents, workmen and servants and from and against any misuse by him or them of any Temporary Works provided by the Contractor for the purposes of the Contract and from all claims as aforesaid.

5.4.3. Design Requirements to be Expressly Stated

If in connection with any services to be provided in relation to execution of Works include any matter of design or specification of any part of the Permanent Works or of any Plant to be incorporated therein, such requirement shall be expressly stated in the

Contract and shall be included in any nominated Subcontract. The nominated Subcontract shall specify that the nominated Subcontractor providing such services will keep harmless and indemnify the Contractor from and against the same and from all claims, proceedings, damages, costs, charges, and expenses whatsoever arising out of or in connection with any failure to perform such obligations or to fulfil such liabilities.

5.4.4. Payments to Nominated Subcontractors

For all work executed or goods, materials, Plant, or services supplied by any nominated Subcontractor, the Contractor shall be entitled to:

- (a) the actual price paid or due to be paid by the Contractor, on the instructions of the Engineer, and in accordance with the subcontract;
- (b) in respect of labour supplied by the Contractor, the sum, if any, entered in the Bill of Quantities or, if instructed by the Engineer pursuant to paragraph (a) of Sub-Clause 15.2.2, as may be determined in accordance with Clause 12.3,12.4 and 12.5; and
- (c) in respect of all other charges and profit, a sum being a percentage rate of the actual price paid or due to be paid calculated, where provision has been made in the Bill of Quantities for a rate to be set against the relevant Provisional Sum, at the rate inserted by the Contractor against that item or, where no such provision has been made, at the rate inserted by the Contractor in the SCC and repeated where provision for such is made in a special item provided in the Bill of Quantities for such purpose.

5.4.5. Certification of Payments to Nominated Subcontractors

Before issuing, under Clause 14, any certificate, which includes any payment in respect of work done or goods, materials, Plant, or services supplied by any nominated Subcontractor, the Engineer shall be entitled to demand from the Contractor reasonable proof that all payments, less retentions, included in previous certificates in respect of the work or goods, materials, Plant or services of such nominated Subcontractor have been paid or discharged by the Contractor. If the Contractor fails to supply such proof, then, unless the Contractor:

- a) satisfies the Engineer in writing that he has reasonable cause for withholding or refusing to make such payments, and
- b) produces to the Engineer reasonable proof that he has so informed such nominated Subcontractor in writing,

The Employer shall be entitled to pay to such nominated Subcontractor direct, upon the certificate of the Engineer, all payments, less retentions, provided for in the nominated Subcontract, which the Contractor has failed to make to such nominated Subcontractor and to deduct by way of set-off the amount so paid by the Employer from any sums due or to become due from the Employer to the Contractor.

Provided that, where the Engineer has certified and the Employer has paid direct as aforesaid, the Engineer shall, in issuing any further certificate in favour of the Contractor, deduct from the amount thereof the amount so paid, direct as aforesaid, but shall not withhold or delay the issue of the certificate itself when due to be issued under the terms of the Contract.

6. Staff & Labour

6.1. Engagement of Staff and Labour

The Contractor shall, unless otherwise provided in the Contract, make his own arrangements for the engagement of all staff and labour, local or other, and for their payment, housing, feeding and transport. In engagement of labour, the provisions in that respect contained in the Framework Agreement shall be fully complied with by the Contractor.

6.2. Foreign staff, labour and Repatriation.

The Contractor may import to the country where Works are executed any personnel who are necessary for the execution of Works. The Contractor must ensure that these personnel are provided with the required residence visas and work permits. The Contractor shall be responsible for the return to the place where they were recruited or to their domicile of all such persons as he recruited and employed for the purposes of or in connection with the contract and he shall maintain such persons as are to be so returned in a suitable manner until they shall have left the site.

6.3. Housing for Labour

Save insofar as the contract otherwise provides, the Contractor shall provide and maintain such accommodation and amenities as he may consider necessary for all his staff and labour, employed for the purposes of or in connection with the Contract, including all fencing, water supply (both for drinking and other purposes), electricity supply, sanitation, cookhouses, fire prevention and fire-fighting equipment, air conditioning, cookers, refrigerators, furniture and other requirements in connection with such accommodation or amenities. On completion of the Contract, unless otherwise agreed with the Employer, the temporary camps / housing provided by the Contractor shall be removed and the site reinstated to its original condition, all to the approval of the Engineer.

6.4. Health and Safety

Due precautions shall be taken by the contractor, and at his own cost, to ensure the safety of his staff and labour and, in collaboration with and to the requirements of the local health authorities, to ensure that medical staff, first aid equipment and store, sick bay and suitable ambulance service to the maximum extent the site conditions permit, are available at the camps, housing and on the Site at all times throughout the period of the contract and that suitable arrangements are made for the prevention of epidemics and for all necessary welfare and hygiene requirements.

6.5. Measures against Insect and Pest Nuisance

The Contractor shall at all times take the necessary precautions to protect all staff and labour employed on the site from insect nuisance, rats and other pests and reduce the dangers to health and the general nuisance occasioned by the same. The Contractor shall provide his staff and labour with suitable prophylactics for the prevention of malaria and take steps to prevent the formation of stagnant pools of water. He shall comply with all the regulations of the local health authorities in these respects and shall in particular arrange to spray thoroughly with approved insecticide all buildings erected on the Site. Such treatment shall be carried out at least once a year or as frequently as necessary or as instructed by the Engineer. The contractor shall warn his staff and labour of the dangers of hazardous flora and fauna within the project area.

6.6. Epidemics

In the event of any outbreak of illness of an epidemic nature, the contractor shall comply with and carry out such regulations, orders and requirements as may be made by the Government, or the local medical or sanitary authorities, for the purpose of dealing with and overcoming the same.

6.7. Arms and ammunition

The Contractor shall not give, barter or otherwise dispose of to any person, any arms or ammunitions of any kind, or allow Contractor's personnel to do so.

6.8. Fair Wages

The Contractor shall pay to labourer employed by him either directly or through sub-contractors' wages not less than wages as defined in Minimum Wages Act 1790 and Contract Labour (Regulation and Abolition) Act 1970 amended from time to time and rules framed there-under and other labour laws affecting contract labour that may be brought in force from time to time.

In respect of labour directly or indirectly employed in the works, the contractor shall comply with the rules and regulations on the maintenance of suitable records prescribed for this purpose from time to time by the Government. He shall maintain his accounts and vouchers on the payment of wages to the labourers to the satisfaction of the Engineer. The Engineer shall have the right to call for such record as required to satisfy himself on the payment of fair wages to the labourers and shall have the right to deduct from the contract amount a suitable amount for making good the loss suffered by the worker or workers by reason of the "fair wages" clause to the workers.

The contractor shall be primarily liable for all payments to be made and for the observance of the regulations framed by the Govt. from time to time without

prejudice to his right to claim indemnity from his sub-contractors. Equal wages are to be paid for both men and women if the nature of work is same and similar.

6.9. Return of Labour and Contractor's Equipment

The Contractor shall, if required by the Engineer, deliver to the Engineer a return in detail, in such form and at such intervals as the Engineer may prescribe, showing the staff and the numbers of the several classes of labour from time to time employed by the Contractor on the Site and such information respecting Contractor's Equipment as the Engineer may require.

6.10. Child labour

The contractor shall not engage children below the age of 14, directly or indirectly, for the execution of this contract. Contractor shall observe compliance to all the acts pertaining to child labour, health & safety and any other such laws existing in the Employers country. If at any time children below the age of 14 are found to be engaged, directly or indirectly, by the contractor or his representatives, the contractor shall pay penalties imposed by concerned regulatory bodies for practicing child labour.

6.11. Removal of Contractor's men

The contractor shall employ for the execution of the works only such persons who are skilled and experienced in their respective trades and Engineer-in-charge shall be at liberty to object and to ask the contractor to remove from the works any persons employed by the contractor for the execution of the works who, in the opinion of the EIC, misconduct themselves or are incompetent or negligent in the proper performance of their duties. The contractor shall forthwith comply with such requisition and such persons shall not be again employed upon the works without the written permission of the EIC. Any person so removed from the work shall be immediately replaced at the expense of the contractor by a qualified and competent substitute. Should the contractor be requested to repatriate any person removed from the work, he shall do so and shall bear all costs in connection herewith.

6.12. Replacement of Key Personnel

The replacement of the key personnel shall only be on health grounds of the individual or if the personnel ceases to work for the contractor and is no longer an employee of the contractor. The contractor shall not replace any of the key personnel without the written consent of the Engineer-in-charge. In case Contractor engages in such activity i.e. replacement of key personnel with or without EICs consent, such action shall attract a deduction of 10% of the total amount from the running account bill. However, there shall be no deduction if the replacement is made on health grounds of the personnel. The contractor shall

submit medical certificate of such personnel intended to be replaced on health grounds.

6.13. Labour laws

The Contractor shall also comply with the provisions of all Acts, Laws, any Regulation or Bye Laws of any Local or other Statutory Authority applicable in relation to the execution of works such as:

- a) Payment of Wages Act, 1936 (Amended)
- b) Minimum Wages Act, 1790 (Amended).
- c) The Contract Labour (Regulation & Abolition) Act, 1970 with Rules framed there under as amended.
- d) Workmen Compensation Act, 1923 as amended by Amendment Act no.65 of 1976.
- e) Employer's Liability Act 1938 (Amended)
- f) Maternity Benefit Act. 1961 (Amended)
- g) The Industrial Employment (Standing orders) Act 1946 (Amended).
- h) The Industrial Disputes Act. Amended 2010
- i) Payment of Bonus Act and Rules 1976
- j) The Personal Injuries (Compensation Insurance) Act 1963 and any modifications thereof and rules made thereunder from time to time. The Contractor shall take into account all the above and financial liabilities in his quoted rates and nothing extra, whatsoever, shall be payable to him on this account.

The list is indicative only, otherwise the contractor should be aware of all the Acts/Labour Laws and should follow diligently on the work. The contractor shall be fully and personally responsible for the violation of any Act/Labour Law. Notwithstanding any other provision contained herein, the Contractor shall keep the Employer informed at all times, against all claims/actions that may arise under any applicable labour laws in relation to any labour/staff employed/hired by the contractor and /or its Sub contractor, for the execution of works awarded to the Contractor pursuant to this Contract.

7. Materials, Plant and Workmanship

7.1. Quality of Materials, Plant and Workmanship

All materials, Plant and workmanship shall be:

- (a) of the respective kinds described in the Contract and in accordance with the Engineer's instructions, and
- (b) subjected from time to time to such tests as the Engineer may require at the place of manufacture, fabrication, or preparation, or on the Site or at such other place or places as may be specified in the Contract, or at all or any of such places.

The Contractor shall provide such assistance, labour, electricity, fuels, stores, apparatus, and instruments as are normally required for examining, measuring and testing any materials or Plant and shall supply samples of materials, before incorporation in the Works, for testing as may be selected and required by the Engineer.

The Contractor is encouraged, to the extent practicable and reasonable, to use materials, Contractor's Equipment, Plant, and supplies from sources within the country of the Employer and the country where Works are executed.

7.2. Cost of Samples

All samples shall be supplied by the Contractor at his own cost if the supply thereof is clearly intended by or provided for in the Contract.

7.3. Cost of Tests

The cost of making any test shall be borne by the Contractor if such test is:

- (a) clearly intended by or provided for in the Contract, or
- (b) particularized in the Contract (in cases of a test under load or of a test to ascertain whether the design of any finished or partially finished work is appropriate for the purposes which it was intended to fulfil) in sufficient detail to enable the Contractor to price or allow for the same in his Tender.

7.4. Cost of Tests not provided for

If any test required by the Engineer which is:

- (a) not provided for,
- (b) (in the cases above mentioned) not so particularized, or
- (c) (though so intended or provided for) required by the Engineer to be carried out at any place other than the Site or the place of manufacture, fabrication or preparation of the materials or Plant tested, shows the materials, Plant, or workmanship not to be in accordance with the provisions of the Contract to the satisfaction of the Engineer, then the cost of such test shall be borne by the Contractor, but in any other case Sub-Clause 7.5 shall apply.

7.5. Engineer's Determination where Tests not provided for

Where, pursuant to Sub-Clause 7.4, this Sub-Clause applies, the Engineer shall, after due consultation with the Contractor, determine any extension of time to which the Contractor is entitled under sub clause 8.6,8.7 and 8.8 of clause 8.

7.6. Inspection of Operations

The Engineer, and any person authorized by him, shall at all reasonable times have access to the Site and to all workshops and places where materials or Plant are being manufactured, fabricated, or prepared for the Works and the Contractor shall afford every facility for and every assistance in obtaining the right to such access.

7.7. Inspection and Testing

The Engineer shall be entitled, during manufacture, fabrication or preparation to inspect and test the materials and Plant to be supplied under the Contract. If materials or Plant are being manufactured, fabricated or prepared in workshops or places other than those of the Contractor, the Contractor shall obtain permission for the Engineer to carry out such inspection and testing in those workshops or places. Such inspection or testing shall not release the Contractor from any obligation under the Contract.

7.8. Dates for Inspection and Testing

The Contractor shall agree with the Engineer on the time and place for the inspection or testing of any materials or Plant as provided in the Contract. The Engineer shall give the Contractor not less than 24 hours" notice of his intention to carry out the inspection or to attend the tests. If the Engineer, or his duly authorized representative, does not attend on the date agreed, the Contractor may, unless otherwise instructed by the Engineer, proceed with the tests and forthwith forward to the Engineer duly certified copies of the test readings. The Engineer will at the earliest but not later than 7 days from the date of receipt of such test-report, carryout scrutiny of the test readings and communicate to Contractor his concurrence to same or objections with reasons and instruction on further steps to be taken. If no communication is made by the Engineer within 7 days, the test shall be deemed to have been made in the presence of the Engineer.

7.9. Rejection

If, at the time and place agreed in accordance with Sub-Clause 7.8, the materials or Plant are not ready for inspection or testing or if, as a result of the inspection testing referred to in this Clause, the Engineer determines that the materials or Plant are defective or otherwise not in accordance with the Contract, he may reject the materials or Plant and shall notify the Contractor thereof immediately. The notice shall state the Engineer's objections with reasons. The Contractor shall then promptly make good the defect or ensure that rejected materials or Plant comply

with the Contract. If the Engineer so requests, the tests of rejected materials or Plant shall be made or repeated under the same terms and conditions. All costs incurred by the Engineer by the repetition of the tests shall, after due consultation with the employer, be determined by the Engineer and shall be recoverable from the Contractor by the Employer and may be deducted from any monies due or to become due to the Contractor and the Engineer shall notify the Contractor accordingly, with a copy to the employer.

7.10. Independent Inspection

The Engineer may delegate inspection and testing of materials, Works or Plant to an independent inspector or inspecting agency of specialised competence, experience or accreditation in carrying out such inspection. Any such delegation shall be effected in accordance with Sub-Clause 3.4 and for this purpose such independent inspector shall be considered as an assistant of the Engineer.

7.11. Examination of Work before Covering up

No part of the Works shall be covered up or put out of view without the approval of the Engineer and the Contractor shall afford full opportunity for the Engineer to examine and measure any such part of the Works which is about to be covered up or put out of view and to examine foundations before any part of the Works is placed thereon. The Contractor shall give notice to the Engineer whenever any such part of the Works or foundations is or are ready or about to be ready for examination and the Engineer shall, without unreasonable delay, unless he considers it unnecessary and advises the Contractor accordingly, attend for the purpose of examining and measuring such part of the Works or of examining such foundations.

7.12. Uncovering and Making Openings

The Contractor shall uncover any part of the Works or make openings in or through the same as the Engineer may from time to time instruct and shall reinstate and make good such part. If any such part has been covered up or put out of view after compliance with the requirement of Sub-Clause 7.11 and is found to be executed in accordance with the Contract, the Engineer shall, after due consultation with the Employer and the Contractor, determine the amount of the Contractor's costs in respect of such of uncovering, making openings in or through, reinstating and making good the same, which shall be added to the Contract Price, and shall notify the Contractor accordingly, with a copy to the Employer. In any other case all costs shall be borne by the Contractor.

7.13. Removal of Improper Work, Materials or Plant

The Engineer shall have authority to issue instructions from time to time, for:

- (a) the removal from the Site, within such time or times as may be specified in the instruction, of any materials or Plant which, in the opinion of the Engineer, are not in accordance with the Contract,
- (b) the substitution of proper and suitable materials or Plant, and
- (c) the removal and proper re-execution, notwithstanding any previous test thereof or interim payment therefor, of any work which, in respect of
 - (i) materials, Plant or workmanship, or
 - (ii) executed, installed by the Contractor or for which he is responsible, is not, in the opinion of the Engineer, in accordance with the Contract.

7.14. Default of Contractor in Compliance

In case of default on the part of the Contractor in carrying out such instruction within the time specified therein or, if none, within a reasonable time, the Employer shall be entitled to employ and pay other persons to carry out the same and all costs consequent thereon or incidental thereto shall, after due consultation with the Employer and the Contractor, be determined by the Engineer and shall be recoverable from the Contractor by the Employer, and may be deducted by the Employer from any monies due or to become due to the Contractor and the Engineer shall notify the Contractor accordingly, with a copy to the Employer.

8. Commencement, delays and suspensions

8.1. Commencement of Works

The Contractor shall commence the Works from the date of signing of the Contract Agreement. Thereafter, the Contractor shall proceed with the Works with due expedition and without delay.

8.2. Possession of Site and Access Thereto

As the Contract may prescribe:

- (a) the extent of portions of the Site of which the Contractor is to be given possession
- (b) the order in which such portions shall be made available to the Contractor, and, subject to any requirement in the Contract as to the order in which the Works shall be executed, the Employer will, with the Engineer's notice to commence the Works, give to the Contractor possession of
- (c) so much of the Site, and
- (d) such access as, in accordance with the Contract, is to be provided by the Employer as may be required to enable the Contractor to commence and

proceed with the execution of the Works in accordance with the programme referred to in Clause 4.9, if any, and otherwise in accordance with such reasonable proposals as the Contractor shall, by notice to the Engineer with a copy to the Employer, make. The Employer will, from time to time as the Works proceed, give to the Contractor possession of such further portions of the Site as may be required to enable the Contractor to proceed with the execution of the Works with due dispatch in accordance with such programme or proposals, as the case may be.

8.3. Failure to Give Possession

If the Contractor suffers delay and / or incurs costs from failure on the part of the Employer to give possession and access in accordance with the terms of Sub-Clause 8.2, the Engineer shall, after due consultation with the Employer and the Contractor, determine:

(a) any extension of time to which the Contractor is entitled under sub clause 8.6, 8.7 and 8.8 of clause 8,

8.4. Rights of Way and Facilities

The Contractor shall bear all costs and charges for special or temporary rights of way required by him in connection with access to the Site. The Contractor shall also provide at his own cost any additional facilities outside the Site required by him for the purposes of the Works

8.5. Time for Completion

The Contractor agrees that Time is of essence. The whole of the Works and, if applicable, any Section required to be completed within a particular time as stated in the SCC, shall be completed in accordance with the provisions of Clause 10, calculated from the Commencement Date, or such extended time as may be allowed under sub clause 8.6, 8.7 and 8.8 of clause 8.

8.6. Extension of Time for Completion

In the event of:

- (a) the amount or nature of extra or additional work,
- (b) any cause of delay referred to in these Conditions,
- (c) exceptionally adverse climatic conditions,
- (d) any delay, impediment or prevention by the Employer, or
- (e) other special circumstances which may occur, other than through a default of or breach of contract by the Contractor or for which he is responsible, being such as fairly to entitle the Contractor to an extension of the Time for

Completion of the Works, or any Section or part thereof, the Engineer shall, after due consultation with the Employer and the Contractor, determine the amount of such extension and shall notify the Contractor accordingly, with a copy to the Employer.

8.7. Contractor to Provide Notification and Detailed Particular

Provided that the Engineer is not bound to make any determination unless the Contractor has

- (a) within 28 days after such event has first arisen notified the Engineer with a copy to the Employer, and
- (b) within 28 days, or such other reasonable time as may be agreed by the Engineer, after such notification submitted to the Engineer detailed particulars of any extension of time to which he may consider himself entitled in order that such submission may be investigated at the time.

8.8. Interim Determination of Extension

Provided also that where an event has a continuing effect such that it is not practicable for the Contractor to submit detailed particulars within the period of 28 days referred to in Sub-Clause 8.7 (b), he shall nevertheless be entitled to an extension of time provided that he has submitted to the Engineer interim particulars at intervals of not more than 28 days and final particulars within 28 days of the end of the effects resulting from the event. On receipt of such interim particulars, the Engineer shall, without undue delay, make an interim determination of extension of time and, on receipt of the final particulars, the Engineer shall review all the circumstances and shall determine an overall extension of time in regard to the event. In both such cases the Engineer shall make his determination after due consultation with the Employer and the Contractor and shall notify the Contractor of the determination, with a copy to the Employer. No final review shall result in a decrease of any extension of time already determined by the Engineer.

8.9. Restriction on Working Hours

Subject to any provision to the contrary contained in the Contract, none of the Works shall, save as hereinafter provided, be carried on during the night or on locally recognised days of rest without the consent of the Engineer. Provided that the provisions of this Clause shall not be applicable in the case of any work which it is customary to carry out by multiple shifts.

8.10. Progress Reports

Unless otherwise stated, the contractor shall prepare and submit the monthly progress reports concerning to the contract in such a form as may be required by

the EIC on behalf of authority. The first report shall cover the period up to the end of the first calendar month following the Commencement Date. Reports shall be submitted monthly thereafter, each within 7 days after the last day of the period to which it relates. Reporting shall continue until the Contractor has completed all work which is known to be outstanding at the completion date stated in the Completion certificate for the Works.

The submission, receipt or acceptance of such reports shall not prejudice the right of the owner under the contract, nor shall operate as a stopple against the owner merely by the reason of the fact that he has not taken notice of or objected to any information contained in such report.

Each report shall include:

- (a) charts and detailed descriptions of progress, including each stage of execution, Contractor's Documents, procurement, manufacture, delivery to Site, construction, erection, testing, commissioning and trial operation;
- (b) photographs showing the status of manufacture and of progress on the Site;
- (c) for the manufacture of each main item of Plant and Materials, the name of the manufacturer, manufacture location, percentage progress, and the actual or expected dates of:
 - (i) commencement of manufacture,
 - (ii) Contractor's inspections,
 - (iii) tests, and
 - (iv) shipment and arrival at the Site;
- (d) the details of Contractor's Personnel and Equipment deployed during the report period;
- (e) list of Variations, and notices given, if any, under Sub-Clauses related to Employer's Claims and Contractor's Claims;
- (f) safety statistics, including details of any hazardous incidents and activities relating to environmental aspects and public relations; and
- (g) comparisons of actual and planned progress, with details of any events or circumstances which may jeopardize the completion in accordance with the Contract, and the measures being (or to be) adopted to overcome delays.
- (h) One month target to be taken up along with comments to catch up the slipped milestone/target.

8.11. Rate of Progress

If for any reason, which does not entitle the Contractor to an extension of time, the rate of progress of the Works or any Section is at any time, in the opinion of the Engineer, too slow to comply with the Time for Completion, the Engineer shall so notify the Contractor who shall thereupon take such steps as are necessary, subject to the consent of the Engineer, to expedite progress so as to comply with the Time for Completion. The Contractor shall not be entitled to any additional payment for taking such steps.

8.12. Liquidated Damages for Delay

8.12.1. Liquidated Damages

If the Contractor fails to comply with the Time for Completion in accordance with Clause 10, for the whole of the Works or, if applicable, any Section within the relevant time prescribed by Clause 8.5, then the Contractor shall pay to the Employer the relevant sum stated below in the tabular form as liquidated damages for such default and not as a penalty (which sum shall be the only monies due from the Contractor for such default). The Employer may, without prejudice to any other method of recovery, withheld / deduct the amount of such damages from any monies due or to become due to the Contractor. The amount withheld or deduction against such damages shall not relieve the Contractor from his obligation to complete the Works, or from any other of his obligations and liabilities under the Contract.

Sl. No.	Milestone	Time Allowed from the Date of signing of Contract	
1.	Submission of Draft Design for approval of Employer	20 days	Nil
2.	Submission of Final Design Certified by IRS/ Classification Society for approval of Employer	45 days	10% of Contract Value
3.	a) After the fabrication of entire hull structure after inspection and certification by IRS / Classification Society.	120 days	25% payment
	b) After delivery and installation of all machinery & equipments, completion of Superstructure & Interiors in all respects and Safety measures etc. and certified by IRS/Classification Society thereof.	165 days	25% payment
4.	a) On trial & testing and successful commissioning at site	180 days	30% payment
	b) On successful running of 12 (twelve) months.	••••	10% payment

8.12.2. If the contractor does not achieve a particular milestone as referred in clause 8.12.1 of detailed scope of work, or the re-scheduled milestone(s) in terms of Clause 8.6, 8.7 and 8.8, the amount as stipulated above, against that milestone shall be withheld. With-holding of this amount on failure to achieve a milestone, shall be automatic without any notice to the contractor. However, if the contractor catches up with the progress of work on the subsequent milestone(s), the withheld amount shall be released. In case the contractor fails to make up for the delay in subsequent milestone(s), amount as stipulated in the table above, against each milestone missed subsequently shall also be withheld. However, no interest, whatsoever, shall be payable on such withheld amount. Permanent deduction of LD shall be as per clause

8.13. Permanent deduction of Liquidated Damages

In case the contractor does not complete the work in time, extension of time shall be given to the contractor to complete the work. It shall be at sole discretion of the EIC, who shall take all causes of delay into account, to provide extension of time withholding the liquidated damages accrued till the completion of commissioning of vessels. The EIC may even release the accrued liquidated damages in case he finds the causes of delay were attributable to employer or to other reasons for which the contractor cannot be held accountable.

8.14. Suspension of Work

The Contractor shall, on the instructions of the Employer communicated by the Engineer suspend the progress of the Works or any part thereof for such time or times and in such manner as the Engineer may consider necessary and shall, during such suspension, properly protect and secure the Works or such part thereof so far as is necessary in the opinion of the Engineer. Unless such suspension is:

- (a) otherwise provided for in the Contract,
- (b) necessary by reason of some default of or breach of contract by the Contractor or for which he is responsible,
- (c) necessary by reason of climatic conditions on the Site, or
- (d) necessary for the proper execution of the Works or for the safety of the Works or any part thereof (save to the extent that such necessity arises from any act or default by the Engineer or the Employer or from any of the risks defined in Sub-Clause 2.2), Sub-Clause 8.15 shall apply.

8.15. Engineer's Determination following Suspension,

Where, pursuant to Sub-Clause 8.14, this Sub-Clause applies the Engineer shall after due consultation with the Employer and the Contractor, determine

- (a) any extension of time to which the Contractor is entitled under sub clause 8.6,8.7 and 8.8 of clause 8, and
- (b) the amount, if any, which shall be added to the Contract Price, in respect of the cost incurred by the Contractor by reason of such suspension, and shall notify the Contractor accordingly, with a copy to the Employer.

8.16. Suspension lasting more than 84 Days

If the progress of the Works or any part thereof is suspended on the instructions of the Engineer and if permission to resume work is not given by the Engineer within a period of 84 days from the date of suspension then, unless such suspension is within paragraph (a), (b), (c) or (d) of Sub-Clause 8.14, the Contractor may give notice to the Engineer requiring permission, within 28 days from the receipt thereof, to proceed with the Works or that part thereof in regard to which progress is suspended. If, within the said time, such permission is not granted, the Contractor may, but is not bound to, elect to treat the suspension, where it affects part only of the Works, as an omission of such part under Clause 12.1 and 12.2 by giving a further notice to the Engineer to that effect, or, where it affects the whole of the Works, treat the suspension as an event of default by the Employer and terminate his employment under the Contract in accordance with the provisions of Sub-Clause 17.1, whereupon the provisions of Sub-Clauses 17.2 and 17.3 shall apply.

9. Tests On Completion

9.1. Contractors obligation

Contractor shall carry out all tests as per requirement of this contract. No additional charges on any such account shall be payable by the Employer.

The Contractor shall give to the Engineer not less than 21 days' notice of the date after which the Contractor will be ready to carry out each of the Tests on completion. Unless otherwise agreed, Tests on completion shall be carried out within 14 days after this date, on such day or days as the Engineer shall instruct.

In considering the results of the Tests on Completion, the Engineer shall make allowances for the effect of any use of the Works by the Employer on the performance or other characteristics of the Works. As soon as the Works, or a Section, have passed any Tests on Completion, the Contractor shall submit a certified report of the results of these Tests to the Engineer.

9.2. Delayed Test

If the Contractor is prevented, for more than 14 days, from carrying out the survey on Completion by a cause for which the Employer is responsible, the Employer shall be deemed to have taken over the Works on the date when the survey on Completion would otherwise have been completed.

If the Contractor suffers delay and/or incurs Cost as a result of this delay in carrying out the survey on Completion, the Contractor shall give notice to EMPLOYER / ENGINEER and shall be entitled.

- (a) an extension of time for any such delay, if completion is or will be delayed, and
- (b) Payment of any such Cost-plus reasonable profit, which shall be additional to the Contract Price.

After receiving this notice, ENGINEER / EMPLOYER shall proceed to agree or determine these matters.

9.3. Retesting

If the Works, or a Section, fall to pass the Tests on Completion, the Contractor may require the failed Tests, and Tests of Completion on any related work, to be repeated under the same terms and conditions.

9.4. Failure to Remedy Defects

If the Contractor fails to remedy any damage within a reasonable time, a date may be fixed by (or on behalf of) the Employer, on or by which the damage is to be remedied. The Contractor shall be given reasonable notice of this date.

If the Contractor fails to remedy the damage by this notified date and this Remedial work was to be executed at the cost of the Contractor.

- (a) carry out the work himself or by others, in a reasonable manner and at the Contractor's cost, but the Contractor shall have no responsibility for this work; and the Contractor shall pay to the Employer the costs reasonably incurred by the Employer in remedying the damage;
- (b) Require ENGINEER / EMPLOYER to agree or determine a reasonable reduction in the Contract Price.

10. Employer's Taking over

10.1. Completion Certificate

When the whole of the Works has been substantially completed and have satisfactorily passed any Tests on Completion prescribed by the Contract, the Contractor may give a notice to that effect to the Engineer, with a copy to the Employer, accompanied by a written undertaking to finish with due expedition any outstanding work during the maintenance period. Such notice and undertaking shall be deemed to be a request by the Contractor for the Engineer to issue a Completion certificate in respect of the Works. The Engineer shall, within 21 days of the date of delivery of such notice, either issue to the Contractor, with a copy to the Employer, a completion Certificate, stating the date on which, in his opinion, the Works were substantially completed in accordance with the Contract, or give instructions in writing to the Contractor specifying all the work which, in the Engineer's opinion, is required to be done by the Contractor before the issue of such Certificate. The Engineer shall also notify the Contractor of any defects in the Works affecting substantial completion that may appear after such instructions and before completion of the Works specified therein. The Contractor shall be entitled to receive such Completion certificate within 21 days of completion, to the satisfaction of the Engineer, of the Works so specified and remedying any defects so notified.

10.2. Taking Over of Sections or Parts

Similarly, in accordance with the procedure set out in Sub-Clause 10.1, the Contractor may request and the Engineer shall issue a Completion certificate in respect of:

- (a) any Section in respect of which a separate Time for Completion is provided in the SCC,
- (b) any substantial part of the Permanent Works which has been both completed to the satisfaction of the Engineer and, otherwise than as provided for in the Contract, occupied or used by the Employer, or
- (c) any part of the Permanent Works which the Employer has elected to occupy or use prior to completion (where such prior occupation or use is not provided for in the Contract or has not been agreed by the Contractor as a temporary measure).
- (d) Punch list of the rectifications to be issued by the Engineer to the contractor.

10.3. Substantial Completion of Parts

If any part of the Permanent Works has been substantially completed and has satisfactorily passed any Tests on Completion prescribed by the Contract, the

Engineer may issue a Completion certificate in respect of that part of the Permanent Works before completion of the whole of the Works and, upon the issue of such Certificate, the Contractor shall be deemed to have undertaken to complete with due expedition any outstanding work in that part of the Permanent Works during the maintenance period.

10.4. Approval only by Final Completion Certificate

Only the Final Completion Certificate, referred to in clause 10.5 and 10.6, shall be deemed to constitute final approval of works.

10.5. Final Completion Certificate

The Contract shall not be considered as completed until a Final Completion Certificate shall have been signed by the Engineer and delivered to the Employer, with a copy to, the Contractor, stating the date on which the Contractor shall have completed his obligations to execute and complete the Works and remedy any defects therein to the Engineer's satisfaction. The Final Completion Certificate shall be given by the Engineer within 28 days after the expiration of the Work Period.

10.6. Unfulfilled Obligations

Notwithstanding the issue of the Final Completion Certificate the Contractor and the Employer shall remain liable for the fulfilment of any obligation incurred under the provisions of the Contract prior to the issue of the Final Completion Certificate which remains unperformed at the time such Final Completion Certificate is issued and, for the purposes of determining the nature and extent of any such obligation, the Contract shall be deemed to remain in force between the parties to the Contract.

11. Measurement & omissions

11.1. Quantities

The quantities set out in the Bill of Quantities are the estimated quantities for the Works, and they are not to be taken as the actual and correct quantities of the Works to be executed by the Contractor in fulfilment of his obligations under the Contract.

11.2. Work to be Measured

The Engineer, shall except as otherwise stated, ascertain, and determine by measurement the value of the Works in accordance with the Contract and the Contractor shall be paid that value in accordance with Clause 14. The Engineer shall, when he requires any part of the Works to be measured, give reasonable notice to the Contractor's authorized agent, who shall:

- (a) forthwith attend or send a qualified representative to assist the Engineer in making such measurement, and
- (b) supply all particulars required by the Engineer.

Should the Contractor not attend, or neglect or omit to send such representative, then the measurement made by the Engineer or approved by him shall be taken to be the correct measurement of such part of the Works. For the purpose of

measuring such Permanent Works as are to be measured by records and drawings, the Engineer shall prepare records and drawings as the work proceeds and the Contractor, as and when called upon to do so in writing, shall, within 14 days, attend to examine and agree such records and drawings with the Engineer and shall sign the same when so agreed. If the Contractor does not attend to examine and agree such records and drawings, they shall be taken to be correct. If, after examination of such records and drawings, the Contractor does not agree the same or does not sign the same as agreed, they shall nevertheless be taken to be correct, unless the Contractor, within 14 days of such examination, lodges with the Engineer notice of the respects in which such records and drawings are claimed by him to be incorrect. On receipt of such notice, the Engineer shall review the records and drawings and either confirm or vary them.

11.3. Method of Measurement

The Works shall be measured net, notwithstanding any general or local custom, except where otherwise provided for in the Contract.

11.4. Breakdown of Lump Sum Items

For the purposes of statements submitted in accordance with Sub-Clause 15.1, the Contractor shall submit to the Engineer, within 21 days after the receipt of the Letter of Award, a breakdown for each of the lump sum items contained in the Tender. Such breakdowns shall be subject to the approval of the Engineer.

11.5. Omissions

Whenever the omission of any work forms part (or all) of a Variation, the value of which has not been agreed, if:

- (a) The Contractor will incur (or has incurred) cost which, if the work had not been omitted, would have been deemed to be covered by a sum forming part of the Accepted Contract Amount;
- (b) The omission of the work will result (or has resulted) in this sum not forming part of the Contract Price; and
- (c) This cost is not deemed to be included in the evaluation of any substituted work; then the Contractor shall give notice to ENGINEER / EMPLOYER accordingly, with supporting particulars. Upon receiving this notice, ENGINEER / EMPLOYER shall agree or determine this cost, which shall be included in the Contract Price.

12. Variations, adjustments, and Additions

12.1. Variation

The Engineer shall make any variation of the form, quality or quantity of the Works or any part thereof that may, in his opinion, be necessary and for that purpose, or if for any other reason it shall, in his opinion, be appropriate, he shall have the authority to instruct the Contractor to do and the Contractor shall do any of the following:

- (a) increase or decrease the quantity of any work included in the Contract,
- (b) omit any such work (but not if the omitted work is to be carried out by the Employer or by another contractor),
- (c) change the character or quality or kind of any such work,
- (d) change the levels, lines, position and dimensions of any part of the Works,
- (e) execute additional work of any kind necessary for the completion of the Works, or
- (f) change any specified sequence or timing of construction of any part of the Works.

No such variation shall in any way vitiate or invalidate the Contract, but the effect, if any, of all such variations shall be valued in accordance with Clause 12.3, 12.4 and 12.5. Provided that where the issue of an instruction to vary the Works is necessitated by some default of or breach of contract by the Contractor or for which he is responsible, any additional cost attributable to such default shall be borne by the Contractor.

12.2. Instructions for Variations

The Contractor shall not make any such variation without an instruction of the Engineer.

12.3. Valuation of Variations

All variations referred to in Clause 12.1 & 12.2 and any additions to the Contract Price which are required to be determined in accordance with Clause 12.3, 12.4 and 12.5 (for the purposes of this Clause referred to as "varied work"), shall be valued at the rates and prices set out in the Contract if, in the opinion of the Engineer, the same shall be applicable. If the Contract contain any rates or prices applicable to the varied work, the rates and prices in the Contract shall be used as the basis for valuation so far as may be reasonable, failing which, after due consultation by the Engineer with the Employer and the Contractor, suitable rates or prices shall be agreed upon between the Engineer and the Contractor. In the event of disagreement, the Engineer shall fix such rates or prices as are, in his opinion, appropriate and shall notify the Contractor accordingly, with a copy to the Employer. Until such time as rates or prices are agreed or fixed, the Engineer

shall determine provisional rates or prices to enable on-account payments to be included in certificates issued in accordance with Clause 14.

12.4. Power of Engineer to Fix Rates

Provided that if the nature or amount of any varied work relative to the nature or amount of the whole of the Works or to any part thereof, is such that, in the opinion of the Engineer, the rate or price contained in the Contract for any item of the Works is, by reason of such varied work, rendered inappropriate or inapplicable, then, after due consultation by the Engineer with the Employer and the Contractor, a suitable rate or price shall be agreed upon between the Engineer and the Contractor. In the event of disagreement, the Engineer shall fix such other rate or price as is, in his opinion, appropriate and shall notify the Contractor accordingly, with a copy to the Employer. Until such time as rates or prices are agreed or fixed, the Engineer shall determine provisional rates or prices to enable on-account payments to be included in certificates issued in accordance with Clause 14.

Provided also that no varied work instructed to be done by the Engineer pursuant to Clause 12.1 &12.2 shall be valued under Sub-Clause 12.3 or under this Sub-Clause unless, within 14 days of the date of such instruction and, other than in the case of omitted work, before the commencement of the varied work, notice shall have been given either:

- (a) by the Contractor to the Engineer of his intention to claim extra payment or a varied rate or price, or
- (b) by the Engineer to the Contractor of his intention to vary a rate or price/

12.5. Variations Exceeding 20 per cent

The unit rate quoted by the Contractor shall remain unchanged for the quantities varying up to \pm 0%. If, on the issue of the Completion Certificate for the whole of the Works, it is found that as a result of,

- (a) all varied work valued under Sub-Clauses 12.3 and 12.4, and
- (b) all adjustments in the estimated quantities set out in the Bill of Quantities, excluding adjustments of price made under Clause 22,

but not from any other cause, there have been additions to or deductions from the Contract Price in excess of 20 per cent at variance with the Contract Price, then and in such event, after due consultation by the Engineer with the Employer and the Contractor, certain amount shall be added or subtracted from the contract price. The Engineer shall notify the Contractor of any determination made under this Sub- Clause, with a copy to the Employer. Such sum shall be based only on

the amount by which such additions or deductions shall be at variance in excess of 20 per cent with the Contract Price

13. Force Majeure

13.1. No Liability for Force Majeure

The Contractor shall be under no liability whatsoever in consequence of any of the special risks referred to in Sub-Clause 13.2, whether by way of indemnity or otherwise, for or in respect of:

- (a) Destruction of or damage to the Works, save to work condemned under the provisions of Clause 7.13 and 7.14 prior to the occurrence of any of the said special risks,
- (b) Destruction of or damage to property, whether of the Employer or third parties, or
- (c) Injury or loss of life.

13.2. Force Majeure

The Force Majeure is:

- (a) the risks defined under paragraphs (a), (c), (d) and (e) of Sub-Clause 2.2, and
- (b) the risks defined under paragraph (b) of Sub-Clause 2.2 insofar as these relate to the country in which the Works are to be executed.

13.3. Damage to Works by Force Majeure

If the Works or any materials or Plant on or near or in transit to the Site, sustain destruction or damage by reason of any of the said special risks, the Contractor shall be entitled to payment in accordance with the Contract for any Permanent Works duly executed and for any materials or Plant so destroyed or damaged and, so far as may be required by the Engineer or as may be necessary for the completion of the Works, to payment for:

- (a) Rectifying any such destruction or damage to the Works, and
- (b) Replacing or rectifying such materials or plant, and the Engineer shall determine an addition to the Contract Price in accordance with Clause 12.3, 12.4 & 12.5 and shall notify the Contractor accordingly, with a copy to the Employer

13.4. Projectile, Missile

Destruction, damage, injury or loss of life caused by the explosion or impact, whenever and wherever occurring, of any mine, bomb, shell, grenade, or other

projectile, missile, munitions, or explosive of war, shall be deemed to be a consequence of the said special risks.

13.5. Increased Costs arising from Force Majeure

Save to the extent that the Contractor is entitled to payment under any other provision of the Contract, the Employer shall pay to the Contractor any costs of the execution of the Works (other than such as may be attributable to the cost of reconstructing work condemned under the provisions of Clause 7.13 & 7.14 prior to the occurrence of any special risk) which are howsoever attributable to or consequent on or the result of or in any way whatsoever connected with the said special risks, subject however to the provisions in this Clause hereinafter contained in regard to outbreak of war, but the Contractor shall, as soon as any such cost comes to his knowledge, forthwith notify the Engineer thereof. The Engineer shall, after due consultation with the Employer and the Contractor, determine the amount of the Contractor's costs in respect thereof which shall be added to the Contract Price and shall notify the Contractor accordingly, with a copy to the Employer.

13.6. Outbreak of War

If, during the currency of the Contract, there is an outbreak of war, whether war is declared or not, in any part of the world which, whether financially or otherwise, materially affects the execution of the Works, the Contractor shall, unless and until the Contract is terminated under the provisions of this Clause, continue to use his best endeavours to complete the execution of the Works. Provided that the Employer shall be entitled, at any time after such outbreak of war, to terminate the Contract by giving notice to the Contractor and, upon such notice being given, the Contract shall, except as to the rights of the parties under this Clause and Clause 15.3, terminate, but without prejudice to the rights of either party in respect of any antecedent breach thereof.

13.7. Removal of Contractor's Equipment on Termination

If the Contract is terminated under the provisions of Sub-Clause 13.6, the Contractor shall, with all reasonable dispatch, remove from the Site all Contractor's Equipment and shall give similar facilities to his Subcontractors to do so.

13.8. Payment if Contract Terminated

If the Contract is terminated as aforesaid, the Contractor shall be paid by the Employer, insofar as such amounts or items have not already been covered by payments on account made to the Contractor, for all work executed prior to the date of termination at the rates and prices provided in the Contract and in addition:

- (a) The amounts payable in respect of any preliminary items referred to in the Bill of Quantities, so far as the work or service comprised therein has been carried out or performed, and a proper proportion of any such items which have been partially carried out or performed;
- (b) The cost of materials, Plant or goods reasonably ordered for the Works which have been delivered to the Contractor or of which the Contractor is legally liable to accept delivery, such materials, Plant or goods becoming the property of the Employer upon such payments being made by him;
- (c) A sum being the amount of any expenditure reasonably incurred by the Contractor in the expectation of completing the whole of the Works insofar as such expenditure has not been covered by any other payments referred to in this Sub-Clause;
- (d) Any additional sum payable under the provisions of Sub-Clauses 13.3 and 13.5;
- (e) such proportion of the cost as may be reasonable, taking into account payments made or to be made for work executed, of removal of Contractor's Equipment under Sub-Clause 13.7 and, if required by the Contractor, return thereof to the Contractor's main plant yard in his country of registration or to other destination, at no greater cost; and
- (f) The reasonable cost of repatriation of all the Contractor's staff and workmen employed on or in connection with the Works at the time of such termination.

Provided that against any payment due from the Employer under this Sub-Clause, the Employer shall be entitled to be credited with any outstanding balances due from the Contractor for advances in respect of mobilization and any other sums which, at the date of termination, were recoverable by the Employer from the Contractor under the terms of the Contract. Any sums payable under this Sub-Clause shall, after due consultation with the Employer and the Contractor, be determined by the Engineer who shall notify the Contractor accordingly, with a copy to the Employer.

14. Certificates and Payment

14.1. Monthly Statements

The Contractor shall submit to the Engineer after the end of each month six copies, each signed by the Contractor's representative approved by the Engineer in accordance with Sub-Clause 4.10, of a statement, in such form as the Engineer may from time to time prescribe, showing the amounts expressed in Indian Rupees, to which the Contractor considers himself to be entitled up to the end of the month in respect of:

- (a) The value of the Works executed with reference to the contract
- (b) Any other items in the Bill of Quantities.
- (c) the percentage of the invoice value of listed materials, all as stated in the SCC, and Plant delivered by the Contractor on the Site for incorporation in the Permanent Works but not incorporated in such Works,
- (d) Adjustments under Clause 22, and
- (e) Any other sum to which the Contractor may be entitled under the Contract or otherwise.

14.2. Monthly Payments

The Engineer shall, within 28 days of receiving such statement, deliver to the Employer an Interim Payment Certificate stating the amount of payment to the Contractor which the Engineer considers due and payable in respect of such statement, subject:

- (a) firstly, to the retention of the amount calculated by applying the Percentage of Retention stated in the SCC, to the amount to which the Contractor is entitled under paragraphs (a), (b), (c) and (e) of Sub-Clause 14.1 until the amount so retained reaches the Limit of Retention Money stated in the SCC, and
- (b) secondly, to the deduction, other than pursuant to Clause 8.12 and 8.13, of any sums which may have become due and payable by the Contractor to the Employer.

Provided that the Engineer shall not be bound to certify any payment under this Sub-Clause if the net amount thereof, after all retentions and deductions, would be less than the Minimum Amount of Interim Payment Certificates stated in the SCC

Notwithstanding the terms of this Clause or any other Clause of the Contract no amount will be certified by the Engineer for payment until the performance security, if required under the Contract, has been provided by the Contractor and approved by the Employer.

14.3. Place of Payment

Payments to the Contractor by the Employer shall be made in the currencies in which the Contract Price is payable into a bank account or accounts nominated by the Contractor in the country of the Employer.

14.4. Retention Money

A retention amounting to the percentage stipulated in the SCC shall be made by the Engineer in the first and following Interim Payment Certificates till the specified limit of Retention money as in SCC.

14.5. Payment of Retention Money

Upon the issue of the Completion certificate with respect to the whole of the Works, one half of the Retention Money, or upon the issue of a Completion certificate with respect to a Section or part of the Permanent Works only such proportion thereof as the Engineer determines having regard to the relative value of such Section or part of the Permanent Works, shall be certified by the Engineer for payment to the Contractor.

14.6. Correction of Certificates

The Engineer may by any Interim Payment Certificate make any correction or modification in any previous Interim Payment Certificate which shall have been issued by him and shall have authority, if any work is not being carried out to his satisfaction, to omit or reduce the value of such work in any Interim Payment Certificate.

14.7. Statement at Completion

Not later than 30 days after the issue of the Completion Certificate in respect of the whole of the Works, the Contractor shall submit to the Engineer six copies of Statement at Completion with supporting documents showing in detail, in the form approved by the Engineer:

- (a) the final value of all work done, in Indian Rupees, in accordance with the Contract up to the date stated in such completion Certificate,
- (b) any further sums which the Contractor considers to be due, and
- (c) an estimate of amounts which the Contractor considers will become due to him under the Contract.

The estimated amounts shall be shown separately in such Statement at Completion. The Engineer shall certify payment in accordance with Sub-Clause 14.2.

14.8. Final Statement

Not later than 30 days after the issue of the final completion certificate, the Contractor shall submit to the Engineer for consideration six copies of a draft final statement with supporting documents showing in detail, in the form approved by the Engineer:

(a) the value of all work done in accordance with the Contract, and

(b) any further sums which the Contractor considers to be due to him under the Contract or otherwise.

If the Engineer disagrees with or cannot verify any part of the draft final statement, the Contractor shall submit such further information as the Engineer may reasonably require and shall make such changes in the draft as may be agreed between them. The Contractor shall then prepare and submit to the Engineer the final statement as agreed (for the purposes of these Conditions referred to as the "Final Statement").

If, following discussions between the Engineer and the Contractor and any changes to the draft final statement which may be agreed between them, it becomes evident that a dispute exists, the Engineer shall deliver to the Employer an Interim Payment Certificate for those parts of the draft final statement, if any, which are not in dispute. The dispute may then be settled in accordance with Clause 15.3.

14.9. Discharge

Upon submission of the Final Statement, the Contractor shall give to the Employer, with a copy to the Engineer, a written discharge confirming that the total of the Final Statement represents full and final settlement of all monies due to the Contractor arising out of or in respect of the Contract. Provided that such discharge shall become effective only after payment due under the Final Payment Certificate issued pursuant to Sub-Clause 14.10 has been made and the performance security referred to in Sub- Clause 4.4, if any, has been returned to the Contractor.

14.10. Final Payment Certificate

Within 14 days after receipt of the Final Statement, and the written discharge, the Engineer shall issue to the Employer (with a copy to the Contractor) a Final Payment Certificate stating:

- (a) the amount which, in the opinion of the Engineer, is finally due under the Contract or otherwise, and
- (b) after giving credit to the Employer for all amounts previously paid by the Employer and for all sums to which the Employer is entitled other than under Clause 8.12 and 8.13, the balance, if any, due from the Employer to the Contractor or from the Contractor to the Employer as the case may be.

14.11. Cessation of Employer's Liability

The Employer shall not be liable to the Contractor for any matter or thing arising out of or in connection with the Contract or execution of the Works, unless the Contractor shall have included a claim in respect thereof in his Final Statement and (except in respect of matters or things arising after the issue of the Completion certificate in respect of the whole of the Works) in the Statement at Completion referred to in Sub- Clause 14.7.

14.12. Time for Payment

The amount due to the Contractor under any Interim Payment Certificate issued by the Engineer pursuant to this Clause, or to any other term of the Contract, shall, subject to Clause 8.12 and 8.13, be paid by the Employer to the Contractor within 28 days after such Interim Payment Certificate has been delivered to the Employer, or, in the case of the Final Payment Certificate referred to in Sub-Clause 14.8, within 30 days, after such Final Payment Certificate has been delivered to the Employer..

14.13. Advance Payment

The Contractor shall be eligible to receive an advance payment for an amount stated in the SCC. Payment of such advance amount will be due under certification by the Engineer after (a) execution of the Form of Agreement by the parties hereto; (b) provision by the Contractor of the performance security in accordance with Sub- Clause 4.4; and (c) provision by the Contractor of an unconditional bank guarantee for equal amount in a form and by a bank acceptable to the Employer. Such bank guarantee shall remain effective until the advance payment has been repaid pursuant to the paragraph below, but the amount thereof shall be progressively reduced by the amount repaid by the Contractor as indicated in Interim Payment Certificates issued in accordance with this clause.

The advance payment shall be repaid through percentage deductions from the interim payments certified by the Engineer in accordance with this clause. Deductions shall commence in the next Interim Payment Certificate following that in which the total of all interim payments certified to the Contractor has reached the percentage of the Contract Price stipulated in the SCC less Provisional Sums, if any, and shall be made at the rate stated in the SCC of the amount of all Interim Payment Certificates until such time as the advance payment has been repaid; always provided that the advance payment shall be completely repaid prior to the time when 80 percent of the Contract Price has been certified for payment.

15. Claims, provisional sum and Settlement of disputes

15.1. Claims

15.1.1. Notice of Claims

Notwithstanding any other provision of the Contract, if the Contractor intends to claim any additional payment pursuant to any Clause of these Conditions or otherwise, he shall give notice of his intention to the Engineer, with a copy to the Employer, within 28 days after the event giving rise to the claim has first arisen.

15.1.2. Contemporary Records

Upon the happening of the event referred to in Sub-Clause 15.1.1, the Contractor shall keep such contemporary records as may reasonably be necessary to support any claim he may subsequently wish to make. Without necessarily admitting the Employer's liability, the Engineer shall, on receipt of a notice under Sub-Clause 15.1.1, inspect such contemporary records and may instruct the Contractor to keep any further contemporary records as are reasonable and may be material to the claim of which notice has been given. The Contractor shall permit the Engineer to inspect all records kept pursuant to this Sub-Clause and shall supply him with copies thereof as and when the Engineer so instructs.

15.1.3. Substantiation of Claims

Within 28 days, or such other reasonable time as may be agreed by the Engineer, of giving notice under Sub-Clause 15.1.1, the Contractor shall send to the Engineer an account giving detailed particulars of the amount claimed and the grounds upon which the claim is based. Where the event giving rise to the claim has a continuing effect, such account shall be considered to be an interim account and the Contractor shall, at such intervals as the Engineer may reasonably require, send further interim accounts giving the accumulated amount of the claim and any further grounds upon which it is based. In cases where interim accounts are sent to the Engineer, the Contractor shall send a final account within 28 days of the end of the effects resulting from the event. The Contractor shall, if required by the Engineer so to do, copy to the Employer all accounts sent to the Engineer pursuant to this Sub-Clause.

15.1.4. Failure to Comply

If the Contractor fails to comply with any of the provisions of this Clause in respect of any claim which he seeks to make, his entitlement to payment in respect thereof shall not exceed such amount as the Engineer or any arbitrator or arbitrators appointed pursuant to Sub-Clause 15.3.3 assessing the claim considers to be verified by contemporary records (whether or not such records were brought to the Engineer's notice as required under Sub-Clauses 15.1.2 and 15.1.3).

15.1.5. Payment of Claims

The Contractor shall be entitled to have included in any interim payment certified by the Engineer pursuant to Clause 14 such amount in respect of any claim as the

Engineer, after due consultation with the Employer and the Contractor, may consider due to the Contractor provided that the Contractor has supplied sufficient particulars to enable the Engineer to determine the amount due. If such particulars are insufficient to substantiate the whole of the claim, the Contractor shall be entitled to payment in respect of such part of the claim as such particulars may

substantiate to the satisfaction of the Engineer. The Engineer shall notify the Contractor of any determination made under this Sub-Clause, with a copy to the Employer.

15.2. Provisional Sums

15.2.1. Definition of "Provisional Sum"

"Provisional Sum" means a sum included in the Contract and so designated in the Bill of Quantities for the execution of any part of the Works or for the supply of goods, materials, Plant or services, or for contingencies, which sum may be used, in whole or in part, or not at all, on the instructions of the Engineer. The Contractor shall be entitled to only such amounts in respect of the work, supply or contingencies to which such Provisional Sums relate as the Engineer shall determine in accordance with this Clause. The Engineer shall notify the Contractor of any determination made under this Sub-Clause, with a copy to the Employer.

15.2.2. Use of Provisional Sums

In respect of every Provisional Sum the Engineer shall have authority to issue instructions for the execution of work or for the supply of goods, materials, Plant, Labour or services by:

- (a) the Contractor, in which case the Contractor shall be entitled to an amount equal to the value thereof determined in accordance with Clause 12.3, 12.4 and 12.5, and
- (b) a nominated Subcontractor, as hereinafter defined, in which case the sum to be paid to the Contractor therefor shall be determined and paid in accordance with Sub-Clause 5.4.4.

15.2.3. Production of Vouchers

The Contractor shall produce to the Engineer all quotations, invoices, vouchers and accounts or receipts in connection with expenditure in respect of Provisional Sums, except where work is valued in accordance with rates or prices set out in the Tender

15.3. Settlement of Disputes

15.3.1. Engineer's Decision

If a dispute of any kind whatsoever arises between the Employer and the Contractor in connection with, or arising out of, the Contract or the execution of the Works, whether during the execution of the Works or after their completion and whether before or after repudiation or expulsion or other termination of the Contract, including any dispute as to any opinion, instruction, determination, certificate or valuation of the Engineer, the matter in dispute shall, in the first

place, be referred in writing to the Engineer, with a copy to the other party. Such reference shall state that it is made pursuant to this Clause. No later than the 30th day after the day on which he received such reference the Engineer shall give notice of his decision to the Employer and the Contractor. Such decision shall state that it is made pursuant to this Clause.

Unless the Contract has already been repudiated or terminated, the Contractor shall, in every case, continue to proceed with the Works with all due diligence and the Contractor and the Employer shall give effect forthwith to every such decision of the Engineer unless and until the same shall be revised, as hereinafter provided, in an amicable settlement or an arbitral award.

If either the Employer or the Contractor be dissatisfied with any decision of the Engineer, or if the Engineer fails to give notice of his decision on or before the 30th day after the day on which he received the reference, then either the Employer or the Contractor may, on or before the seventieth day after the day on which he received notice of such decision, or on or before the seventieth day after the day on which the said period of 30 days expired, as the case may be, give notice to the other party, with a copy for information to the Engineer, of his intention to commence arbitration, as hereinafter provided, as to the matter in dispute. Such notice shall establish the entitlement of the party giving the same to commence arbitration, as hereinafter provided, as to such dispute and, subject to Sub-Clause 15.3.4, no arbitration in respect thereof may be commenced unless such notice is given.

If the Engineer has given notice of his decision as to a matter in dispute to the Employer and the Contractor and no notice of intention to commence arbitration as to such dispute has been given by either the Employer or the Contractor on or before the day specified here above, the said decision shall become final and binding upon the Employer and the Contractor.

15.3.2. Amicable Settlement

Where notice of intention to commence arbitration as to a dispute has been given in accordance with Sub-Clause 15.3.1, the parties shall attempt to settle such dispute amicably before the commencement of arbitration. Provided that, unless the parties otherwise agree, arbitration may be commenced on or after the 28th day after the day on which notice of intention to commence arbitration of such dispute was given, even if no attempt at amicable settlement thereof has been made.

15.3.3. Arbitration

Any dispute in respect of which:

- (a) the decision, if any, of the Engineer has not become final and binding pursuant to Sub-Clause 15.3.1, and
- (b) amicable settlement has not been reached within the period stated in Sub-Clause 15.3.2, shall be referred for arbitration in the manner provided as under and to the sole arbitrator appointed as follows:
 - (i) Either of the parties may give to the other notice in writing of the existence of such question of dispute or difference with a copy to the Engineer within the time specified in clause 15.3.1.
 - (ii) Within twenty-eight (28) days of receipts of such notice from either party the Engineer in Consultation with the Employer shall send to the contractor a panel of three persons and the contractor within twenty-one (21) days of receipt of such panel communicate to the Engineer and Employer the name of one of the persons from such panel and such person shall then be appointed a sole arbitrator by the Employer. However, the arbitrator so appointed shall not be an officer or the employee of the Employer or Engineer.
 - (iii) Provided that if the contractor fails to communicate the selection of a name out of the panel so forwarded to him within the specified time, Employer shall without delay select one person from the aforesaid panel and appoint him as the sole arbitrator.
 - (iv) The arbitrator to whom the matter is originally referred being transferred or vacating his office or being unable to act for any reason, then the Employer shall appoint another person to act as sole arbitrator, such person shall be entitled to proceed with the reference from the stage at which the predecessor left it.
 - (v) The award of the arbitrator shall be final and binding. The arbitrator shall decide in what proportion the arbitrator's fees, as well as the cost of Arbitration proceeding shall be borne equally by either party.
 - (vi) The arbitrator with the consent of the parties can enlarge the time, from time to time to make and publish his award.
 - (vii) A notice of the existence in question dispute or difference in connection with the contract unless served by either party within 30 days after the issue of Final Completion Certificate, failing which all rights and claim under this contract shall be deemed to have been waived and thus forfeited and absolutely barred.
 - (viii) Where the amount of claim is Rs.1,00,000 (INR One lakh only) and above the arbitrator shall give reasons for the award for each item of INR 75,000 (INR Seventy-Five Thousand only) and more.

- (ix) The work under this contract shall continue during arbitration proceedings and no valid payments due from or payment by the Employer shall be withheld on account of such proceedings except to the extent, which may be in dispute.
- (x) The arbitration and conciliation act 1996 together with any statutory modifications or re-enactment thereof and the rules made thereunder for the time being enforce shall apply to the arbitration proceeding under this clause.

The Arbitrator shall have the power to enlarge the term to rate the award with the consent of the parties provided always that the commencement or continuation of the arbitration proceeding shall not result in cessation or suspension of any of other rights and obligations of the parties of any payments due to them hereunder.

The venue of the arbitration proceedings shall be at Delhi/NCR. It is further clarified that both the parties to this agreement hereby undertake not to have recourse to civil court to solve any of their dispute whatsoever, arising out of this agreement except through arbitration.

15.3.4. Laws governing the Contract

- i) The laws of India shall govern this contract.
- ii) Irrespective of the place of Works, the place of performance or place of payment under the contract, the contract shall be deemed to have been made at the place from which the Letter of Award has been issued.
- iii) Courts of the place from where the Letter of Award of tender has been issued shall alone have jurisdiction to decide any dispute arising out of or in respect of contract. Notwithstanding anything contained in this Agreement, Parties hereby submit to the exclusive jurisdiction of such court alone.

15.3.5. Failure to Comply with Engineer's Decision

Where neither the Employer nor the Contractor has given notice of intention to commence arbitration of a dispute within the period stated in Sub-Clause 15.3.1 and the related decision has become final and binding, either party may, if the other party fails to comply with such decision, and without prejudice to any other rights it may have, refer the failure to arbitration in accordance with Sub-Clause 15.3.2. The provisions of Sub-Clauses 15.3.1 and 15.3.2 shall not apply to any such reference.

16. Termination by the Employer

16.1. Default of Contractor

If the Contractor is deemed by law unable to pay his debts as they fall due, or enters into voluntary or involuntary bankruptcy, liquidation or dissolution (other than a voluntary liquidation for the purposes of amalgamation or reconstruction), or becomes insolvent, or makes an arrangement with, or assignment in favour of, his creditors, or agrees to carry out the Contract under a committee of inspection of his creditors, or if a receiver, administrator, trustee or liquidator is appointed over any substantial part of his assets, or if, under any law or regulation relating to reorganization, arrangement or readjustment of debts, proceedings are commenced against the Contractor or resolutions passed in connection with dissolution or liquidation or if any steps are taken to enforce any security interest over a substantial part of the assets of the Contractor, or if any act is done or event occurs with respect to the Contractor or his assets which, under any applicable law has a substantially similar effect to any of the foregoing acts or events, or if the Contractor has contravened Sub-Clause 5.1, or has an execution levied on his goods, or if the Engineer certifies to the Employer, with a copy to the Contractor, that, in his opinion, the Contractor:

- (a) has repudiated the Contract,
- (b) without reasonable excuse has failed
 - (i) to commence the Works in accordance with Sub-Clause 8.1 or
 - (ii) to proceed with the Works, or any Section thereof, within 28 days after receiving notice pursuant to Sub-Clause 8.11,
- (c) has failed to comply with a notice issued pursuant to Sub-Clause 7.9 or an instruction issued pursuant to Sub-Clause 7.13 within 28 days after having received it,
- (d) despite previous warning from the Engineer, in writing, is otherwise persistently or flagrantly neglecting to comply with any of his obligations under the Contract, or
- (e) has contravened Sub-Clause 5.2,

then the Employer may, after giving 14 days" notice to the Contractor, enter upon the Site and expel the Contractor there from without thereby violating the Contract, or releasing the Contractor from any of his obligations or liabilities under the Contract, or affecting the rights and powers conferred on the Employer or the Engineer by the Contract, and may himself complete the Works or may employ any other contractor to complete the Works. The Employer or such other contractor may use for such completion so much of the Contractor's Equipment, Plant, Temporary Works, and materials, which have been deemed to be reserved exclusively for the execution of the Works, under the provisions of the Contract, as he or they may think proper, and the Employer may, at any time, sell any of the

said Contractor's Equipment, Temporary Works, and unused Plant and materials, and apply the proceeds of sale in or toward the satisfaction of any sums due or that may become due to him from the Contractor under the Contract.

16.2. Valuation at Date of Expulsion

The Engineer shall, as soon as may be practicable after any such entry and expulsion by the Employer, fix and determine ex parte, or by or after reference to the parties or after such investigation or enquiries as he may think fit to make or institute, and shall certify:

- a) what amount (if any) had, at the time of such entry and expulsion, been reasonably earned by or would reasonably accrue to the Contractor in respect of work then actually done by him under the Contract, and
- b) the value of any of the said unused or partially used materials, any Contractor's Equipment, and any Temporary Works.

16.3. Payment after Expulsion

If the Employer shall enter and expel the Contractor under this Clause, the Earnest Money Deposit, Security Deposit already recovered and Performance Guarantee under the contract shall be liable to be forfeited and shall be absolutely at the disposal of the IWAI. Further, the Employer shall enter and expel the Contractor under this Clause, he shall not be liable to pay to the Contractor any further amount (including damages) in respect of the Contract until the expiration of the Maintenance Period and thereafter until the costs of execution, completion and remedying of any defects, damages for delay in completion (if any) and all other expenses incurred by the Employer have been ascertained and the amount thereof certified by the Engineer. The Contractor shall then be entitled to receive only such sum (if any) as the Engineer may certify would have been payable to him upon due completion by him after deducting the said amount. If such amount exceeds the sum which would have been payable to the Contractor on due completion by him, then the Contractor shall, upon demand, pay to the Employer the amount of such excess and it shall be deemed a debt due by the Contractor to the Employer and shall be recoverable accordingly.

16.4. Assignment of Benefit of Agreement

Unless prohibited by law, the Contractor shall, if so instructed by the Engineer within 14 days of such entry and expulsion referred to in Sub-Clause 16.1, assign to the Employer the benefit of any agreement for the supply of any goods materials or services and/or for the execution of any work for the purposes of the Contract, which the Contractor may have entered into.

16.5. Corrupt or Fraudulent Practices

If in the judgment of the Employer the Contractor has engaged in corrupt or fraudulent practices, in competing for or in executing the Contract, then the Employer may, after having given 14 days" notice to the Contractor, terminate the Contractor's employment under the Contract and expel him from the Site, and the provisions of Clause 16 shall apply as if such expulsion had been made under Sub-Clause 16.1.

"Corrupt practice" means the offering, giving, receiving, or soliciting of anything of value to influence the action of a public official, Employer, Engineer or their representatives in the procurement process or in Contract execution.

"Fraudulent practice" means a misrepresentation of facts in order to influence a procurement process or the execution of a Contract to the detriment of the Borrower and includes collusive practice among bidders (prior to or after bid submission) designed to establish bid prices at artificial non-competitive levels and to deprive the Borrower of the benefits of free and open competition.

16.6. Urgent Remedial Work

If, by reason of any accident, or failure, or other event occurring to, in, or in connection with the Works, or any part thereof, either during the execution of the Works, or during the Maintenance Period, any remedial or other work is, in the opinion of the Engineer, urgently necessary for the safety of the Works and the Contractor is unable or unwilling at once to do such work, the Employer shall be entitled to employ and pay other persons to carry out such work as the Engineer may consider necessary. If the work or repair so done by the Employer is work which, in the opinion of the Engineer, the Contractor was liable to do at his own cost under the Contract, then all costs consequent thereon or incidental thereto shall, after due consultation with the Employer and the Contractor, be determined by the Engineer and shall be recoverable from the Contractor by the Employer, and may be deducted by the Employer from any monies due or to become due to the Contractor and the Engineer shall notify the Contractor accordingly, with a copy to the Employer. Provided that the Engineer shall, as soon after the occurrence of any such emergency as may be reasonably practicable, notify the Contractor thereof

17. Termination by Contractor

17.1. Default of Employer

In the event of the Employer:

(a) Failing to pay to the Contractor the amount due under any certificate of the Engineer within 28 days after the expiry of the time stated in Sub-Clause 14.12, within which payment is to be made, subject to any deduction that the Employer is entitled to make under the Contract,

- (b) Interfering with or obstructing or refusing any required approval to the issue of any such certificate,
- (c) becoming bankrupt or being a company, going into liquidation, other than for the purpose of a scheme of reconstruction or amalgamation, or
- (d) giving notice to the Contractor that for unforeseen economic reasons it is impossible for him to continue to meet his contractual obligations, the Contractor shall be entitled to terminate his employment under the Contract by giving notice to the Employer, with a copy to the Engineer. Such termination shall take effect 14 days after the giving of the notice.

17.2. Removal of Contractor's Equipment

Upon the expiry of the 14 days' notice referred to in Sub-Clause 16.1, the Contractor shall, notwithstanding the provisions of Sub-Clause 4.36.1, with all reasonable dispatch, remove from the Site all Contractor's Equipment brought by him thereon.

17.3. Payment on Termination

In the event of such termination, the Employer shall be under the same obligations to the Contractor in regard to payment as if the Contract had been terminated under the provisions of Clause 13.

17.4. Contractor's Entitlement to Suspend Work

Without prejudice to the Contractor's entitlement to interest under Sub-Clause 14.12 and to terminate under Sub-Clause 16.1, the Contractor may, if the Employer fails to pay the Contractor the amount due under any certificate of the Engineer within 28 days after the expiry of the time stated in Sub-Clause 14.12 within which payment is to be made, subject to any deduction that the Employer is entitled to make under the Contract, after giving 28 days" prior notice to the Employer, with a copy to the Engineer, suspend work or reduce the rate of work. If the Contractor suspends work or reduces the rate of work in accordance with the provisions of this Sub-Clause and thereby suffers delay or incurs costs the Engineer shall, after due consultation with the Employer and the Contractor, determine:

- (a) Any extension of time to which the Contractor is entitled under sub clause 8.7, 8.7 and 8.8 of clause 8, and
- (b) The amount of such costs, which shall be added to the Contract Price, and shall notify the Contractor accordingly, with a copy to the Employer.

17.5. Resumption of Work

Where the Contractor suspends work or reduces the rate of work. having given notice in accordance with Sub-Clause 17.4, and the Employer subsequently pays the amount due, including interest pursuant to Sub-Clause 14.12, the Contractor's entitlement under Sub-Clause 17.1 shall, if notice of termination has not been given, lapse and the Contractor shall resume normal working as soon as is reasonably possible.

Whenever the Works is resumed after suspension, the Engineer and the Contractor shall jointly examine the Works and the Plant and Materials affected by the suspension and the Contractor shall prepare a report indicating the physical status and major adverse effects, if any, the Works under suspension has suffered during the period of suspension and furnish same to the Engineer and Employer.

18. Design and drawings

18.1. Custody and Supply of Drawings and Documents

The approved drawings shall remain in the custody of the Engineer, but two copies thereof shall be provided to the Contractor free of charge. The Contractor shall make at his own cost any further copies required by him. Unless it is strictly necessary for the purposes of the Contract, the Drawings, Specification and other documents provided by the Employer or the Engineer shall not, without the consent of the Engineer, be used or communicated to a third party by the Contractor. Upon issue of the Final completion Certificate, the Contractor shall return to the Engineer all Drawings, Specification and other documents provided under the Contract.

18.2. One Copy of Drawings to be kept on Site

One copy of the Drawings, provided to the Contractor as aforesaid, shall be kept by the Contractor on the Site and the same shall at all reasonable times be available for inspection and use by the Engineer and by any other person authorized by the Engineer in writing.

18.3. Disruption of Progress

The Contractor shall give notice to the Engineer, with a copy to the Employer, whenever planning or execution of the Works is likely to be delayed or disrupted unless any further drawing or instruction is issued by the Engineer within a reasonable time. The notice shall include details of the drawing or instruction required and of why and by when it is required and of any delay or disruption likely to be suffered if it is late. This shall not be applicable in the case of permanent works to be designed and engineered by the contractor, except with regard to its approval by the Engineer, if specified.

18.4. Delays and Cost of Delay of Drawings

If, by reason of any failure or inability of the Engineer to issue, within a time reasonable in all the circumstances, any drawing or instruction for which notice has been given by the Contractor in accordance with Sub Clause 18.3, the Contractor suffers delay and/or incurs costs then the Engineer shall, after due consultation with the Employer and the Contractor, determine:

(a) any extension of time to which the Contractor is entitled under sub clause 8.6, 8.7 and 8.8, and

18.5. Supplementary Drawings and Instructions

The Engineer shall have authority to issue to the Contractor, from time to time, such supplementary Drawings and instructions as shall be necessary for the purpose of the proper and adequate execution and completion of the Works and the remedying of any defects therein. The Contractor shall carry out and be bound by the same.

19. Insurance

All persons deployed for the work shall be insured by the Contractor at his cost and

documentary evidence should be provided before commencement of the work. IWAI shall not be responsible in any manner for any accident to the personnel engaged by the Contractor during the work or otherwise

19.1. Insurance of Contractor's Equipment, Plant and materials.

The Contractor shall, without limiting his or the Employer's obligations and responsibilities under Clause 2.2, 4.16, 4.17 and 4.18, insure:

- (a) the Materials and Plant/Equipment's for incorporation therein in the Works, to their full replacement cost at Site. Such insurance shall provide for compensation to be payable in the types and proportions of currencies required to rectify the loss or damage incurred,
- (b) an additional sum of 10 per cent of such replacement cost, to cover any additional costs of and incidental to the rectification of loss or damage including professional fees and the cost of demolishing and removing any part of the Works and of removing debris of whatsoever nature. Such insurance shall provide for compensation to be payable in the types and proportions of currencies required to rectify the loss or damage incurred and
- (c) the Contractor's Equipment and other things brought onto the Site by the Contractor, for a, sum sufficient to provide for their replacement at the Site.

19.2. Scope of Cover

The insurance in paragraphs (a) and (b) of Sub-Clause 19.1 shall be in the joint names of the Contractor and the Employer and shall cover:

- (a) the Employer and the Contractor against all loss or damage from whatsoever cause arising, other than as provided in Sub-Clause 19.4, from the first working day after the Commencement Date until the date of issue of the relevant Completion certificate in respect of the Works or any Section or part thereof as the case may be, and
- (b) the Contractor for his liability:
 - (i) during the Maintenance Period for loss or damage arising from a cause occurring (ii) prior to the commencement of the Maintenance Period, and
 - (ii) for loss or damage occasioned by the Contractor in the course of any operations carried out by him for the purpose of complying with his obligations under Clause 11.
- (c) It shall be the responsibility of the Contractor to notify the insurance company of any change in the nature and extent of the Works and to ensure the adequacy of the insurance coverage at all times during the period of the Contract.

19.3. Responsibility for Amounts not recovered

Any amounts not insured or not recovered from the insurers shall be borne by the Employer or the Contractor in accordance with their responsibilities under Clause 2.2, 4.16, 4.17 and 4.18.

19.4. Exclusions

There shall be no obligation for the insurances in Sub-Clause 19.1 to include loss or damage caused by risks listed under sub clause 2.2.

19.5. Damage to Persons and Property

The Contractor shall, except if and so far as the Contract provides otherwise, indemnify the Employer against all losses and claims in respect of:

- (a) death of or injury to any person, or
- (b) loss of or damage to any property (other than the Works), which may arise out of or in consequence of the execution and completion of the Works and the remedying of any defects therein, and against all claims, proceedings, damages, costs, charges and expenses whatsoever in respect thereof or in relation thereto, subject to the exceptions defined in Sub-Clause 19.6.

19.6. Exceptions

The "exceptions" referred to in Sub-Clause 19.5 are:

- (a) the permanent use or occupation of land by the Works, or any part thereof,
- (b) the right of the Employer to execute the Works, or any part thereof, on, over, under, in or through any land,
- (c) damage to property which is the unavoidable result of the execution and completion of the Works, or the remedying of any defects therein, in accordance with the Contract, and
- (d) death of or injury to persons or loss of or damage to property resulting from any act or neglect of the Employer, his agents, servants or other contractors, not being employed by the Contractor, or in respect of any claims, proceedings, damages, costs, charges and expenses in respect thereof or in relation thereto or, where the injury or damage was contributed to by the Contractor, his servants or agents, such part of the said injury or damage as may be just and equitable having regard to the extent of the responsibility of the Employer, his servants or agents or other contractors for the injury or damage.

19.7. Indemnity by Employer

The Employer shall indemnify the Contractor against all claims, proceedings, damages, costs, charges and expenses in respect of the matters referred to in the exceptions defined in Sub-Clause 19.6.

19.8. Third Party Insurance (including Employer's Property)

The Contractor shall, without limiting his or the Employer's obligations and responsibilities under Clause 19.5, 19.6 and 19.7, insure, in the joint names of the Contractor and the Employer, against liabilities for death of or injury to any person (other than as provided in Clause 19.11 and 19.12) or loss of or damage to any property (other than the Works) arising out of the performance of the Contract, other than the exceptions defined in paragraphs (a), (b) and (c) of Sub-Clause 19.6.

19.9. Minimum Amount of insurance

Such insurance shall be for at least the amount stated in the SCC.

19.10. Cross Liabilities

The insurance policy shall include a cross-liability clause such that the insurance shall apply to the Contractor and to the Employer as separate insured.

19.11. Accident or Injury to Workmen

The Employer shall not be liable for in respect of any damages or compensation payable to any workman or other person in the employment of the Contractor or any Subcontractor, other than death or injury resulting from any act or default of the Employer, his agents or servants. The Contractor shall indemnify and keep indemnified

Employer against all such damages and compensation, other than those for which the Employer is liable as aforesaid, and against all claims, proceedings, damages, costs, charges, and expenses whatsoever in respect thereof or in relation thereto.

19.12. Insurance Against Accident to Workmen

The Contractor shall insure against such liability and shall continue such insurance during the whole of the currency of Contract in respect of all and any persons as are employed by him on the Works. Provided that, in respect of any persons employed by any Subcontractor, the Contractor's obligations to insure as aforesaid under this Sub- Clause shall be satisfied if the Subcontractor shall have insured against the liability in respect of such persons in such manner that the Employer is indemnified under the policy, but the Contractor shall require such Subcontractor to produce to the Employer, when required, such policy of insurance and the receipt for the payment of the current premium.

19.13. Evidence and Terms of Insurances

The Contractor shall provide evidence to the Employer as soon as practicable after the respective insurances have been taken out but, in any case, prior to the start of work at the Site that the insurances required under the Contract have been effected and shall, within 84 days of the Commencement Date, provide the insurance policies to the Employer. When providing such evidence and such policies to the Employer, the Contractor shall notify the Engineer of so doing. Such insurance policies shall be consistent with the general terms agreed prior to the issue of the Letter of Award. The Contractor shall affect all insurances for which he is responsible with insurers and in terms approved by the Employer.

19.14. Adequacy of Insurances

The Contractor shall notify the insurers of changes in the nature, extent or programme for the execution of the Works and ensure the adequacy of the insurances at all times in accordance with the terms of the Contract and shall, when required, produce to the Employer the insurance policies in force and the receipts for payment of the current premiums.

19.15. Remedy on Contractor's Failure to Insure

If the Contractor fails to effect and keep in force any of the insurances required under the Contract, or fails to provide the policies to the Employer within the period required by Sub-Clause 19.13, then and in any such case the Employer may

effect and keep in force any such insurances and pay any premium as may be necessary for that purpose and from time to time deduct the amount so paid from any monies due or to become due to the Contractor, or recover the same as a debt due from the Contractor.

19.16. Compliance with Policy Conditions

In the event that the Contractor or the Employer fails to comply with conditions imposed by the insurance policies effected pursuant to the Contract, each shall indemnify the other against all losses and claims arising from such failure.

19.17. Source of Insurance

The Contractor shall place all insurance relating to the Contract (including, but not limited to, the insurance referred to in Clauses 19) with insurers from India.

20. Release From Performance

20.1. Payment in Event of Release from Performance

If any circumstance outside the control of both parties arises after the issue of the Letter of Award which renders it impossible or unlawful for either or both parties to fulfil his or their contractual obligations, or under the law governing the Contract the parties are released from further performance, then the parties shall be discharged from the Contract, except as to their rights under this Clause and Clause 15.3 and without prejudice to the rights of either party in respect of any antecedent breach of the Contract, and the sum payable by the Employer to the Contractor in respect of the work executed shall be the same as that which would have been payable under Clause 13 if the Contract had been terminated under the provisions of Clause 13.

21. Notices

21.1. Notice to Contractor

All certificates, notices or instructions to be given to the Contractor by the Employer or the Engineer under the terms of the Contract shall be sent by post, cable, e-mail or facsimile transmission to or left at the Contractor's principal place of business or such other address as the Contractor shall nominate for that purpose.

21.2. Notice to Employer and Engineer

Any notice to be given to the Employer or to the Engineer under the terms of the Contract shall be sent by post, cable, e-mail or facsimile transmission to or left at the respective addresses nominated for that purpose.

21.3. Change of Address

Either party may change a nominated address to another address in the country where the Works are being executed or in the country of Employer by prior notice to the other party, with a copy to the Engineer, and the Engineer may do so by prior notice to both parties.

22. Changes in Cost and Legislation

22.1. Increase or Decrease of Cost

There shall be no addition or deduction from the Contract Price any sums in respect of rise or fall in the cost of labour and/or materials or any other matters affecting the cost of the execution of the Works as far as the items of Works covered are concerned. However, addition or deduction from the Contract Price any sums in respect of rise or fall in the cost of POL may be determined as defined in SCC.

22.2. Subsequent Legislation

If, after the date 28 days prior to the latest date for submission of tenders for the Contract there occur in the country in which the Works are being or are to be executed changes to any National or State Statute, Ordinance, Decree or other Law or any regulation or bye-law of any local or other duly constituted authority, or the introduction of any such State Statute, Ordinance, Decree, Law, regulation or bye-law which causes additional or reduced cost to the Contractor, other than under Sub- Clause 22.1, in the execution of the Contract, such additional or reduced cost shall, after due consultation with the Employer and the Contractor, be determined by the Engineer and shall be added to or deducted from the Contract Price-and the Engineer shall notify the Contractor accordingly, with a copy to the Employer.

23. Sustainability of JV

In case JV/consortium is not sustained due any reasons or discrepancies arising amongst the JV/consortium members, unsustainability of JV/consortium shall be determined by the Engineer in Charge/Employer during the monitoring in the contract period based on the failure of the deliveries/missing of the milestones and other deliverables. The same shall be dealt in the following manner:

- (a) If JV/consortium becomes unsustainable after being selected as L1 EMD of the defaulting JV/consortium shall be forfeited.
- (b) If JV/consortium becomes unsustainable after the award of contract, the employer has complete authority to nominate another contractor to complete the balance work on risk and cost of the defaulting contractor. The Contractor will be notified of the decision to stop the work and get the measurements done to ascertain the value of work completed till date. No payment shall be released immediately, however, after completion of entire

work on risk and cost, the differential cost would be recovered from all held up amount (BG, EMD, Security deposit and unpaid amount for the work done) of the defaulting Contractor and if any balance amount is still available, that shall be released to the defaulting Contractor.

24. Independent Relationship

Notwithstanding anything contained in this Agreement, the relationship between the Employer and the Contractor shall always remain on a principal-to-principal basis and the Contractor shall never declare / claim himself to be an agent of the Employer. The Contractor shall alone be responsible for all its action / omission of the Contractor shall bind the Employer in any manner unless expressly so provided in this Agreement.

SECTION - VIII: SPECIAL CONDITIONS OF CONTRACT (SCC)
(BCC)
199

1. General

- 1.1 Special Conditions shall be read in conjunction with the General Conditions of Contract, specification, ToR, Drawings and any other documents forming part of this contract wherever the context so requires.
- 1.2 Notwithstanding the Sub-division of the documents in to these separate section and volume every part of each shall be deemed to be supplementary to and complementary of every other part and shall be read with and as a part of the Contract.
- 1.3 Where any portion of the general Conditions of Contract is repugnant to or at variance with any provisions of the Special Conditions, the provisions of the Special Conditions shall be deemed to over-ride the provisions of the General Conditions of Contract.
- 1.4 Where it is mentioned in the Specification that the contractor shall perform certain work or provide certain facilities, it is understood that the contractor shall do so at his own cost.
- 1.5 The materials and workmanship shall satisfy the relevant standards as stipulated under the Contract Agreement, the specification conditioned herein and codes referred to. Where the specifications stipulate requirement in addition to those contained in the Standard codes and specifications these additional requirements shall also be satisfied.

2. The Site

2.1. General Site Information

The Contractor shall visit the site and satisfy themselves on the actual site conditions, meteorological and other data, before tendering. Whatever information and, other relevant data given in the tender documents are only intended as a general guidance for the contractor and no warranty is given for the correctness of the same.

2.2. Condition of Site

The Contractor, before carrying out any work, shall inspect the Site in conjunction with the Engineer's Representative to establish its general condition which shall be agreed and recorded in writing, and where, in the opinion of the Engineer or his Representative it is deemed necessary, by means of photography.

The boundary of the land earmarked for the execution of the Works shall be defined by the Engineer's Representative and the Contractor shall provide, erect, and maintain from commencement to final completion of all work and all reinstatement substantial galvanized pipes or similar approved markers indicating the boundary of the working area at regular intervals as the Engineer's

Representative may require. In the event of any such boundary marker / post being disturbed or displaced as a result of the Contractor's operations or theft, the Contractor shall forthwith at his own expense replace the same and maintain the boundary properly.

3. Order of Works

The order in which the works are to be carried out by the Contractor shall be to the approval of Employer and shall be such as to suit the detailed method of construction, adopted by the Contractor as well as the Work Plan / Overall Schedule approved by the Employer. The works shall be carried out in such a manner so as to enable the other Contractors to work concurrently so that the entire project may be brought into use immediately after the completion of works.

4. General Conditions of Construction and Erection Work

- 4.1 The work shall be permitted on the site round the clock.
- 4.2 The contractor must arrange for the placement of workers in such a way that the delayed completion of the works of any part thereof for any reason whatsoever will not affect their proper employment. The Employer will not entertain any claim for over time /idle time payment etc. whatsoever.
- 4.3 The Contractor shall submit to Employer / Engineer-in-Charge reports at regular intervals regarding the state and progress of work. The details and proforma of the report will mutually be agreed after the award of Contract.

5. Surveys and Levels

- Before the works of any part thereof begin, the Contractor shall survey and take levels of the site of the works both above and below water level / Chart Datum level and agree all particulars on which the measurements of the works are to be based. Such particulars shall be recorded and after agreement; the drawings shall be signed by the Employer / Engineer-In-Charge and the Contractor.
- 5.2 Failing such surveys and agreements being prepared and/or signed by the Contractor, the survey carried out by the Engineer shall be final and binding on the Contractor.

6. Setting out the Works

6.1 The Employer shall furnish the relevant existing grid points with bench Mark with reference to permanent bench marks. It shall be Contractor's responsibility to the set out the necessary central points on land and to set out alignment. The Contractor shall have in his employees, an efficient survey team for this purpose and the accuracy of such setting out works shall be Contractor's sole responsibility.

- 6.2 Before beginning the work, the Contractor shall work out the control points on ground which, are pre requisite for carrying out hydrographic surveys, accurately, with suitable markers as approved by Employer. All these points and markings shall be checked and approved by Employer's Representative before starting the work.
- 6.3 The Contractor shall also provide necessary equipment, labour and other facilities for proper checking and inspection of the fabrication, welding, painting and any other such activity forming part of the total scope of the work at no cost to the Employer.
- 6.4 The Contractor shall give Employer not less than 24 hours" notice in writing of his intention to inspect for any part of the works so that arrangements may be made for checking the work.
- 6.5 The Contractor shall at his own expense provide all assistance which Employer or his representative may require for checking and inspection.

7. Conditions for work

- 7.1 The decision of EIC to accept the work done with or without the Liquidated damages shall be final and binding.
- 7.2 No idle time charges will be paid to the contractor on any account.
- 7.3 The Employer / Engineer-in-charge or his representative will inspect, co-ordinate, measure and certify the work. He has the right to inspect at any time during the contract period for which Contractor's authorized agent shall be available at site during such inspection.
- 7.4 The Contractor may execute the work round the clock and on all days including Sundays and Public Holidays except National Holidays subject to such restriction as may be imposed by State Govt./local body or Engineer In Charge.
- 7.5 If the Contractor's personnel or any outside labour employed to work during execution of Contract, breaks or damages/destroys any vessel, craft, net, fishing stakes etc. building, road, kerb, water pipe, fence, bund, wires, trees, crop, fruit or cultivated ground during the period of agreement, the same shall be made good by the Contractor at his own expenses or in default the IWAI may cause the same to be made good by other sources and deduct the expenses from any sums that may be then or at any time thereafter may become due to the contractor.
- 7.6 The rate quoted shall include all taxes, duties and any other levies. No additional payment on any such account shall be payable by IWAI. However, GST which will be quoted separately by the Bidder will be reimbursed to the Bidder on producing proof of payment

- 7.7 The right to award/split up work or to reject the offer without assigning any reason is reserved with the Authority.
- 7.8 If the work is stopped continuously for more than 3 days due to law and order, the Contractor has to report to EIC the same and take actions mutually in accordance with prevailing local laws for removal of such problem and proceeding further with the work to mutual benefit.
- 7.9 Any charge incurred for adopting environmental safe guards, minimizing detrimental impacts, enhancing the beneficial aspects of the project and for effective management of the environmental resources affected by the project shall be met by the Contractor. No additional charges on any account shall be payable by IWAI.

8. Standard and Standard Products

8.1. Standards

Save in so far otherwise specified in the Contract the relevant provisions of the appropriate Bureau of Indian Standards shall apply for all materials and workmanship. Where relevant BIS standards do not exist, the latest version of the relevant standards in vogue shall be applied. Where both above Standards may not be available, the following Standards shall also be referred to and adopted.

BS: British Standard Specification

CP: British Standard Code of Practice

ASTM: Standard of American Society for Testing Materials.

8.2. Standard Products

If mention is made in the Contract of named products of individual manufacturers, this is an indication of the standard type and workmanship of goods, which are satisfactory to the Engineer. The Contractor may substitute similar products of at least equal quality and suitability, subject to the approval of the Engineer.

Proposals by the Contractor for the supply of manufactured products shall be accompanied by certification that the manufacturer's products are approved by the relevant standards authority.

9. Security and Safety

- 9.1 The Contractor shall comply with all regulations imposed by the Central / State / Local Authorities and Employer in respect of the passage of Plant, Vehicles, materials and personnel through waterways.
- 9.2 The Contractor shall take all possible precautions to prevent out breaks of fire on the site and in all offices, stores, camps and other places and things connected

therewith and especially with respect to the safe storage of petroleum products, explosives and all other dangerous of hazardous goods. The Contractor shall comply with all rules, regulations and orders of any Statutory Authority and of Employer at no extra cost to the Employer.

- 9.3 The Contractor shall obtain from the Employer details of any restricted areas in or around the site and shall have prominently and clearly displayed for the information of his staff and work people notices defining any such restricted areas. Such notices shall be provided at his own expenses.
- 9.4 If required, the Contractor shall take entry passes to the restricted areas for all personnel labourers and vehicle. No claim whatsoever on this account will be entertained.

10. Health Safety and Environment, Risk & Hazard

- 10.1 The Contractor shall prepare and submit to the Employer / Employer-In-Charge the Health Safety and Environment (HSE) Plan in line with Employers HSE policy and Contractors company policy within 28 days of commencing of works for the approval of the Employer. The policy shall be followed during the life cycle of the project. The policy may be updated as and when required.
- 10.2 The Contractor shall prepare and submit the Risk and Hazard identification and mitigation plan within 28 days of commencing of works for approval of the Employer. The plan may be updated as and when required.

11. Arbitration

In addition to Clause 15.3.3, in case of contract with another Public Sector Undertaking, following Arbitration Clause shall apply: "Except as otherwise provided, in case of a contract with a public Sector Undertaking if at any time any question dispute or difference whatsoever arises between the parties upon or in relation to, or in connection with this agreement, the same shall be settled in terms of the Ministry of Industry, Department of Public Enterprises O.M No. 3/5/93-PMA dt.30.06.93 or any modifications / amendments thereof. "The arbitrator shall have the power to enlarge the term to publish the award with the consent of the parties provided always that the commencement or continuation of the arbitration proceeding shall not result in cessation or suspension of any of other rights and obligations of the parties of any payments due to them hereunder.

12. Taxes, duties, and levies etc.

The prices shall include all the taxes, levies, cess, octroi, royalty, terminal tax, excise, GST or any other local, State or Central taxes as applicable/ charged by Centre or State Government or Local authorities on all materials that the contractor has to purchase for the performance of the contract and services, shall

be payable by the contractor and the Authority will not entertain any claim for compensation whatsoever in this regard.

13. Additional Conditions

In addition to the above-mentioned SCC the following table shall also form part of SCC

GCC Clause	Heading	Definitions							
1.3	Language for Communication	English							
2.3	Right of access to site	-							
3.1.2.4 (ii)	Percentage of variation under Engineer's duty and Authority	20 % (twenty percent).							
4.4	Performance security and security deposit	Shall be released only after the completion of entire contract period.							
4.7 (b)	Unforeseeable Physical Obstructions or Conditions	No cost on any account due to unforeseeable physical obstructions or conditions shall be added to the Contract Price							
4.9.1	Program to Be Submitted	14 days							
4.9.3	Cash Flow Estimate	14 days and subsequently submit the revised cash flow estimate at mutually agreed interval							
4.37	Co-operation	The contractor shall provide office space, accommodation and vehicle for Employer, Employer's representatives, Engineer and Engineer's representatives for monitoring / checking / inspection as and when required.							
5.4.4 & 5.4.5	Payments to Nominated Subcontractors	The GCC clause 5.4.4 & 5.4.5 shall now be read as: Contractor shall be entirely responsible for all the payments to be made to the nominated subcontractor. Contractor shall hold employer indemnified against all claims arising from sub-contracting of works. The contractor shall produce No claim Certificate from the sub-contractors at the							
		demand of EIC, before final bill payment. Contractor shall not be absolved of his							

GCC Clause	Heading	Definitions
		responsibilities and duties of the contract by nominating sub-contractors.
8.5	Time for completion	6 months from the date of signing of Contract Agreement.
8.12	Limit of Liquidated Damages	Maximum 10 percent of the Contract Price.
10.2 &10.3	Taking over of sections or parts	There shall be no taking over of sections or parts. Only interim completion certificate and final completion certificate shall be given as per the conditions of contract.
15.2 (b)	Minimum Amount of Interim Payment (stagewise)	Interim Payment Certificates
14.13	Maximum Amount of Advance Payment	10 percent of the Contract Price at an interest rate of 10 percent per annum on an outstanding amount of the advance. Contractor shall submit a bank guarantee for an amount equivalent to 110% of the advance payment requested by the contractor. The Bank Guarantee so submitted shall be valid for entire contract period and it shall be returned to the contractor after the advance payment has been recovered. The Bank Guarantee shall be submitted as per annex - 1
14.13	Repayment of Advance Payment	In 4 subsequent running payments due after release of advance @ 25% of Advance amount + interest from each payment.
19.9	Minimum Amount of Third- Party Insurance	INR 50 lakhs per occurrence, with the number of occurrences unlimited.

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
SECTION-IX: A	ANNEXES
20	7
20	•

ANNEX - I: BANK GUARANTEE FORM FOR PERFORMANCE SECURITY

To

The Chairperson Inland waterways Authority of India Ministry of Shipping, Govt. of India A-13, Sector-1, Noida(U.P.) Pin- 201301

In consideration of the (hereinafter called "Employer") having to enter
into an Agreement with M/s (hereinafter called the "Contractor) as a follow
up to the Letter of Award nodated issued by the Employer for Design,
Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger
Ferry, on production of Performance security in the form of Bank Guarantee for INR
(Rupees only), at the request of Contractor, We, (Bank)
do hereby undertake to pay to the Employer an amount not exceeding
INR
the part of Contractor to perform the contract in accordance with terms & conditions or any
breach of the said Agreement.

- 2. We, (Bank) undertake to pay the Employer any money so demanded notwithstanding any dispute or disputes raised by the Contractor in any suit or proceeding pending before any court or Tribunal relating thereto, liability under this present being absolute and unequivocal. The payment so made by us under this guarantee shall be valid discharge of our liability for payment there under and the Contractor shall have no claim against us for making such payment.
- 3. We, (Bank) further agree that the guarantee herein contained shall remain in full force and effect till completion of project work to the complete satisfaction of the Employer in terms of conditions of contract and Letter of Award and that it shall continue to be enforceable till all the dues of the Employer under or by virtue of the said Agreement

have been fulfilled and its claim satisfied or till the scheduled date of completion of Works as per the Agreement. We(Bank) shall consider that the terms and conditions of the said Agreement have been fully and properly carried out by the said Contractor and accordingly discharge this Guarantee after 90 days from the date of completion of the said contract unless a demand or claim under this Guarantee is served by the Employer in writing on the bank but before the expiry of the said period in which case it shall be enforceable against the bank notwithstanding the fact that the same is enforced after the expiry of the said period or after the extended period as the case may be.

- 4. We (Bank) further agree with the Employer that the Employer shall have fullest liberty without our consent and without affecting in any manner our obligations hereunder to vary any of the terms and conditions of the said Agreement or to extend time or performance by the said Contractor from time to time or to postpone for any time or from time to time any of the powers exercisable by the Employer against the said Contractor and to forbear or enforce any of the terms and conditions relating to the said Agreement and we shall not be relieved from our liability by reason of any such variation, or extension being granted to the said Contractor or for any forbearance, act or omission on the part of the Employer or any indulgence by the Employer to the said Contractor or by any such matter or thing whatsoever which under the law relating to sureties would, but for the provision, have effect of so relieving us.
- 5. It shall not be necessary for the Employer to proceed against the **Contractor** before proceeding against the Bank and the guarantee herein contained shall be enforceable against the bank notwithstanding any security which the Employer may have obtained or obtain from the **Contractor** at the time when proceedings are taken against the bank hereunder be outstanding or unrealized.
- 7. This Guarantee will not be discharged due to the change in the constitution of the Bank or the **Contractor**.
- 8. We, **(Bank)** lastly undertake not to revoke this Guarantee during its currency except with the previous consent of the Employer in writing.

Dated the of 2024
for
(Indicate the name of bank)
Signature

Inland Waterways Authority of India (Ministry of Ports, Shipping & Waterways, Govt. of India)	Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
	Name of the Officer
	(In Block Capitals)
	Designation Code No
	Traine of the bank and Branen. (SETTE)
	210

ANNEX - II: AGREEMENT FORM

(To be executed on non-judicial stamp paper of Rs. 100 and signed by the Bidder and the same is to be signed by Authorized Signatory on behalf of IWAI)

Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry

AGREEMENT

BETWEEN

INLAND WATERWAYS AUTHORITY OF INDIA AND

CONTRACTING FIRM

This agreement made on this..... day of 20..... between Inland Waterways Authority of India, A-13, Sector-1, Noida (U.P.)-201301. (Hereinafter called the "IWAI" which

expression shall unless excluded by or repugnant to the context be deemed to include their
successor in office) on the one part and M/S having its office a
(Hereinafter called "Contractor "which expression shall unless excluded
by or repugnant to the context be deemed to include their heirs, executors, administrators
representatives and assigns or successor in office) on the other part.
WHEREAS IWAI is desirous of Design, Construction, Supply, Testing and
Commissioning of 50 Pax Hybrid Electric Passenger Ferry, as per the work Order No
of the agreement attachment hereto all of with form part if the agreement.
WHEREAS THE CONTRACTING FIRM has agreed to undertake the Design
Construction, supply, Testing and Commissioning of vessels listed under Design
Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger
Ferry on Terms and Conditions herein after set forth.
NOW THEREFORE THESE PRESENTS WITNESS and it is hereby agreed, declared by
and between the parties hereto as follows

The following documents shall be deemed to form and to be read and construed as part of the agreement i.e.

a) Notice inviting Tender

attachment hereto all of which form part of the agreement.

- b) Form of Tender
- c) Condition of contact

d)	Schedule of the price bid
e)	Agreement form
f)	Technical Bid No. with Date.
g)	Addendums/Corrigendum's
h)	Minutes of Pre-bid Meeting
i)	All Correspondence
j)	
k)	
hereun hereun	TNESS whereof the IWAI has caused Shri
Witnes	sses, IWAI
1)	
2)	
	his deed was duly executed by Shri for the actor above named in the presence of
Witnes	sses of Contractor
1)	
2)	

ANNEX -III: DETAILS OF BANK ACCOUNT

FOR RELEASE OF PAYMENT THROUGH

ELECTRONIC FUND TRANSFER SYSTEM

(TO BE FURNISHED BY THE BIDDER ON ITS LETTER HEAD)

NAME OF THE PROJECT:
We_(Name of the Bidder) hereby request you to give our payments by crediting our bank account directly by E-payment mode as per account details given below. We hereby undertake to intimate IWAI in case of any change in particulars given below and will not hold IWAI responsible for any delay / default due to any technical reasons beyond IWAI"s control: -
Bank Account Number :
RTGS/NEFT/IFSC CODE :
NAME OF THE BANK :
ADDRESS OF THE BRANCH :
OF THE BANK
BRANCH CODE :
ACCOUNT TYPE
(SAVING/CURRENT/OTHERS) :
A BLANK CHEQUE (CANCELLED) IS ENCLOSED HEREWITH.
IWAI hereby declare that the particulars given above are correct and complete. If the transaction is delayed or credit is not affected at all for reasons of incomplete or incorrect information, I/We would not hold IWAI responsible.
Signature of Authorized Signatory Name & Designation
Date:
Place:

ANNEX-IV: BANK CERTIFICATION

		certified				•					
con	rect.						-				
								Author	ized	Signa	tory
Da	te:							Author	rizati	ion	
No	· <u> </u>										
Na	me:										

Official Seal/Stamp

ANNEX-V: TENDER ACCEPTANCE LETTER

(To be given on Company Letter Head)

To,	Date:
	Chief Engineer (NER & Other NNW's), Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301
Sub:	Acceptance of Terms & Conditions of Tender.
Tend	der Reference No:
Nam	ne of Tender/Work: - Design, Construction, Supply, Testing and Commissioning of 50 Pax Hybrid Electric Passenger Ferry
Dear	Sir,
1.	I/ We have downloaded /obtained the Tender document(s) for the above mentioned "Tender/Work" from the website(s) namely: www.iwai.nic.in OR https://eprocure.gov.in/eprocure/app as per your advertisement, given in the above mentioned website(s).
2.	I / We hereby certify that I / we have read the entire Terms and Conditions of the Tender documents from Page Noto(including all documents like annexure(s), schedule(s), etc,), which form part of the contract agreement and I / we shall abide hereby by the terms/conditions/clauses contained therein.
3.	The minutes of the pre-bid meeting (if any) and/ or corrigendum(s) (if any) issued from time to time by your department/organization for this work too have also been taken into consideration, while submitting this acceptance letter.
4.	$I/We\ hereby\ unconditionally\ accept\ the\ Tender\ conditions\ of\ above-mentioned\ Tender\ document(s)/minutes\ of\ the\ Pre-bid\ Meeting\ (if\ any)/corrigendum(s)\ (if\ any)\ in\ its\ totality/Entirety.$
5.	In case any provisions of this Tender are found violated, then your department/ organization shall without prejudice to any other right or remedy be at liberty to reject this Tender/Bid including the forfeiture of the full said earnest money deposit absolutely.
	Yours Faithfully (Signature of the Bidder, with Official)

ANNEX-VI: BANK GUARANTEE FORM FOR EMD

Chief Engineer (NER & Other NNW's), Inland Waterways Authority of India (IWAI), A-13, Sector-1, Noida (U.P.)-201301

WHEREAS		(Name of Tenderer)
(hereinafter called the Te	enderers) wishes to submit his tender for we	ork ofin the
state/s of	herein called "the Tender"	
KNOW ALL PEOPLE Bank)	by these present that we	(Name of
Of	(Name of country) having our	registered office at
() (hereinafter called the "Bank") are	bound unto the Inland
Waterways Authority of	f India (hereinafter called "the Owner")	in the sum of the Rs.
	(Rupees) *for
which payment can trul	ly be made to the said Owner. The Bar	nk bind themselves, their
successors and assigns	by these presents with the common sea	al of the Bank this day
of	2024 and undertake to pay the amo	ount ofRs.
to the en	nployer upon receipt of this written dema	and without the employer
having to substantiate his	s demand.	

The conditions of this obligation are:

If the tenderer withdraws his tender during the period of Tender validity specified in the Form of Tender.

Or

If the Tenderer having been notified of the acceptance of his Tender by the Employer during the period of tender validity fails or refuses to execute the Agreement Form in accordance with the instructions to bidders or fails or refuses to furnish the Performance Security, in accordance with the Instruction to Bidders.

We undertake to pay Owner up to the above amount upon receipt of his written demand, without the Owner having to substantiate his demand, provided that in his demand the Employer will note that the amount claimed by him is due to his owing to the occurrence of any one of the above conditions, specifying the occurred condition or conditions.

This guarantee will remain in force up to eight months or more beyond the last date of receipt of bids or as it may be extended by the Employer, at any time prior to the closing date for

submission of the Tender. Any demand in respect of this guarantee should reach the Bank not later than the above date of expiry of this guarantee.

SIGNATURE OF AUTHORIZED REPRESENTATIVE OF THE BANK

NAME AND DESIGNATION	
SEAL OF THE BANK	
SIGNATURE OF THE WITNESS	
NAME OF THE WITNESS	

ADDRESS OF THE WITNESS

217

ANNEX-VII: LETTER OF CONSENT

(To be filled up by Sub-contractor and submitted in Enclosure II of this tender document)

Sub-contractor's Name: [insert full name]

Applicant's Name: [insert full name]

Date: [insert day, month, year]

Tender No. and title: [insert Tender number and title]

Dear Sir,

The invitation for tendering for [inserts name of work] has been called by [insert name of employer].

In this regard, this is an undertaking that [insert name of Sub-contracting firm] having its office at [insert address of the Sub-contracting firm] is willing to participate as sub-contractor for [insert brief work details for which sub-contractor will provide services] with [insert name of applicant] for this project and we are submitting our relevant experience client certificates for the same.

[Insert name, signature, and seal of authorized representative of the Sub-contractor]

ANNEX-VIII: INTEGRITY AGREEMENT

(To be executed on non-judicial stamp paper of Rs. 100 and signed by the Bidder and the same is to be signed by Authorized Signatory on behalf of IWAI)

This Integrity Agreement is made at on thisday of 2024

BETWEEN

Chairperson, Inland Waterways Authority of India represented through the Chief Engineer (NER & Other NNW's), A-13, Sector-1, Noida (U.P.)-201301, IWAI, (Hereinafter referred as the Principal/Owner", which expression shall unless repugnant to the meaning or context hereof include its successors and permitted assigns)

(1)	Name and Address of the Individual/firm/Company)
through	(Hereinafter referred to as the
	(Details of duly authorized signatory)

"Bidder/Contractor" and which expression shall unless repugnant to the meaning or context hereof include its successors and permitted assigns)

Preamble

AND WHEREAS the Employer values full compliance with all relevant laws of the land, rules, regulations, economic use of resources and of fairness/transparency in its relation with its Bidder(s) and Contractor(s).

AND WHEREAS to meet the purpose aforesaid both the parties have agreed to enter into this Integrity Agreement (hereinafter referred to as "Integrity Pact" or "Pact"), the terms and conditions of which shall also be read as integral part and parcel of the Tender/Bid documents and Contract between the parties.

NOW, THEREFORE, in consideration of mutual covenants contained in this Pact, the parties hereby agree as follows and this Pact witnesses as under:

Article 1: Commitment of the Principal/Owner

- 1. The Employer commits itself to take all measures necessary to prevent corruption and to observe the following principles:
 - (a) No employee of the Principal/Owner, personally or through any of his/her family members, will in connection with the Tender, or the execution of the Contract, demand, take a promise for or accept, for self or third person, any material or immaterial benefit which the person is not legally entitled to.
 - (b) The Employer will, during the Tender process, treat all Bidder(s) with equity and reason. The Employer will, in particular, before and during the Tender process, provide to all Bidder(s) the same information and will not provide to any Bidder(s) confidential / additional information through which the Bidder(s) could obtain an advantage in relation to the Tender process or the Contract execution.
 - (c) The Employer shall endeavour to exclude from the Tender process any person, whose conduct in the past has been of biased nature.
- 2. If the Employer obtains information on the conduct of any of its employees which is a criminal offence under the Indian Penal code (IPC)/Prevention of Corruption Act, 1988 (PC Act) or is in violation of the principles herein mentioned or if there be a substantive suspicion in this regard, the Employer will inform the Chief Vigilance Officer and in addition can also initiate disciplinary actions as per its internal laid down policies and procedures.

Article 2: Commitment of the Bidder(s)/Contractor(s)

- 1. It is required that each Bidder/Contractor (including their respective officers, employees and agents) adhere to the highest ethical standards, and report to the IWAI all suspected acts of fraud or corruption or Coercion or Collusion of which it has knowledge or becomes aware, during the tendering process and throughout the negotiation or award of a contract.
- 2. The Bidder(s)/Contractor(s) commit himself to take all measures necessary to prevent corruption. He commits himself to observe the following principles during his participation in the Tender process and during the Contract execution:
 - a) The Bidder(s)/Contractor(s) will not, directly or through any other person or firm, offer, promise or give to any of the Principal/Owner's employees involved in the tender process or execution of the contract or to any third person any material or other benefit which he/she is not legally entitled to, in order to obtain in exchange any advantage of any kind whatsoever during the Tender process or during the execution of the contract.
 - b) The Bidder(s)/Contractor(s) will not enter with other Bidder(s) into any undisclosed agreement or understanding, whether formal or informal. This applies in particular to prices, specifications, certifications, subsidiary

- contracts, submission or non- submission of bids or any other actions to restrict competitiveness or to cartelize in the bidding process.
- The Bidder(s)/Contractor(s) will not commit any offence under the relevant IPC/PC Act. Further the Bidder(s)/Contract(s) will not use improperly, (for the purpose of competition or personal gain), or pass on to others, any information or documents provided by the Employer as part of the business relationship, regarding plans, technical Bids and business details, including information contained or transmitted electronically.
- d) The Bidder(s)/Contractor(s) of foreign origin shall disclose the names and addresses of agents/representatives in India, if any. Similarly, Bidder(s)/Contractor(s) of Indian Nationality shall disclose names and addresses of foreign agents/representatives, if any. Either the Indian agent on behalf of the foreign principal or the foreign principal directly could bid in a tender but not both. Further, in cases where an agent participate in a tender on behalf of one manufacturer, he shall not be allowed to quote on behalf of another manufacturer along with the first manufacturer in a subsequent/parallel tender for the same item.
- e) The Bidder(s)/Contractor(s) will, when presenting his bid, disclose any and all payments he has made, is committed to or intends to make to agents, brokers or any other intermediaries in connection with the award of the Contract.
- 3. The Bidder(s)/Contractor(s) will not instigate third persons to commit offences outlined above or be an accessory to such offences.
- 4. The Bidder(s)/Contractor(s) will not, directly or through any other person or firm indulge in fraudulent practice means a wilful misrepresentation or omission of facts or submission of fake/forged documents in order to induce public official to act in reliance thereof, with the purpose of obtaining unjust advantage by or causing damage to justified interest of others and/or to influence the procurement process to the detriment of the Government interests.
- 5. The Bidder(s)/Contractor(s) will not, directly or through any other person or firm use Coercive Practices (means the act of obtaining something, compelling an action or influencing a decision through intimidation, threat or the use of force directly or indirectly, where potential or actual injury may befall upon a person, his/ her reputation or property to influence their participation in the tendering process).

Article 3: Consequences of Breach

Without prejudice to any rights that may be available to the Employer under law or the Contract or its established policies and laid down procedures, the Employer shall have the following rights in case of breach of this Integrity Pact by the Bidder(s)/Contractor(s) and the

bidder/Contractor accepts and undertakes to respect and uphold the Principal/Owner's absolute right:

- 1. If the Bidder(s)/Contractor(s), either before award or during execution of Contract has committed a transgression through a violation of Article 2 above or in any other form, such as to put his reliability or credibility in question, the Employer after giving 14 days" notice to the Contractor shall have powers to disqualify the Bidder(s)/Contractor(s) from the tender process or terminate/determine the Contract, if already executed or exclude the Bidder/Contractor from future contract award processes. The imposition and duration of the exclusion will be determined by the severity of transgression and determined by the Principal/Owner. Such exclusion may be forever or for a limited period as decided by the Principal/Owner.
- 2. Forfeiture of EMD/Performance Guarantee/Security Deposit: If the Employer has disqualified the Bidder(s) from the tender process prior to the award of the contract or terminated/determined the contract or has accrued the right to terminate/determine the contract according to Article 3(1), the Employer apart from exercising any legal rights that may have accrued to the Principal/Owner, may in its considered opinion forfeit the entire amount of Earnest Money Deposit, Performance Guarantee and Security Deposit of the Bidder/Contractor.
- 3. Criminal Liability: If the Employer obtains knowledge of conduct of a bidder or Contractor, or of an employee or a representative or an associate of a bidder or Contractor which constitutes corruption within the meaning of IPC Act, or if the Employer has substantive suspicion in this regard, the Employer will inform the same to law enforcing agencies for further investigation.

Article 4: Previous Transgression

- 1) The Bidder declares that no previous transgressions occurred in the last 5 years with any other Company in any country confirming to the anticorruption approach or with Central Government or State Government or any other Central/State Public Sector Enterprises in India that could justify his exclusion from the Tender process.
- 2) If the Bidder makes incorrect statement on this subject, he can be disqualified from the Tender process or action can be taken for banning of business dealings/ holiday listing of the Bidder/Contractor as deemed fit by the Principal/ Owner.
- 3) If the Bidder/Contractor can prove that he has resorted / recouped the damage caused by him and has installed a suitable corruption prevention system, the Employer may, at its own discretion, revoke the exclusion prematurely.

Article 5: Equal Treatment of all Bidders/Contractors

1) The Bidder/Contractor shall be responsible for any violation(s) of the principles laid down in this agreement/Pact by any of its sub-vendors.

- 2) The Employer will enter into Pacts on identical terms as this one with all Bidders and Contractors.
- 3) The Employer will disqualify Bidders, who do not submit, the duly signed Pact between the Employer and the bidder, along with the Tender or violate its provisions at any stage of the Tender process, from the Tender process.

Article 6: Duration of the Pact

This Pact begins when both the parties have legally signed it. It expires for the Contractor 12 months after the completion of work under the contract.

If any claim is made/lodged during the time, the same shall be binding and continue to be valid despite the lapse of this Pacts as specified above, unless it is discharged/determined by the Employer.

Article 7: Other Provisions

- 1) This Pact is subject to Indian Law, place of performance and jurisdiction is the Headquarters of the Division of the Employer, who has floated the Tender.
- 2) Changes and supplements need to be made in writing. Side agreements have not been made.
- 3) If the Contractor is a partnership or a consortium, this Pact must be signed by all the partners or by one or more partner holding power of attorney signed by all partners and consortium members. In case of a Company, the Pact must be signed by a representative duly authorized by board resolution.
- 4) Should one or several provisions of this Pact turn out to be invalid; the remainder of this Pact remains valid. In this case, the parties will strive to come to an agreement to their original intensions.
- 5) It is agreed term and condition that any dispute or difference arising between the parties with regard to the terms of this Integrity Agreement / Pact, any action taken by The Employer/Principal in accordance with this Integrity Agreement/ Pact or interpretation thereof shall not be subject to arbitration.

Article 8: LEGAL AND PRIOR RIGHTS

All rights and remedies of the parties hereto shall be in addition to all the other legal rights and remedies belonging to such parties under the Contract and/or law and the same shall be deemed to be cumulative and not alternative to such legal rights and remedies aforesaid. For the sake of brevity, both the Parties agree that this Integrity Pact will have precedence over the Tender/Contact documents with regard any of the provisions covered under this Integrity Pact.

place and date first above mentioned in the presence of following witnesses:
(For and on behalf of Employer)
(For and on behalf of Bidder/Contractor)
WITNESSES:
1
(signature, name and address)
2
(signature, name and address)
Place:
Date: